LA--12312-MS DE92 017291

TRANSX 2: A Code for Interfacing MATXS Cross-Section Libraries to Nuclear Transport Codes

R. E. MacFarlanc





AND THE PROPERTY OF THE TRANSPORT

Los Alamos National Laboratory Los Alamos New Mexico 87545

CONTENTS

Ι.	II	NTRODUCTION	1
11.	D	ATA NEEDS OF TRANSPORT CODES	3
	Α.	Transport Approximations	4
	B .	Fission Source	5
	C	Transport Tables	7
	D.	Neutron-Photon Coupled Sets	8
	E.	Charged Particle Coupled Sets	10
	Г. Г	The Adjuint Problem	11
	r. C	Solf Shielding	11
	и. и	Denself Connections	15
	n.	Lateralation in Const Section Tables	10
	1.	Interpolation in Cross Section Tables	10
	J.		11
	K.		18
	L.		20
	M .	Photon Production Self-Shielding	21
	Ν.	Thermal Cross Sections	21
	0.	Epithermal Self-Shielding	22
III.	H	OW TO RUN TRANSX	23
	Α.	TRANSX Input Instructions	23
	B .	Free Form Input	26
	С.	Simple Transport Tables	26
	D.	Advanced Response-Function Edits	29
	Ε.	Self-Shielded Fast Reactor Problems	34
	F.	Self-Shielded Thermal Reactor Problems	39
IV.	IP	TERFACE FILES	45
	Α.	The MATXS Generalized Cross-Section File	45
	B.	RTFLUX, RZFLUX, RMFLUX, and RZMFLX Files	48
	B. C	RTFLUX, RZFLUX, RMFLUX, and RZMFLX Files	48 49
	B. C. D	RTFLUX, RZFLUX, RMFLUX, and RZMFLX Files CARD 6E12 Format	48 49 49
	B. C. D. E	RTFLUX, RZFLUX, RMFLUX, and RZMFLX Files CARD 6E12 Format CLAW Format FIDO Output	48 49 49 49
	B. C. D. E.	RTFLUX, RZFLUX, RMFLUX, and RZMFLX Files CARD 6E12 Format CLAW Format FIDO Output ANISN Output	48 49 49 49 49 50
	B. C. D. E. F.	RTFLUX, RZFLUX, RMFLUX, and RZMFLX Files CARD 6E12 Format CLAW Format FIDO Output ANISN Output LSOTXS Output	48 49 49 49 50
	B. C. D. E. F. G.	RTFLUX, RZFLUX, RMFLUX, and RZMFLX Files CARD 6E12 Format CLAW Format FIDO Output ANISN Output ISOTXS Output	48 49 49 49 50 50
Ň	B. C. D. E. F. G. H.	RTFLUX, RZFLUX, RMFLUX, and RZMFLX Files CARD 6E12 Format CLAW Format FIDO Output ANISN Output ISOTXS Output GOXS Output	48 49 49 50 50 51
V.	B. C. D. E. F. G. H. S	RTFLUX, RZFLUX, RMFLUX, and RZMFLX Files CARD 6E12 Format CLAW Format FIDO Output ANISN Output ISOTXS Output GOXS Output OMF. MATXS LIBRARIES	48 49 49 50 50 51 53
۷.	B. C. D. E. F. G. H. S. A.	RTFLUX, RZFLUX, RMFLUX, and RZMFLX Files CARD 6E12 Format CLAW Format FIDO Output ANISN Output ISOTXS Output GOXS Output OMF. MATXS LIBRARIES MATXS10	48 49 49 50 50 51 53 53
۷.	B. C. D. E. F. G. H. S. A. B.	RTFLUX, RZFLUX, RMFLUX, and RZMFLX Files CARD 6E12 Format CLAW Format FIDO Output ANISN Output ISOTXS Output GOXS Output OMF. MATXS LIBRARIES MATXS5	48 49 49 50 50 51 53 53 53
V.	B. C. D. E. F. G. H. S. A. B. C.	RTFLUX, RZFLUX, RMFLUX, and RZMFLX Files CARD 6E12 Format CLAW Format FIDO Output ANISN Output ISOTXS Output GOXS Output OMF. MATXS LIBRARIES MATXS10 MATXS11	48 49 49 50 50 51 53 53 53 53
٧.	B. C. D. E. F. G. H. S. A. B. C. D.	RTFLUX, RZFLUX, RMFLUX, and RZMFLX Files CARD 6E12 Format CLAW Format FIDO Output ANISN Output ISOTXS Output GOXS Output OMF. MATXS LIBRARIES MATXS10 MATXS11 MATXS6	48 49 49 50 50 51 53 53 53 53 53 53
V.	B. C. D. E. F. G. H. S. A. B. C. D. E.	RTFLUX, RZFLUX, RMFLUX, and RZMFLX FilesCARD 6E12 FormatCLAW FormatFIDO OutputANISN OutputISOTXS OutputGOXS OutputOMF. MATXS LIBRARIESMATXS10MATXS11MATXS6MATXS12	48 49 49 50 50 51 53 53 53 53 53 54 54
v.	B. C. D. E. F. G. H. S. A. B. C. D. E. F.	RTFLUX, RZFLUX, RMFLUX, and RZMFLX FilesCARD 6E12 FormatCLAW FormatFIDO OutputANISN OutputISOTXS OutputGOXS OutputOMF. MATXS LIBRARIESMATXS10MATXS11MATXS12MATXS12MATXS12MATXS7	48 49 49 50 51 53 53 53 53 53 54 54 54
V.	B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. F. G.	RTFLUX, RZFLUX, RMFLUX, and RZMFLX FilesCARD 6E12 FormatCLAW FormatFIDO OutputANISN OutputISOTXS OutputGOXS OutputOMF. MATXS LIBRARIESMATXS10MATXS5MATXS11MATXS12MATXS7MATXS13	48 49 49 50 51 53 53 53 53 54 54 54 54
v.	B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. H. S. B. C. D. E. F. G. H. S. C. D. E. F. G. H. S. C. D. E. F. G. H. S. C. D. E. F. G. H. S. G. H. S. G. H. S. G. H. S. G. H. S. G. H. S. G. H. S. G. H. S. G. H. S. G. H. S. G. H. S. G. H. S. G. H. S. G. H. S. G. H. S. G. B. S. S. S. S. S. S. S. S. S. S. S. S. S.	RTFLUX, RZFLUX, RMFLUX, and RZMFLX FilesCARD 6E12 FormatCLAW FormatFIDO OutputANISN OutputISOTXS OutputGOXS OutputOMF. MATXS LIBRARIESMATXS10MATXS11MATXS6MATXS12MATXS13MATXS13MATXS8	48 49 49 50 51 53 53 53 53 53 54 54 54 55
V. VI.	B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. H. P	RTFLUX, RZFLUX, RMFLUX, and RZMFLX Files CARD 6E12 Format CLAW Format FIDO Output ANISN Output ISOTXS Output GOXS Output OMF. MATXS LIBRARIES MATXS10 MATXS11 MATXS12 MATXS13 MATXS13 MATXS8 ROGRAMMING DETAILS	48 49 49 50 51 53 53 53 53 53 54 54 54 55 57
V. VI.	B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. H. S. S. A. B. C. D. E. F. G. H. S. S. A. B. C. D. E. F. G. B. F. S. S. S. S. S. S. S. S. S. S. S. S. S.	RTFLUX, RZFLUX, RMFLUX, and RZMFLX FilesCARD 6E12 FormatCLAW FormatFIDO OutputANISN OutputISOTXS OutputGOXS OutputOMF. MATXS LIBRARIESMATXS10MATXS11MATXS12MATXS12MATXS13MATXS13MATXS8ROGRAMMING DETAILSVersion Control	48 49 49 50 51 53 53 53 53 54 54 55 57 57
V. VI.	B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. H. S. A. B. B. F. B. B. B. B. B. B. B. B. B. B. B. B. B.	RTFLUX, RZFLUX, RMFLUX, and RZMFLX FilesCARD 6E12 FormatCLAW FormatFIDO OutputANISN OutputISOTXS OutputGOXS OutputOMF. MATXS LIBRARIESMATXS10MATXS11MATXS11MATXS6MATXS12MATXS13MATXS8ROGRAMMING DETAILSVersion ControlCode Conversion	48 49 49 50 51 53 53 53 53 54 54 55 57 59
V. VI.	B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. H. S. C. D. E. F. G. H. S. C. D. E. F. G. H. S. C. D. E. F. G. H. S. C. D. E. F. G. H. S. C. D. E. F. G. B. C. D. E. F. G. B. C. D. E. F. G. B. C. D. E. F. G. B. C. D. C. S. C. D. C. S. C. S. C. S. C. S. C. S. C. S. S. S. S. S. S. S. S. S. S. S. S. S.	RTFLUX, RZFLUX, RMFLUX, and RZMFLX Files CARD 6E12 Format CLAW Format FIDO Output ANISN Output ISOTXS Output GOXS Output MATXS10 MATXS5 MATXS12 MATXS12 MATXS13 MATXS13 MATXS13 MATXS8 ROGRAMMING DETAILS Version Control Code Conversion Storage Allocation	48 49 50 51 53 53 53 53 54 54 55 57 59 60
V. VI.	B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. H. S. C. D. E. F. G. H. S. C. D. E. F. G. H. S. D. E. F. G. H. S. D. E. F. G. H. S. D. E. F. G. H. S. D. E. F. G. H. S. D. E. F. G. H. S. D. E. F. G. D. E. F. G. D. E. F. G. D. E. F. G. D. D. E. F. G. D. D. D. D. D. D. D. D. D. D. D. D. D.	RTFLUX, RZFLUX, RMFLUX, and RZMFLX FilesCARD 6E12 FormatCLAW FormatFIDO OutputANISN OutputISOTXS OutputGOXS OutputOMF. MATXS LIBRARIESMATXS10MATXS11MATXS5MATXS12MATXS12MATXS13MATXS8ROGRAMMING DETAILSVersion ControlCode ConversionStorage AllocationDescription of TRANSX	48 49 50 51 53 53 53 54 54 55 57 50 61
V. VI.	B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. H. P. F.	RTFLUX, RZFLUX, RMFLUX, and RZMFLX FilesCARD 6E12 FormatCLAW FormatFIDO OutputANISN OutputISOTXS OutputGOXS OutputOMF. MATXS LIBRARIESMATXS10MATXS11MATXS6MATXS12MATXS13MATXS8ROGRAMMING DETAILSVersion ControlCode ConversionStorage AllocationDescription of TRANSXError Messages	48 49 49 50 51 53 53 53 53 54 49 57 57 59 60 70
V. VI.	B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. H. S. D. E. F. G. H. S. D. E. F. G. H. S. S. S. G. B. S. S. S. S. S. S. S. S. S. S. S. S. S.	RTFLUX, RZFLUX, RMFLUX, and RZMFLX Files CARD 6E12 Format CLAW Format FIDO Output ANISN Output ISOTXS Output GOXS Output OMF. MATXS LIBRARIES MATXS10 MATXS11 HATXS6 MATXS12 MATXS13 MATXS13 MATXS8 ROGRAMMING DETAILS Version Control Code Conversion Storage Allocation Description of TRANSX Error Messages	48 49 49 50 51 53 53 53 53 54 49 57 59 60 61 73
V. VI. VII.	B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. H. S. C. D. E. F. G. H. S. C. D. E. F. G. H. S. C. D. E. F. G. H. S. C. D. E. F. G. H. S. C. D. E. F. G. H. S. C. D. E. F. G. H. S. C. D. E. F. G. B. C. D. E. F. G. B. C. D. E. F. G. B. C. D. E. S. C. D. E. S. C. D. C. S. C. D. C. D. C. D. C. S. S. S. S. S. S. S. S. S. S. S. S. S.	RTFLUX, RZFLUX, RMFLUX, and RZMFLX Files CARD 6E12 Format CLAW Format FIDO Output ANISN Output ISOTXS Output GOXS Output OMF. MATXS LIBRARIES MATXS10 MATXS11 P(ATXS6) MATXS12 MATXS13 MATXS13 MATXS13 MATXS8 ROGRAMMING DETAILS Version Control Code Conversion Storage Allocation Description of TRANSX Error Messages sing BBC BBX Input Instructions	48 49 49 50 51 53 53 53 53 54 49 57 59 60 70 73 73
V. VI. VII.	B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. H. S. A. B. C. D. E. F. G. H. S. C. D. E. F. G. H. S. C. D. E. F. G. H. S. C. D. E. F. G. H. S. C. D. E. F. G. H. S. C. D. E. F. G. H. S. C. D. E. F. G. B. C. D. E. F. G. B. C. D. E. F. G. B. C. D. E. F. G. B. C. D. E. F. G. B. C. D. C. D. C. B. C. D. C. B. C. D. C. B. C. D. C. B. B. C. D. C. B. C. B. C. D. C. B. C. B. C. D. C. B. C. D. C. B. C. D. C. B. C. D. C. B. C. D. C. D. C. C. D. C. C. D. D. D. C. D. C. D. C. D. C. D. D. C. D. C. D. C. D. C. D. D. C. D. C. D. C. D. D. D. D. C. D. C. D. D. D. D. D. D. D. D. D. D. D. D. D.	RTFLUX, RZFLUX, RMFLUX, and RZMFLX Files CARD 6E12 Format CLAW Format FIDO Output ANISN Output ISOTXS Output GOXS Output OMF. MATXS LIBRARIES MATXS10 MATXS11 P(ATXS6) MATXS12 MATXS13 MATXS13 MATXS13 MATXS8 ROGRAMMING DETAILS Version Control Code Conversion Storage Allocation Description of TRANSX Error Messages sing BBC BBX Input Instructions	$\begin{array}{c} 48\\ 49\\ 49\\ 50\\ 51\\ 53\\ 53\\ 53\\ 53\\ 54\\ 54\\ 55\\ 57\\ 59\\ 61\\ 77\\ 73\\ 75\\ \end{array}$

VIII. ACKN	OWLEDGMENTS	76
IX. REFER	RENCES	77
Appendix A.	SAMPLE OUTPUT LISTINGS	81
Appendix B.	MATXS FILE DESCRIPTION	89
Appendix C.	RTFLUX FILE DESCRIPTION	97
Appendix D.	RZFLUX FILE DESCRIPTION	99
Appendix E.	RZMFLX FILE DESCRIPTION	101
Appendix F.	ISOTXS FILE DESCRIPTION	103
Appendix G.	GOX ⁴ FILE DESCRIPTION	111
Appendix H.	EXAMPLE OF A MATXS INDEX FILE	115

.

TRANSX 2: A Code for Interfacing MATXS Cross-Section Libraries to Nuclear Transport Codes

by

R. E. MacFarlane

ABSTRACT

TRANSX 2.0 is a computer code that reads nuclear data from a library in MATXS format and produces transport tables compatible with many discrete-ordinates (S_N) and diffusion codes. Tables can be produced for neutron, photon, charged-particle, or coupled transport. Options include adjoint tables, mixtures, homogeneous or heterogeneous self-shielding, group collapse, homogenization, thermal upscatter, prompt or steady-state fission, transport corrections, elastic removal corrections, and flexible response function edits. This report describes version 2.0 of TRANSX, which was prepared in January of 1992.

I. INTRODUCTION

Discrete-ordinates (S_N) transport codes, which solve the Boltzmann equation for the distribution of neutrons and photons in nuclear systems, have reached a high level of development. The early one-dimensional codes DTF-IV¹ and ANISN² are very widely used. The development of diffusion acceleration³ as well as increasing computer speed and capacity has made detailed transport calculations more economical; as a result, codes such a ONEDANT⁴ and TWODANT⁵ are seeing increasing use. The toroidal geometry capability of TRIDENT-CTR⁶ and its successor TRISM⁷ are very useful for fusion reactor analysis. The DIF-3D⁸ diffusion code and Monte-Carlo codes with multigroup capability like MCNP⁹ are also used frequently.

However, many of the users of transport codes have the same complaint: it is hard to get good, up-to-date, documented cross-section data and prepare them for input into these codes. The difficulties are multiplied if there is anything unusual about the problem, such as fine groups, self-shielding, transport cross sections, or sophisticated response edits (*e.g.*, heating, damage, or gas production). This report describes a utility code called TRANSX (for transport cross-section code) that works together with a generalized cross-section library called MATXS (for material cross-section library) to give the transport code user easier access to appropriate nuclear data and some capabilities difficult or impossible to get with any other system.

The version of TRANSX described in this report can be used to construct data for fusion reactors, fast fission reactors, thermal fission reactors, and shielding problems. Its main weakness is in computing resonance effects in thermal reactors.

TRANSX was originally developed in the late seventies to handle cross sections for fission, fusion, and shielding applications at Los Alamos. In the early eighties, extensions to handle heterogeneous self-shielding problems for fast reactors were added. In 1984, a version without the fast reactor features was released as TRANSX.CTR.¹⁰ It was especially well-suited for problems in controlled thermonuclear research (CTR) because of its ability to prepare coupled tables and response edits for heating, damage, gas production, and delayed activity. Various later versions have been made available without formal release or complete documentation; for example, Version 1.11 has been available as S11 at Los Alamos, on the Magnetic Fusion Energy (MFE) computer system at Livermore, and on the San Diego Super Computer Center system since late 1987. Version 1.13, which includes charged particle capabilities, has been used extensively at Los Alamos. This gradual accumulation of changes and new capabilities, plus the need to react to a new version of the MATXS format developed for NJOY 91, plus the need for a fully documented code more consistent with modern Quality Assurance (QA)requirements, has led to the creation of this new version.

Section II of this report discusses the data needs of the typical S_N codes and describes the theoretical background; Section III gives detailed examples of a wide variety of problems that can be solved using TRANSX; Section IV contains descriptions of the input and output files used by TRANSX, including the MATXS cross-section library; Section V summarizes several available MATXS libraries; and Section VI gives a detailed discussion of the code aimed at programmers who have to convert, modify, or maintain the code. Several appendices are attached that give the format specifications for the input and output interface files and partial listings of the outputs from the sample problems.

II. DATA NEEDS OF TRANSPORT CODES

The S_N transport codes solve the equation¹¹

$$\mu \frac{\partial}{\partial x} \phi_g(\mu, x) + \sigma_g^{SN}(x) \phi_g(\mu, x) = \sum_{\ell=0}^N P_\ell(\mu) \sum_{g'} \sigma_{\ell g \leftarrow g'}^{SN}(x) \phi_{lg'} + S_g(\mu, x) \quad , \quad (1)$$

where one-dimensional plane geometry has been used for simplicity, μ is the scattering cosine, x is position, $\phi(\mu, x)$ is the angular flux for group g, ϕ_{lg} is the Legendre flux for group g, $P_{\ell}(\mu)$ is a Legendre polynomial, and $S_g(\mu, x)$ is the external and fission source into group g. The cross sections in Eq. (1) must be defined to make ϕ_g as close as possible to the solution of the Boltzmann equation. As shown in the reference,¹¹ the multigroup Boltzmann equation can be written in the P_N form:

$$\mu \frac{\partial}{\partial x} \psi(\mu, x) + \sum_{\ell=0}^{N} P_{\ell}(\mu) \sigma_{\ell \ell g}^{PN}(x) \psi_{\ell g} = \sum_{\ell=0}^{N} P_{\ell}(\mu) \sum_{g'} \sigma_{\ell g \leftarrow g'}^{PN}(x) \psi_{\ell g'} + S_{g}(\mu, x) , \quad (2)$$

where the P_N cross sections are given by the following group averages:

$$\sigma_{\ell tg}^{PN} = \frac{\int_{g} \sigma_{\ell}(E) W_{\ell}(E) dE}{\int_{g} W_{\ell}(E) dE} , \qquad (3)$$

and

$$\sigma_{\ell g \leftarrow g'}^{PN} = \frac{\int_{g'} dE' \int_{g} dE \, \sigma_{\ell}(E' \rightarrow E) \, W_{\ell}(E')}{\int_{g'} dE' \, W_{\ell}(E')} \quad . \tag{4}$$

In these formulas, $\sigma_t(E)$ and $\sigma_t(E' \to E)$ are the basic energy dependent total and scattering cross sections, and $W_t(E)$ is a weighting flux that should be chosen to be as similar to ψ as possible. As will be seen later, these P_N cross sections are available on the MATXS libraries produced by NJOY.^{12, 13, 14, 15} When Eq. (2) is compared with Eq. (1). it is evident that the S_N equations require

$$J_{\ell g \leftarrow g'}^{SN} = \sigma_{\ell g \leftarrow g'}^{PN} \quad \text{for } g' \neq g \quad , \tag{5}$$

and

$$\sigma_{\ell g \leftarrow g}^{SN} = \sigma_{\ell g \leftarrow g}^{PN} - \sigma_{\ell \iota g}^{PN} + \sigma_g^{SN} \quad , \tag{6}$$

where σ_g^{SN} is not determined. The choice of σ_g^{SN} gives rise to a "transport approximation" and various recipes are in use.

A. Transport Approximations

It is convenient to write

$$\sigma_{\ell_g \leftarrow g}^{SN} = \sigma_{\ell_g \leftarrow g}^{PN} - (\sigma_{\ell_l g}^{PN} - \sigma_{olg}^{PN}) - \Delta_g^N \quad , \tag{7}$$

and

$$\sigma_g^{SN} = \sigma_{otg}^{PN} - \Delta_g^N \quad . \tag{8}$$

The term in parentheses corrects for the anisotropy in the total reaction rate term of the Boltzmann equation, and Δ_g^N can be chosen to minimize the effects of truncating the Legendre expansion at $\ell=N$. The recipes available in TRANSX are as follows:

Consistent-P approximation:

$$\Delta_{\mathbf{g}}^{N} = \mathbf{0} \quad , \tag{9}$$

Inconsistent-P approximation:

$$\Delta_g^N = \sigma_{olg}^{PN} - \sigma_{N+1,lg}^{PN} , \qquad (10)$$

Diagonal transport approximation:

$$\Delta_g^N = \sigma_{otg}^{PN} - \sigma_{N+1,ig}^{PN} + \sigma_{N+1,g-g}^{PN} , \qquad (11)$$

Bell-Hansen-Sandmeier or extended transport approximation:

$$\Delta_g^N = \sigma_{otg}^{PN} - \sigma_{N+1,tg}^{PN} + \sum_{g'} \sigma_{N+1,g' \leftarrow g}^{PN} , \qquad (12)$$

and

Inflow transport approximation:

$$\Delta_g^N = \sigma_{otg}^{PN} - \sigma_{N+1,tg}^{PN} + \frac{\sum\limits_{g'} \sigma_{N+1,g \leftarrow g'}^{PN} \phi_{N+1,g'}}{\phi_{N+1,g}} \quad . \tag{13}$$

The first two approximations are most appropriate when the scattering orders above N are small. The inconsistent option removes most of the delta-function

of forward scattering introduced by the correction for the anisotropy in the total scattering rate and should normally be more convergent than the consistent option. For libraries produced with an *l*-independent flux guess and in the absence of selfshielding, the difference between "consistent" and "inconsistent" vanishes.

The diagonal and Bell-Hansen-Sandmeier (BHS) recipes make an attempt to correct for anisotropy in the scattering matrix and are especially effective for forward-peaked scattering. The BHS form is most often used, but the diagonal option can be substituted when BHS produces negative values.

The inflow recipe makes the N+1 term of the P_N expansion vanish, but it requires a good knowledge of the N+1 flux moment from some previous calculation. Inflow reduces to BHS for systems in equilibrium by detail balance (*i.e.*, the thermal region). The diffusion approximation obtained using the inflow formula is equivalent to a P_1 transport solution.

These corrections require data from the (N+1)-th Legendre moments of the cross sections to prepare a corrected N-table set. The user of the TRANSX program should be certain that, in fact, (N+1) tables are available on the MATXS file when producing an N-table transport-corrected set. Absence of the (N+1) tables is not fatal, but obviously the cross sections will not contain the desired corrections.

B. Fission Source

The source of fission neutrons into a group is given by

$$S_g = \sum_{g'} \sigma_{fg \leftarrow g'} \phi_{0g'} \quad , \tag{14}$$

where $\sigma_{fg \leftarrow g'}$ is the fission matrix that is available on the MATXS file. However, most existing transport codes do not use this matrix form because the upscatter is expensive to handle and a reasonably accurate alternative exists. Except for relatively high neutron energies, the spectrum of fission neutrons is only weakly dependent on initial energy. Therefore, the fission source can be written as

$$S_{g} = \chi_{g} \sum_{g'} \bar{\nu}_{g'} \sigma_{fg'} \phi_{0g'} , \qquad (15)$$

5

where $\bar{\nu}_g$ is the fission neutron yield, σ_{fg} is the fission cross sections, and χ_g is the average fission spectrum, which can be defined by

$$\chi_{g} = \frac{\sum_{g'} \sigma_{fg \leftarrow g'} \phi_{0g'}}{\sum_{g'} \bar{\nu}_{g'} \sigma_{fg'} \phi_{0g'}} , \qquad (16)$$

where the fission neutron production rate can also be written as

$$\sum_{g'} \bar{\nu}_{g'} \,\sigma_{fg'} \,\phi_{0g'} = \sum_{g} \sum_{g'} \sigma_{fg \leftarrow g'} \,\phi_{0g'} \quad . \tag{17}$$

Clearly, χ_g as given by Eq. (16) depends on the flux in the system of interest. The dependence is weak except for high incident energies, and a rough guess for ϕ_{0g} usually gives an accurate spectrum. When this is not the case, a sequence of calculations can be made, using the flux from each step to improve the χ_g for the next step.

The matrix in a MATXS file represents the prompt part of fission only. Steadystate (SS) fission is obtained using two auxiliary pieces of data: delayed $\bar{\nu}$ and delayed χ . Therefore,

$$\bar{\nu}_{g'}^{SS}\sigma_{fg'} = \sum_{g} \sigma_{fg \leftarrow g'} + \bar{\nu}_{g'}^D \sigma_{fg'} \quad , \tag{18}$$

and

$$\chi_{g}^{SS} = \frac{\sum_{g'} \sigma_{jg \leftarrow g'} \phi_{0g'} + \chi_{g}^{D} \sum_{g'} \bar{\nu}_{g'}^{D} \sigma_{jg'} \phi_{0g'}}{\sum_{g} \sum_{g'} \sigma_{jg \leftarrow g'} \phi_{c0'} + \sum_{g'} \bar{\nu}_{g'}^{D} \sigma_{jg'} \phi_{0g'}'}$$
(19)

For large group structures, the square fission matrices can consume a large part of the library. As already noted, the shape of the fission spectrum is not strongly dependent on initial energy. In fact, the shape is independent of energy at low energies. Therefore, the fission matrix can be split into slow and fast parts,

$$\sigma_{fg \leftarrow g'} = \chi_g^S \left(\bar{\nu} \sigma_f \right)_{g'}^S + \sigma_{fg \leftarrow g'}^F \quad . \tag{20}$$

Starting with NJOY 91.0, the GROUPR module automatically determines the energy range over which the fission spectrum is constant, and it generates a production cross section and normalized spectrum for this range. At higher energies, it produces a rectangular fission matrix. The new version of the MATXS format established for NJOY 91 uses a generalized format for transfer matrices that allows for this decomposition of a matrix into two vectors and a smaller matrix. TRANSX 2.0 can then reconstruct the full representation of fission using Eq. (20).

The fission representation for some materials may be divided into the separate reactions (n, f), (n, n'f), (n, 2nf), and (n, 3nf). Because the second-, third-, and fourth-chance fission channels are only open at high energies (for example, above 6 MeV for ²³⁵U), their corresponding fission matrices are always rectangular on the MATXS library and do not cause a space problem. In any case, the prompt fission source produced by TRANSX includes all of these partial matrices. This full matrix treatment is important for 14-MeV source problems (for example, fusion-fission hybrid design).

This approach gives TRANSX a very general and efficient fission source capability. The same library can be used to compute χ for a fast or thermal system; in fact, if region fluxes are available, χ for a given material can be different in each region of a problem.

C. Transport Tables

The data required for an S_N flux solution using Eq. (1) have been shown to be

$$\sigma_g, \sigma_{\ell g \leftarrow g'}, \bar{\nu}_g \sigma_{fg}, \text{ and } \chi_g,$$

where the superscript SN has been omitted for simplicity. The S_N codes also traditionally use a particle-balance cross section (often loosely called "absorption") defined by

$$\sigma_{ag} = \sigma_g - \sum_{g'} \sigma_{0g' \leftarrow g} \quad . \tag{21}$$

This quantity is also computed by TRANSX by adding all the absorption reactions $[(n, \gamma), fission, (n, p), (n, \alpha), etc.]$ and subtracting (n, 2n), twice (n, 3n), and so on. The two methods are formally equivalent, except that small numerical differences due to cross-section processing lead to unreasonable values for σ_{ag} , as computed from Eq. (21), when σ_{ag} is small with respect to σ_g . In such cases, σ_{ag} is replaced by the value from the direct calculation, and the σ_g position of the transport table is adjusted accordingly. Note that σ_{ag} can be negative if more particles are produced by (n, xn) reactions than are absorbed. Negative values occur in coupled sets as well, as will be seen below.

When the flux calculation is complete, it is often necessary to compute some response such as Leating, radiation damage, gas production, photon production,

or dose to tissue. Therefore, S_N codes allow for reading several response-function edit cross sections, σ_{Eg} .

The original S_N codes read χ as a special array, and the cross sections were arranged into "transport tables" by "position" as shown in Table 1. Note that the positions containing scattering data give all the source groups that scatter into the same final or "sink" group. Even the newer codes retain many features of this structure, and TRANSX input uses terminology based on the table.

A transport table like this is required for each group, Legendre order, and material. The tables may be "material ordered" or "group ordered".

Material ordering is the natural result of preparing a library from evaluated nuclear data. The MATXS files are material ordered. The tables arc written onto a library with the outermost loop being material, then Legendre order, then group as the inner loop.

However, group ordering is the way the cross sections are needed in the codes. The S_N equations are solved by sweeping down from group 1 to the lower energy groups, so cross sections for all materials and orders are needed for each group in sequence. The library is ordered with group as the outermost loop, then Legendre order, then material. Of course, the S_N codes contain the logic to change the input library from material to group ordering if the entire cross-section set will fit into the available memory. If the group structure is too large for the internal logic, the group-ordered TRANSX output can be requested, thereby allowing very large group structures to be used for transport calculations. It may be noted that fusion-related transport calculations (other than hybrid studies) do not require iteration to converge a fission source, so they characteristically employ larger group structures than are practical with multiplying systems. Thus, this group-ordered option, as well as the adjoint option discussed below, may be of particular value in fusion.

D. Neutron-Photon Coupled Sets

The discrete-ordinates equation can be applied equally well to neutron and photon transport; it is only necessary to provide the appropriate neutron or photon cross sections for the transport tables. In both cases, downscatter predominates (excluding fission), and the equations can be solved by sweeping through the groups from high energy to low energy.

Since photon production from neutron reactions is usually stronger than photoneutron production, the same downscatter structure can be obtained by defining a

Position		Contents for Group g	
1 : NED	}	σ _{Eg}	Response edits
NED+1	••••	 σ _{eg}	
NED+2		ν _g σ _{fg}	Standard edits
NED+3		σg	
NED+4		•••••	•••••
•	}	$\sigma_{g \leftarrow g'}$	Upscatter $(g' > g)$
NED+NUP+3			
• • • • • • • • • • • • • • • • • • •	•••••	••••	• • • • • • • • • • • • • • • • • • • •
NED+NUP+4	•	σ _{g⊷g'}	In-group $(\gamma' = g)$
NED+NUP+5	•••••	••••	•••••
:	}	σ _{g⊷g} ,	Downscatter $(g' < g)$
NTABL			

NED	= number of extra response edits (NED≥0)
UP	= maximum number of a scatter groups (0 < NUP < NGROUP)
NTABL	= table length (NED+4+NUP <ntabl<ned+4+nup+ngroup)< td=""></ntabl<ned+4+nup+ngroup)<>
NGROUP	= number of energy groups
IPTOT	= NED+3 = position of total cross section
	NED UP NTABL NGROUP IPTOT





Figure 1: Arrangement of neutron (n) and photon (g) cross sections in a coupled transport table with no upscatter.

composite gro_p structure with N neutron groups arranged as groups 1 through N, and G photon groups attached as groups N+1 through N+G. The arrangement is shown schematically in Fig. 1. Photoneutron production could be included with fission.

A coupled set provides the convenience of solving both the neutron and photon problems simultaneously with a single transport code run. Because more than one photon is normally produced by a neutron reaction (i.e., cascades), the particlebalance absorption edit σ_{ag} will usually be large and negative for a coupled set. Special features are available in some transport codes to allow the photon part of a response to be computed separately from the neutron part. As will be seen later, the GOXS interface format includes a parameter LNG to define the break point between the neutron and photon parts of the group structure.

E. Charged-Particle Coupled Sets

This version of TRANSX is capable of producing transport tables for chargedparticle transport and coupled sets for more than two particles. The main difference between charged-particle cross sections and neutral-particle cross sections occurs for elastic scattering, because the Rutherford cross section becomes singular as the scattering angle goes to zero. In practice, this singularity is screened out by the electronic structure of the material. This is no longer a strictly nuclear problem, and the forward transport of charged particles is normally handled as a continuous slowing down calculation using stopping powers. NJOY constructs charged particle elastic matrices that include scattering to cosines less than 0.96 only (approximately 20°).

Coupled transport is natural for charged particle problems. High energy neutrons normally produce secondary protons, and these protons can produce additional secondary neutrons. The structure of a coupled set for neutron-proton-photon transport is shown in Fig. 2. Note that a (p, n) reaction looks like an upscatter event in this type of table. The user must be careful to use NUP>O, and the transport calculation may run more slowly because it will have to use outer iterations.

F. The Adjoint Problem

Solutions to the adjoint transport equation are useful for calculating the change in the response of a system to a small perturbation and for calculating the source that will give a desired response. Adjoint fluxes are being used more and more frequently to compute the sensitivity of a response to changes in the basic cross sections.

The adjoint transport equation can be obtained from the direct or forward equation discussed above by transposing the scattering and fission matrices. However, this converts the downscatter calculation into an upscatter calculation. If the order of the groups is then reversed, the cross sections can be used with the regular transport equation; the solution for the direction $\vec{\Omega}$ is just the adjoint flux for direction $-\vec{\Omega}$ numbered backwards.

The existing S_N codes perform this operation of transposition and reordering of the groups, but only if the entire cross-section table fits into core storage at once. TRANSX can produce an adjoint set in group ordering, thereby allowing fine-group adjoint solutions to be obtained. When TRANSX adjoint tables are used, the S_N code must be run in forward mode, and the adjoint flux is obtained from the S_N flux solution $\sigma(g, \vec{\Omega})$ using

$$\phi^+(g,\vec{\Omega}) = \phi(N-g+1,-\vec{\Omega})$$
 (22)

G. Self-Shielding

Eq. (1) is a multigroup approximation to the Boltzmann equation, and the energy texture is necessarily coarse with respect to the resonance structure in many materials. Fig. 3 compares the pointwise and groupwise representations of



Figure 2: Layout of a coupled table for the simultaneous transport of neutrons, protons, and gamma rays. Normally, only the lower triangle of the "p to a" block would contain values for the upscatter portion of the table (the part above the line in the middle). The "group" index increases from left to right, and the "position" index increases from top to bottom.



Figure 3: Comparison of pointwise and multigroup representations of ²³⁸U capture. The solid histogram shows the effective multigroup cross sections for infinite dilution, and the dashed histogram shows the effective cross sections for a background cross section of 30 barns/atom. This is the 69-group structure used in the MATXS7 library.

the first three capture resonances of ²³⁸U. A resonance material in a dilute mixture or in small pieces does not disturb a smooth flux very much by its presence—this is called the infinitely dilute case—but more resonance material will cause sharp dips in the flux corresponding to each resonance. The reaction rate $\sigma\phi$ will be reduced. The resonance is said to be "self-shielded."

The magnitude of this self-shielding effect is in general a complicated function of the geometry and composition of the system. However, it has been found that a simple model called the Bondarenko Method¹⁶ or Background Cross Section Method¹⁷ does a surprisingly good job of representing the effects for many applications. The flux is assumed to vary inversity as the total macroscopic cross section. The model flux for the group averages for isotope *i* is written as

$$\phi_{\ell}^{i}(E) = \frac{C(E)}{[\sigma_{0}^{i} + \sigma_{\ell}^{i}(E)]^{\ell+1}} , \qquad (23)$$

where C(E) is the smooth part of the shape of the flux, and σ_0^i is called the background cross section (it represents the effects of all the other isotopes). The effect of the total cross section in the denominator is to put a dip in the flux for each peak in the cross section, and σ_0^i controls the relative size of the dip $(\sigma_0^i = \infty \text{ gives the infinitely dilute case discussed above})$. The ℓ dependence shown

here is appropriate for a large system with nearly isotropic scattering (i.e., the B_0 approximation), and it was used when the MATXS files were generated. The result in Eq. (23) is based on the narrow resonance approximation, and it will be less accurate for some of the wider low-energy resonances important in thermal systems. The multigroup form of this model i.ux is

$$\phi_{lg}^{i} = \frac{C_{g}}{[\sigma_{0g}^{i} + \sigma_{lg}^{i}]^{l+1}} \quad . \tag{24}$$

For an infinite homogeneous mixture, the appropriate background cross section is

$$\sigma_{0g}^{i} = \frac{1}{N_{i}} \sum_{j \neq i} N_{j} \sigma_{ig}^{j}(\sigma_{0g}^{j}) , \qquad (25)$$

where N_i is the number density for the isotope. Because σ_i depends on σ_0 , finding σ_0 is clearly an iterative process.

For a lump of resonance material embedded in a large moderating region, escapes from the lump also increase the background cross section. This additional escape cross section is given by

$$\sigma_{xg}^{i} = \frac{1}{N_{i}\overline{\ell}} \quad , \tag{26}$$

where Z is the mean chord length of the lump given by

$$l = \frac{4V}{S} \quad , \tag{27}$$

and where V and S are the volume and surface area of the lump (e.g., for a cylinder, $\overline{\ell}=2r$). The mean chord $\overline{\ell}$ can be adjusted away from the geometric value of Eq. (27) to compensate for the presence of other lumps (Dancoff corrections) or for shortcoming in the escape probability model used to obtain Eq. (26) (Bell factors); the main point is that TRANSX option provides an additional component for the background cross section that is parameterized by $\overline{\ell}$.

Eqs. (25) and (26) show that there is some kind of equivalence between the self-shielding effects of an admixed moderator and the self-shielding effects of heterogeneity. Physically, in the narrow-resonance regime, if a neutron is produced at the energy of an absorption resonance of material *i*, either losing energy by scattering from one of the competing materials *j* or escaping from the lump will assure that it will not be absorbed by that resonance. In fact, the effects are additive. As

an example, consider a 1.cm rod of uranium oxide surrounded by a large region of moderator:

$$\sigma_0 = \frac{.046 \text{ at/b-cm} \times 3.76 \text{ b/at}}{.023 \text{ at/b-cm}} + \frac{1 \text{ cm}}{.023 \text{ at/b-cm}}$$

= 7.52 b/at + 43.48 b/at
= 51.0 b/at

where .023 and .046 are the atomic densities of the 238 U and oxygen, respectively, and 3.76 barns is the oxygen cross section.

H. Daacoff Corrections

If the lumps, rods, or plates containing the resonance material are arranged in a lattice or stack, it is likely that a neutron starting from one lump could pass through the matrix material and react with the fuel in another lump. Such an event is equivalent to never having escaped from the first lump at all; therefore, the escape cross section is reduced. This is called a Dancoff correction, and several options are available in TRANSX:

Cylinder by the Bell approximation,

$$C = \frac{1}{1+\lambda} \quad ; \tag{28}$$

Cylinder In an hexagonal lattice by the Sauer approximation,

$$C = \frac{e^{-\tau\lambda}}{1+(1-\tau)\lambda} \quad ; \tag{29}$$

where

$$\tau = \left(.9523\sqrt{1 + \frac{V_M}{V_M + V_C}} - 1\right)\frac{\sqrt{1 + V_C}}{V_M} - .12\left(1 + \frac{1}{2}\sqrt{\frac{V_C}{1 + V_C}}\right) \quad , \quad (30)$$

Cylinder in a square lattice by the Sauer approximation,

$$C = \frac{e^{-\tau\lambda}}{1+(1-\tau)\lambda} \quad ; \tag{31}$$

where

$$\tau = \left(.8862\sqrt{1 + \frac{V_M}{V_M + V_C}} - 1\right)\frac{\sqrt{1 + V_C}}{V_M} - .08\left(1 + \frac{1}{2}\sqrt{\frac{V_C}{1 + V_C}}\right) \quad , \quad (32)$$

15

Slab cell by the Bell approximation,

$$C = \frac{0.5}{1 + \lambda_R} + \frac{0.5}{1 + \lambda_L} \quad (33)$$

and

Slab cell by E₃ collision probabilities,

$$C = E_3(\lambda_R) + E_3(\lambda_L) \quad . \tag{34}$$

In these expressions, λ stands for the optical path in the moderator region, $\bar{l}_m \sigma_m$, λ_L and λ_R are the optical paths to the left and right from the region containing material *i* to the next slab containing the same material, V_M and V_C are the ratios of the moderator volume and clad volume to the fuel volume, and E₃ is the third-order elliptic integral. In all the cases, the Dancoff correction results in an escape cross section of

$$\sigma_{xg}^{i} = \frac{1}{N_{i}\tilde{\ell}} \frac{b_{1}(1-C)}{1+(b_{2}-1)C} , \qquad (35)$$

where b_1 and b_2 are called Bell corrections (appropriate values are 1.09 for slabs and 1.35 for cylinders). When $b_1=b_2$, the constant is usually called A, the Levine factor. TRANSX provides for the automatic calculation of fuel and moderator mean chords and homogenized moderator cross sections, given the volumes and compositions of the various regions of a cell. Examples are given later in this report.

I. Interpolation in Cross Section Tables

The self-shielded cross sections for the MATXS libraries are computed in the GROUPR module of the NJOY system using Eqs. (3) and (4) together with the flux model of Eq. (23) and, optionally, using the flux calculator, for a preselected set of temperatures and background cross sections σ_0 . To obtain a cross section at some other temperature and background cross section, it is necessary to interpolate between the values in the library.

The variations of the cross sections with temperature are normally fairly simple. Earlier versions of TRANSX used a N-th order Lagrangian interpolation scheme on $\sigma(T)$ and T, where T is the temperature and N is the number of values in the temperature grid. However, this scheme would sometimes give unphysical or negative cross sections due to oscillations in the interpolation function. This version of TRANSN reverts to using simple linear interpolation for its stability.

For σ_0 , the shielding curve has a shape similar to the tanh function (see Kidman).¹⁸ Like tanh, it has a limited radius of convergence, and no single polynomial can be used to represent it for all σ_0 . Earlier versions of TRANSX used a combination of $1/\sigma_0$, Lagrangian, and quadratic interpolation for different σ_6 ranges. Just as for temperature interpolation, this scheme sometimes led to large unphysical oscillations in the interpolated cross sections. Therefore, this version of TRANSX uses simple linear interpolation based on $\log(\sigma_0)$, except for very large σ_0 values, where it uses $1/\sigma_0$, interpolation. The lowest cross-section value is used for σ_0 less than the lowest grid point in the self-shielding table. This approach is stable and reliable, but it requires that data be given on a finer σ_0 grid than the older method for equivalent accuracy. Note also that linear interpolation preserves the condition that the total cross section be equal to the sum of its parts.

J. Elastic Scattering Corrections

In order to make a library of multigroup constants, as defined by Eqs. (3) and (4), it is necessary to choose a model-weighting flux. In an actual system, the real flux may be quite different from this model. Because the flux will usually vary from region to region, the mismatch between the actual and model flux cannot always be removed by improving the library flux. For the vector cross sections (e.g., capture, fission), the size of a group can usually be made small enough so that the weighting flux difference is not too important. However, the elastic downscatter for the heavier materials always takes place from a range of energies very close to the bottom of the group. If r is the ratio of the flux near the bottom of a group to the group-average flux, the elastic removal cross section is very sensitive to any difference between r calculated for the actual flux and r calculated for the model flux.

Starting with the 1DX code,¹⁹ various attempts have been made^{20, 21} to correct the elastic removal based on the shape of the computed flux using an iterative scheme. However, the complicated consequences of resonances in the cross sections make this dangerous except in regions where the calculated flux varies smoothly. Therefore, TRANSX does not attempt to correct for variations that only show up in one or two groups. Instead it smoothes the flux by averaging over several groups above and below the group in question. This results in an effective slope for the flux near the bottom of the group, which can be used to correct the removal as shown below. The corrected cross-section set should give improved answers for any responses that depend on the average removal effect such as k_{eff} , but the actual flux shape near a particular resonance may not be improved at all.

The removal from group g to group g' can be written

$$R_{g' \leftarrow g}^{M} = \int_{g} du \int_{g'} du' \sigma_{\epsilon}(u' \leftarrow u) \phi^{M}(u) \quad , \tag{36}$$

where u is lethargy, σ_e is the elastic scattering differential cross section, and ϕ^M is the model flux. If the actual flux at the bottom of the group has a different slope than the model flux,

$$\phi(u) = [a - b(u_g - u)]\phi^M(u) , \qquad (37)$$

where u_g is the lethargy at the bottom of group g, the removal rate becomes

$$R_{g' \leftarrow g} = R_{g' \leftarrow g} [a - b\gamma_{g' \leftarrow g}] \quad , \tag{38}$$

where

$$\gamma_{g' \cdots g} = \frac{\int_{g} du \int_{\gamma'} du' (u_{g} - u) \sigma_{\epsilon}(u' \leftarrow u) \phi^{M}(u)}{\int_{g} du \int_{g'} du' \sigma_{\epsilon}(u' \leftarrow u) \phi^{M}(u)} \qquad (39)$$

The quantity $\gamma_g = \sum_g \gamma_{g' \leftarrow g}$ is available on the MATXS library, so this γ is used for all groups and Legendre orders to obtain the corrections to the elastic scattering matrix elements.

It is then assumed that the smoothed flux ratio can be represented by

$$\ln(\phi/\phi^M) = x_0 + x_1(u_g - u) + x_2(u_g - u)^2 , \qquad (40)$$

in the group g. The coefficients are computed using a least squares fit to six values of the ratio assigned to the lethargy center of each of the six groups (three above u_g and three below). The parameters a and b are determined at the bottom of the group, and the smoothed value of the ratio at the center of the group is used to find the change in cross section that will give the desired change in removal rate.

The method is very stable. It avoids the divergences in removal cross sections often sections a method shand it usually converges adequately in one iteration.

K. Group Collapse

It is necessary to produce cross-section libraries with fairly narrow energy groups to minimize the consequences of an inappropriate model flux discussed above. However, it may not be economical (or even possible) to perform a multidimensional S_N solution using the fine-group structure directly. In such cases it is desirable to collapse to a coarser group structure. The collapse can be defined in such a way as to preserve each term of the P_N equations when they are summed over all fine groups g in coarse group G as follows:

Coarse group flux

$$\phi_{\ell G} = \sum_{g \in G} \phi_{\ell g} \quad , \tag{41}$$

Absorption, $\bar{\nu}\sigma_f$, and edit cross sections

$$\sigma_{zG} = \sum_{g \in G} \sigma_{zg} \frac{\phi_{0g}}{\phi_{0G}} \quad , \tag{42}$$

Transport cross section for leakage calculation

$$\sigma_{irG}^{PN} = \sum_{g \in G} \sigma_{irg}^{PN} \frac{\phi_{1g}}{\phi_{1G}} , \qquad (43)$$

Pe-weighted total cross section

$$\sigma_{\ell \iota G}^{PN} = \sum_{g \in G} \sigma_{\ell \iota g}^{PN} \frac{\phi_{\ell g}}{\phi_{\ell G}} \quad . \tag{44}$$

Scattering matrix cross sections

$$\sigma_{\ell G \leftarrow G'}^{PN} = \sum_{g \in G} \sum_{g' \in G'} \sigma_{\ell g' \leftarrow g}^{PN} \frac{\phi_{\ell g'}}{\phi_{\ell G'}} , \qquad (45)$$

and

Steady-state and delay fission spectrum χ

$$\chi_G = \sum_{g \in G} \chi_g \quad . \tag{46}$$

The coarse-group S_N equations can be derived from the coarse-group P_N equations; therefore, Eqs. (7) through (13) apply with g replaced by G.

19

The flux components can be read in from some previous calculation or generated from the model weight function used to generate the MATXS library with

$$\phi_{\ell g} = \frac{C_g}{\prod_{k=0}^{\ell} [\sigma_{\upsilon g} + \sigma_{k t g}]} \quad . \tag{47}$$

In practice only k=0 and k=1 are available on the MATXS library, and k=1 is used for all higher orders.

Note that there are four different uses for input fluxes in TRANSX: (1) collapse, (2) fission χ , (3) transport cross section and (4) elastic scattering corrections.

L. Spatial Collapse

It is often necessary to replace the complex geometry of a reactor cell with some effective average cross section before doing two- or three-dimensional calculations for an entire reactor corp. This spatial collapse is similar to the group collapse described above in that both try to preserve the terms of the P_N equation. Using v_i for the region volumes and V for the cell volume gives the following results:

Cell-average flux

$$\phi_{\ell g}(V) = \sum_{i \in \text{cell}} \frac{\phi_{\ell g}(v_i) v_i}{V} , \qquad (48)$$

and

Cell-average macroscopic cross section

$$\Sigma_{\ell g}(V) = \sum_{i \in crill} \sum_{j} \frac{\rho_j(v_i) \sigma_{\ell g}^j(v_i) \phi_{\ell g}(v_i) v_i}{\phi_{\ell g}(V) V} \quad .$$
(49)

In these equations, the v_i are the region volumes, V is the cell volume, $\rho_j(v_i)$ is the number density for material j in region i, and $\sigma_{\ell g}^j(v_i)$ is the microscopic cross section for material j in region i.

These calculations of cell-averaged quantities require a good knowledge of the region fluxes by group. These fluxes can be obtained in the form of a CCCC standard-interface file from a separate transport calculation and read into TRANSX. A typical procedure is to do a simple one-dimensional calculation for a reactor pin cell, and then to use the flux from this calculation to prepare cell-averaged cross sections to be used for a reactor-core calculation. An example of this process will be found in the next section of this report.

It is often necessary to obtain a "constituent" of a mixture like this, for example, the microscopic cross section for the ²³⁵U in the uranium oxide in the fuel region of a reactor pin. Other times, the user may need the microscopic cross section averaged over the cell. TRANSX calls the first type of constituent cross section a "region constituent" and the second type a "cell constituent."

Region constituents are useful when the transport calculation is going to be done using the detailed region-wise geometry of the cell and when the user needs to make the mixtures in the transport code (c.g., to do concentration searches). Real material densities must be used with region constituents.

Cell constituents include the effects of any flux depression ("disadvantage factors") or flux enhancement in the regions of the cell. Therefore, they are needed when the transport calculation is going to be made using a simplified geometry and when the user needs to make the mixtures in the transport code. Cell-averaged densities must be used with cell constituents.

M. Photon Production Self-Shielding

Photons are produced by capture and fission, and both reactions usually show significant self-shielding in resonance energy ranges. Although the MATXS format has the capability of storing shielded cross sections for any photon production cross section, the current libraries contain only infinite-dilution photon matrices, and TRANSX uses shielding factors for the corresponding neutron reaction to self-shield each photon production matrix.

N. Thermal Cross Sections

When the energy of a neutron is comparable with the energy of thermal motion of the atoms in a material, the neutron can gain energy in a collision, or "upscatter." Most S_N codes can handle upscatter at some increase in run time. The location of the upscatter groups in the transport table is defined in Table 1.

The detailed thermal cross sections and scattering matrices depend upon the binding between the atoms of the material.²² Therefore, MATXS thermal libraries will contain several sets for a particular isotope, for example, H in H₂O, H in polyethylene, and H in ZrH. In addition, thermal scattering can have both coherent and incoherent components. Coherent scattering takes place with no energy change (in-group only) and includes the diffraction effects of the neutron wave interacting with the entire crystalline lattice. The incoherent part contains both energy-gain and energy-loss cross sections due to interactions with lattice vibrations or

vibrations or molecular rotation and vibration modes. For the heavier materials, it is often a good approximation to treat the atoms as a gas of free particles at the appropriate temperature. Free-gas cross sections and matrices are included on MATXS thermal files for all materials in addition to any bound cross sections given for important moderator materials.

Thermal cross sections are defined over a limited energy range (e.g., E < 4 eV). To prepare an upscatter set, TRANSX corrects the total cross section by subtracting the normal static elastic cross section (NELAS) and adding back the appropriate incoherent and coherent thermal cross sections over the requested energy range. This step ensures that any absorption reactions are included. For the scattering matrix, TRANSX uses the static matrix above the specified energy range and the requested incoherent and coherent matrices below the energy cutoff. Sources from above the cutoff to below use elements of the static matrix. Sources from below the cutoff to above are truncated into the last group of the thermal range so that there is no upscatter for groups above the cutoff.

Some ENDF/B thermal evaluations give the scattering from each atom in the molecule, and the different atoms must be combined in TRANSX. An example is water; here "H in 'I₂O" must be combined with free gas oxygen. Other evaluations (e.g., methane) give the entire scattering for the molecule with the principal scatterer (everything is renormalized to the cross section of the principal scatterer). In these cases, TRANSX must mix in the thermal scattering for the principal scatterer and omit the contribution for the secondary atom. Methods for doing this are discussed in the next section.

O. Epithermal Self-Shielding

The biggest weakness of the TRANSX code as described so far occurs for thermal reactor analysis in the "epithermal" energy region from about 4 eV to around 200 or 300 eV. In this region, the resonances are no longer "narrow," and the flux shape given by Eq. (23) is no longer sufficiently accurate. This failure is normally only important for resonance materials that are present in large concentration, such as ²³⁸U. TRANSX results for this energy range can be improved by preparing special cross sections for the ²³⁸U using the NJOY flux calculator. As an example, you could prepare a new isotope called U238B with cross sections appropriate for ²³⁸UO₂ in pins of a certain size surrounded by a water region. You could then add this isotope to your MATXS library using BBC. By asking for U238B instead of U238 in your TRANSX run, you would get improved results for resonance self-shielding in the epithermal region.

III. HOW TO RUN TRANSX

TRANSX is capable of solving complex problems, and the input is necessarily complex. However, an attempt has been made to simplify the input by providing defaults so that the user interested only in the simpler problems can ignore some of the complexities. This section is organized from the simple to the complex; a user can stop reading when he knows enough to solve his problem.

The input instructions are given in the following subsection. They are explained in the subsequent subsections by using examples. The examples also act as test problems for the code, and some partial listings are included in Appendix A. Input decks and full output listings for all of these sample problems are included in the TRANSX 2.0 distribution package.

A. TRANSX Input Instructions

The following user input instructions for TRANSX were copied from the comment cards at the beginning of the TRANSX source deck. It is always a good idea to check these cards in the current version of the code, because they are more current than the version printed here.

User II	PUT (FREE FORM)
CARD 1	RUN TITLE (HOLLERITH DESCRIPTIVE CONNENT) 72 CHARS MAX. NO DELIMITERS OR TERMINATOR REQUIRED. STOP TERMINATES JOB
CARD 2	OPTIOES
IPRINT	PRINT (O=LONG/1=SEORT)
IOUT	OUTPUT FORMAT
	o none
	1 CARD (6E12)
	2 CLAW (CARD WITH SEPARATE EDITS)
	3 FIDO
	4 ANISH (FIDO WITH 2+L+1 PACTOR)
	5 GOIS (GROUP-ORDERED BINARY)
	6 ISOTIS (CCCC-IV MATERIAL-ORDERED BINARY)
IPROB	PROBLEN (O=DIRECT/1=ADJOINT)
ISET	ISET=I FOR A SIMPLE SET FOR PARTICLE I,
	ISET=IJ FOR A COUPLED SET FOR PARTICLES I AND J,
	ISET=IJX FOR A THREE-WAY COUPLED SET, ETC.
IFORM	FORN (1=NATWISE/2=GROUPWISE)
ITINE	FISSION TIME (1=STEADY-STATE/2=PROMPT)
IDECAY	DECAY TIME (RESERVED)
ITRC	TRANSPORT CORRECTION
	O NONE (NOT FOR SN CODES)
	1 CONSISTENT-P
	2 DIAGOBAL

BELL-HANSEN-SANDNEIER

4 ! : LOV

IN LAL FLUX

b Consistent-P

```
ICOLL CONTRACTOR (O=NO/NFINE=YES)
```

18177

- O USE LIBRARY WEIGHT FOR PO FLUX
- 1 READ PO FLUX ON CARDS
- 2 READ RTFLUX INTERFACE (CCCC-IV TOTAL PLUX)
- 3 READ RZMFLI INTERFACE (ZONE MONENTS FLUI)
- 4 READ RZFLUX INTERFACE (CCCC-IV ZONE FLUX)
- 12-14 TRIGGERS REMAP OF IMPUT FLUX
- -N MAKE ELASTIC SCATTERING PLUX CORRECTION

(FOR LAST FIVE ITEMS, DEFAULT=FIRST OPTION)

- CARD 3 PARAMETERS
- NGROUP NUMBER OF GROUPS IN TABLE
- NL NUMBER OF LEGENDRE TABLES
- NTABL TABLE LENGTH
- NUP NUMBER OF UPSCATTER GROUPS
- NTHG NUMBER OF THERMAL GROUPS
- NNIX NUMBER OF OUTPUT MIXES OR MATERIALS
- NREG NUMBER OF REGIONS
- NMIXS NUMBER OF MIX SPECIFICATIONS
- NED NUMBER OF EXTRA EDITS
- NEDS NUMBER OF EDIT SPECIFICATIONS
- CARD 4 MIX NAMES
- HMIX UNIX HOLLERITH MIX NAMES (6 CHARACTERS MAX)

CARD 5 REGION AND HETEROGENEITY SPECS

EREGN REGION BOLLERITH NAME (6 CHARACTERS MAXIMUM)

- RTEMP REGION TEMPERATURE (K) (DEFAULT=300)
- **RVOL** REGION VOLUME (DEFAULT=1.)

IHET BETEROGENEITY OPTION (DEFAULT=0=NO SELF SHIELDING)

- 0 NO SELF SHIELDING
- 1 CONSTANT ESCAPE CROSS SECTION
- 2 CYLINDER BY BELL APPROXIMATION
- 3 CYLINDER IN HEX LATTICE BY SAUER APPROXIMATION
- 4 CYLINDER IN SQUARE LATTICE BY SAUER APPROXIMATION
- 5 SLAB CELL BY BELL APPROXIMATION (REFLECTIVE)
- 6 SLAB CELL BY E3 APPROXIMATION (REPLECTIVE)
- 7 SLAB CELL BY BELL APPROXIMATION (PERIODIC)
- 8 SLAB CELL BY E3 APPROXIMATION (PERIODIC)
- -N CONTINUE SPECIFYING THE CELL

THE REST OF THE PARAMETERS ON CARD 5 DEPEND ON THE VALUE OF IMET. THESE VALUES ARE ONLY GIVEN FOR THE FIRST REGION OF BACH CELL.

-- IHET=1 -- CONSTANT ESCAPE

CHORD NEAN CHORD LENGTH (CN)

-- IBET=2 THRU 8 -- DANCOFF APPROXIMATIONS

BELLI BELL CORRECTION TO RATIONAL APPROX. BELL2 BELL CORRECTION TO DEBONIWATOR

REPEAT CARD 5 BREG TIMES

- CARD 6 REMAP INSTRUCTIONS (ABS(INITF).GE.12 ONLY) PLUX POINT, ZONE, OR REGION FROM INPUT FILE TO BE ASSIGNED TO EACH TRANSX REGION (NREG INTEGERS)
- CARD 7 NIX SPECIFICATIONS INIX MIX NUMBER (O TO NMIX) IREG **REGION NUMBER** ENIX NAME OF ISOTOPE FROM MATES DENS MULTIPLIER (FRACTION OR DENSITY) (DEFAULT=1) BINC THERMAL INELASTIC NAME (DEFAULT=BLANK). OR CONSTITUENT CROSS SECTION FLAG ("RC" FOR RZGION CONSTITUENT, OR "CC" FOR CHLL) HCOH THERMAL ELASTIC NAME (DEFAULT=BLANK) NATRIX NOD REACTION (DEFAULT=BLANK=TCTAL) EXOD

REPEAT CARD 7 MMIRS TIMES

CARD 8 EDIT NAMES

HED NED HOLLERITH EDIT NAMES (6 CHARACTER MAXIMUM)

CARD 9 EDIT SPECIFICATIONS

JEDPOS EDIT POSITION (USUALLY IF RANGE FROM 1 TO NED)

- (-N TO PUT DELAYED GANNAS IN GROUP N OF COUPLED SET)
- HEDNAM NAME OF REACTION FROM MATIS
- EDFACT MULTIPLIER (DEFAULT=1)

HEDNAT ISOTOPE HAME FROM MATES (DEFAULT-BLANK)

IF BLANK, USE SPECIFICATION FOR ALL ISOTOPES

REPEAT CARD 9 NEDS TIMES

CARD 10 COLLAPSE SPECIFICATIONS (IF ICOLL.NE.O) INPUT NUMBER OF FINE GROUPS IN EACH COARSE GROUP

CARD 11 INPUT FLUX FUNCTION (INITF=1 ONLY) INVT FLUX FOR FINE GROUPS TO BE USED FOR ALL ORDERS AND ALL REGIONS

SPECIAL INPUT VALUES

HED=+CHI+	THIS POSITION TO CONTAIN FISSION CHI
NEDNAN=+CHI+	DITTO
HED=+TRD+	THIS POSITION TO CONTAIN TRANSPORT
	CROSS SECTION FOR DIFFUSION. IF A P1 FLUX
	IS GIVEN OR COMPUTED, THE INFLO DEFINITION
	WILL BE USED. OTHERWISE, OUTFLO WILL BE USED.
HEDHAM=+TRD+	DITTO
BED=+GAN+	THIS POSITION TO CONTAIN THE FIRST OF HGG GANNA

PRODUCTION EDITS

B. Free Form Input

All input to TRANSX is free form with no limitation to fixed fields. Numbers do not require decimal points, but the "E" is required when exponents are given. Hollerith strings are delimited with stars (*) or preceded by "nH" as in FORTRAN formats. Delimiters are not necessary if the string is a single word (*i.e.*, six or fewer characters) and starts with a letter. The construction "nR" causes the following entry to be repeated n times. Entries can be separated by any character not used for another purpose. A comma works, but spaces look better. Lines can be terminated early with /, and any input values not given are defaulted as defined in the input instructions. Any text appearing to the right of a / (other than a / embedded in a Hollerith string) will be treated as a comment. Lines beginning with a \$ sign will also be treated as comments. Care must be used to include the / when needed; otherwise, the input routine will keep on reading cards until it has found the requested number of entries or reaches the end of the file. The proper input syntax should be made clear by the examples below.

TRANSX input card 1 is not read by the free-format input processor; it is read by a simple FORTRAN statement. Therefore, stars are not required. The "stop" at the end of a TRANSX input deck is a special case of card 1, and stars are not required there either (although they can be given).

C. Simple Transport Tables

The following examples illustrate simple microscopic and macroscopic crosssection sets without self-shielding or thermal upscatter. Simple response edits, group ordering, and collapse are shown also.

Sample Problem 1. Simple Material-Ordered Tables

```
TEST 1 -- MATERIAL ORDERED TABLES
   1
     0 1 1
              1 0 3 0 0
30 2 34 6 0 3 1 6 1 1
CARBON TUNGST U-235
• •/
     CBAT/
1
  1
2
     ¥182
          .263/
           . 143/
2
  1
     V183
           .307/
2 1
     V184
          .286/
2
  1
     V186
3 1 0235/
```

HEAT 1 HEAT 1.602E-19/ CONVERTED TO JOULE-BARNS STOP

The cross sections will be read from a file named MATXS (see Section V), and the neutron tables will be printed and written onto a file named CARD as 6E12 cards (see Section IV for a complete description of the interface files) in material ordering. Fission will be steady state, and the BIIS transport correction will be used. Because two tables are requested (P_0 and P_1), the BIIS correction will be derived from the P2 data on the library. The table length of 34 is enough to accommodate all 30 groups, the three standard edits (σ_{s} , $\dot{\nu}\sigma_{f}$, σ_{t}), and one special addit. There are no apscatter positions or thermal groups requested. Three output materials are defined with the names given. Note that these names are independent of the isotopes on the library and will only be used to label the output and to name the materials on the output file CARD. The materials lie in a region with temperature 300 K (by default) and a blank name, but "region" has no other senificance here in the absence of an input flux or self-shielding. The six mix instructions use default values for simplicity. This table works as follows: The material HMIX is multiplied by DENS (default=1.) and added into mixture INIX. The names used for HNIX must correspond exactly to material names on the MATXS library. The use of atom fractions for DENS in mix 2 produces the correct microscopic cross sections for natural tungsten. Later examples will show how to generate mixed macroscopic cross sections. The response edit "KEAT" is intended to give nuclear heating in watts/cm³ when folded with a flux in a transport code. The table works just like the mix table. The cross section HEDHAH from the MATXS library is multiplied by EDFACT and added into position JEDPOS. The EDFACT shown here converts KERMA factors from eV-barns to Joule-barns. Just as for material names, the reaction names (e.g., HEAT) for HEDNAN must correspond to names on the library, but the edit name (e.g., HEAT) is only used for labeling the output. The last card is read as the title for the next case, but "STOP" has the special meaning of ending the run. The meaning of these input parameters may be clarified by the partial listings of the OUTPUT and CARD files in Appendix A.

Sample Problem 2. Group Ordering

This is the same as Problem 1 except for group-ordered printed output. No library is written because no card image format is available for IFORM=2. See Appendix A.

```
Sample 1'r: blem 3. C. pled Set

TEST 3 - COUPLED SET

0 3 0 3 1 1 0 3 0 0

42 2 46 0 0 1 1 2 1 2

AL-27

• •/

1 1 AL27/

HEAT

1 HEAT/ BA AND B: PART

1 GHEAT/ GG PART

STOP
```

This time output in FIDO format was requested. As described in Section IV-E, FIDO suppresses many of the zeroes in the output file. The coupled option was selected, and 42 had to be given for the number of groups (MATXS5 is a 30 x 12 library). The nu, ng, and gg blocks for AL27 are automatically moved to the correct part of the coupled set. The photoatomic data are really elemental in nature, but the MATXS file now contains duplicate tables for each isotope, for convenience, and it is no longer necessary to name both AL27 and AL in TRANSX input decks. The edit cross sections GREAT and HEAT are also automatically moved to their proper positions in the coupled set: the former ends up as position 1 for groups 31 to 42, the latter in groups 1 to 30. A partial listing of the FIDO tables is given in Appendix A.

Sample Problem 4. Cross Sections for GODIVA

```
TEST 4 -- GODIVA
0 1 0 1 1 1 0
                  3 30 1
20 4 34 0 6 1 1 3 1 0
CORE
CORE/
1 1 U23, .0450/
1 1 0238 .002498/
1 1 0234 .000492/
CEI
19R1 11/ COLLAPSE OUT BOTTON 11 GROUPS
6.80E+8 1.72E-7 5.78E-7 3.30E-6 1.88E-5 5.92E-5 3.52E-4 3.21E-4
4.00E-4 4.43E-4 4.61E-4 8.79E-4 7.74E-4 5.90E-4 3.53E-4 2.83E-4
6.30E-5 9.77E-6 27E-6 1.62E-7 1.87E-8 2.23E-9 3.41E-10 3.56E-11
5.38E-12 2.29E 12 9.15E-13 1.30E-15 7.63E-19 1.59E-22/
STOP
```

GODIVA is a bare sphere of enriched uranium that is often used as a cross-section benchmark.²³ In this example, macroscopic P_3 cross sections are produced directly. As a result, it is not necessary to mix cross sections in the transport code. In addition to saving time and storage space, this approach allows a mixture-dependent fission χ to be formed. Because the spectrum of GODIVA is rather hard, the lower energy groups are collapsed down to fewer groups (note the use of the repeat specification). An initial flux guess is read in from the input file to improve the estimate for the fission spectrum χ and to improve the group collapse. Note that ICOL=30, the number of fine groups, and NGROUP=20, the number of coarse groups for the output material. CHI is a special edit command that does not require any further specification.

D. Advanced Response-Function Edits

One of the powers of TRANSX is its ability to produce varied response-function edit cross sections. The ability arises partly out of TRANSX, which can produce any edit that is a linear combination of any cross sections on the MATXS file, and partly out of MATXS, which contains all of the partial cross sections available in the original ENDF/B evaluations as well as some additional quantities, such as nuclear heating and radiation damage cross sections.

As an example of the power of linear combination, consider the problem of computing the helium production rate in graphite. The ¹²C evaluation in ENDF/B-IV includes an (n, α) reaction (called NA) and a continuum $(n, n')3\alpha$ reaction (NCNAAA). An edit for helium production rate (arbitrarily named N.HE4) can be obtained using the following edit specifications:

```
N.HE4
1 NA/
1 NCHAAA 3./
```

The BA reaction is multiplied by 1 (default) and added into position 1, and the NCWAAA reaction is multiplied by 3 and added into position 1. Caution: A different representation is used in ENDF/B-V and VI.

Sometimes it is useful to define special edits for particular materials. As an example, the ratios of ²³⁵U capture to ²³⁵U fission and ²³⁸U fission to ²³⁸U fission were measured at the center of the GODIVA critical assembly.²³ The following edit specifications added to Problem 4 will provide the responses needed to compute these reaction rate ratios:

CE	I F25	F 28	C25
2	NPTOT	1.	U235
3	NPTOT	1.	V238
4	XG	1.	U235
	••		

Note that the standard transport tables reserve NED words for each group, order, and material for edits that are not required during the flux calculation but only at the end of the problem when responses are computed. Also, the edits only appear in the P_0 table, so the words reserved in the higher tables are wasted. Storage requirements can be reduced by defining a special edit table with NTABL=NED, which is only read during the response function edit calculation in the S_N code. The ONEDANT⁴ code uses this approach.

Sample Problem 5. Detailed Edit Table

```
TEST 5 -- EDIT TABLE
0
   5 0 1 2 1 0 0
                       0
                         0
30 1 13 0 0 1 1 1 13 37
AL-27
AL-27/
   1 AL27/
1
ELAS NNG N2N N3N NG NA
HP
    ND NT NNP NNA
N.1H
      N.48E
1
  HELAS/
2
   HINEL/
2
   $75P
         -1/
2
   178P
         -1/
2
   180P
         -1/
2 N81P
         -1/
2
  1824
        -1/
2
  X83P
        -1/
2
   384P
        -1/
2
   X85P
        -1/
2
  IS6A
        -1/
  121/
3
4
  331/
5
   JG/
   XA/
6
7
  SP/
8
 D/
9
   IT/
10 175P/
10 178P/
10 N80P/
10 N81P/
10 N83P/
10
   184P/
10 ¥85P/
11 1824/
11
   1861/
12 NP/
12 175P/
12 178P/
12 180P/
12 N83P/
```

12 884P/ 12 885P/ 13 84/ 13 882A/ 13 886A/ STOP

The ²⁷Al evaluation includes some discrete-inelastic levels that decay by proton or alpha emission rather than by a cascade of photons. The cross-sections for these reactions are subtracted from the total inelastic cross section in position 2 to obtain an isolated $(n, n')\gamma$ reaction. The parts that were subtracted are added back into positions 10 and 11 to obtain (n, n'p) and $(n, n'\alpha)$, respectively. They are also added to the (n, p) and (n, α) reactions in positions 1? and 13 to obtain the gas production cross sections for ¹H and ⁴He, respectively.

Note that the detailed aluminum edits are written into a group-ordered binary interface file with the name GOXS (see Section $IV \cdot H$). If its name is switched to SNXEDT, the ONEDANT edit module can read the file directly.

An important mechanism of radiation damage in metals is the displacement of atoms from their normal lattice positions caused by the recoil particle of a nuclear reaction. The energy available for producing displacements is given on the MATXS file as the damage energy production cross section DAME in eV-barns; it is less than the primary recoil energy because some of the energy causes electronic excitations rather than displacements. The primary recoil atom loses some of its energy ejecting another atom, giving a pair of displacements; each of these generates another pair, and so on, until all the energy is used up. The sumber of displacements produced by this cascade is equal to the damage energy divided by twice the energy required to displace one atom from its normal site. Various values of the displacement energy are used in practice; rome values²⁴ are given in Table 2, which can be used to determine the EDFACT needed to convert the damage edit DAME into a "Displacement Per Atom" (DPA) edit. The use of this factor is illustrated below.

Except for the delayed neutron parameters CHID and WUP, all the officer data on the current MATXS files are prompt data. However, many reprinted as

$$n + {}^{7}\text{Li} \rightarrow {}^{8}\text{Li} \rightarrow \beta^{-} + 2\alpha \quad (850 \text{ ms})$$

and

$$n + {}^{27}\text{Al} \rightarrow {}^{28}\text{Al} \rightarrow {}^{28}\text{Si} + \beta^{-} + \gamma \quad (2.24 \text{ m})$$

31

Material	Energy (eV)	Material	Energy (eV) 40
Be	31	Fe	
С	31	Co	40
Na	25	Ni	40
Mg	25	Cu	40
Al	27	Zr	40
Si	25	Nb	40
K	40	Мо	60
Ca	40	Ag	60
Ti	40	Та	90
V	40	W	90
Cr	-10	Au	30
Mn	40	РЪ	25

Table 2: Typical values for the atomic displacement energy needed to compute DPA (<u>D</u>isplacements <u>Per Atom</u>).

have decays that take place in times that are short with respect to the response desired. The steady-state heating due to capture in ⁷Li should really be the sum of the prompt HEAT from the MATXS file and 9.31 MeV times the capture cross section. It can easily be produced with the following TRANSX edit specifications:

```
SSBEAT

1 HEAT/ PROMPT PART

1 NG 9.31E6/ DELAYED PART FROM 9.31 NEW CAPTURE GAMMA

...
```

The photon source from neutron reactions is also an interesting response that can be obtained with the special edit name GAM. Decay photons, such as the 1.779. MeV photon from the ²⁸Al β^- decay mentioned above, can also be added into the photon source as shown in the next sample problem.

Sample Problem 6. Heat, Damage, and Photon Source

```
TEST 6 -- HEAT, DANAGE, PHOTON SOURCE

0 1 0 1 1 1 0 3 0 0

30 2 47 0 0 1 1 1 14 4

AL-27

ALUN/

1 1 AL27/

HEAY DPA GAN GAN GAN GAN GAN

GAN GAM GAN GAN GAN GAN
```

```
1 BEAT/
1 NG 1.125E6/
2 DAME .02/
11 NG/
STOP
```

The 12 occurrences of GAM referve spaces for the 12 photon groups. Group 9 (1.2 MeV) is in position 11, and the last edit specification adds the equilibrium 1.779-MeV photon production rate into that group (one such delayed gamma is produced per capture event). Note also that damage energy is converted to DPA in position 2 using 25 eV for the displacement energy $0.02 = [1/(2 \cdot 25)]$. Most S_N codes could use these cross sections and print out steady-state photon source vectors by region to be used in a subsequent photon transport calculation.

Coupled sets, like the one defined in Sample Problem 3, can also be prepared including steady-state radioactive decay contributions. Putting a negative sign on JEDPOS defines the gamma group in the coupled set that will receive the contributions defined by that edit specification. The Problem 3 input would become

```
TEST 3A -- COUPLED SET VITE STEADY-STATE DECAY
 0303110300
       46 0 0 1 1 2 1 2
 42 2
 AL-27
  • •/
 1 1 AL27/
 BEAT
    HEAT/ NN AND NG PART
 1
    GHEAT/ GG PART
 1
        1.125e6
                AL27 / DELAYED HEAT AND GANNAS FOR AL27
    IG
 1
                 AL27
-8
    IG
        1.
        3.205
                 AL.27
    ¥P
 1
        .72
                 AL27
-10
    IP
-9 MP
        . 28
                 AL27
 STOP
```

Delayed heat and photon production edit specifications for some important materials are given in the DELAY12 and DELAY24 files included in the TRANSX distribution. As an example, the 12-group data for ²³⁶U are repeated below:

1 N2N 0.060E+6 U238 0.1602+6 0238 IG 1 8.580E+6 U238 1 TP 5.995E-4 1238 -4 TP -5 4.072E-3 V238 TF T 2.036E-2 U238 -6 -7 9.2252-2 U238 ١F 3.0338-1 **J238** -2 TF -9 ¥7 1.442E+0 U238
-10	lF	3. 384E +0	U238
-11	#F	4.726E+0	U238
-12	IF	1.514E+0	U238

The heating part (position 1) includes the beta decay heating for ²³⁹U and ²³⁹Np; thus, it includes all the heat up to the appearance of the final ²³⁹Pu product. The contributions to heating and photon production from fission come from the decay of the fission products.

E. Self-Shielded Fast Reactor Problems

The sin:plest kind of self-shielding problem is one with simple homogeneous mixtures, for example, the one-dimensional model of the ZPR-6 assembly 7 critical assembly (see the ENDF benchmark descriptions).²³

```
Sample Problem 7. ZPR-6/7 Spherical Homogeneous Model
```

```
TEST 7 -- 2PR-6/7 SPE. HONOGENEOUS NODEL
0
  1
      0
         1 1 1 0 3 0
                            0
   1 50 0 0 2 2
                       20
                            1
80
                               0
CORE BLANKET
CORE 300. 1. 1 10000/ HONOGENEOUS SELFSEIELDING
                          HONOGENEOUS SELFSHIELDIEG
BLANKT 300. 1. 1 10000/
1
      PU239 8.8672E-4/
   1
      PU240
             1.1944E-4/
1
   1
1
      PU241
            1.33E-5/
   1
1
      U235
              1.268-5/
   1
1
   1
      U238
             5.780368-3/
1
   1
      NONAT 2.367E-4/
1
      1123
             9.2904E-3/
   1
1
   1
      016u
             .01398/
      FESS
1
   1
             1.2978-2/
1
      CR52
             2.7098-3/
   1
      HI55
             1.2402-3/
1
   1
1
   1
      NN 55
             2.128-4/
2
   2
      U235
             8.56E-5/
2
   2
      U238
             3.961798-2/
2
   2
      NONAT
            3.8E-6,
2
   2
      016
             2.4E-5/
2
   2
      FE56
             4 6378-3/
2
   2
      CR57
             1.2958-3/
2
   2
      1158
             5.635E-4/
   2
      N356
             9.98Z-5/
2
CII
STOP
```

The ENDF/B-VI libraries (which are used for all the test problems for TRANSX 2.0) contain the isotopes for iron, nickel, and chromium instead of the elemental

evaluations given in previous versions. The dominant isotope of each element has been used here to keep the problem input compact.

The cross sections are appropriate for diffusion theory or P₀ transport theory. Note that the table length is less than the maximum 80+3+1. The scattering matrix is truncated in a way that preserves the scattering cross section (*i.e.*, down the diagonal). In this problem, there are two regions and two mixes. The mixes are macroscopic, and the fission χ will be different in each region because the compositions are different. Self-shielding uses the constant-escape option with a temperature of 300 K. The region sizes are not needed for this problem, and they have been set to 1.0 for simplicity. The mean chord length for each region has been made very large (*i.e.*, no escape from the region) in order to obtain the mixture part of the self-shielding cnly. The background cross sections obtained are displayed on the printed output.

This homogeneous model is not completely adequate because the core and blanket regions are really made up of stacks of long slabs of the various materials arranged as repeating cells in a stainless-steel matrix. Figure 4 shows a simplified core cell. This heterogeneity has two effects; first, the background cross section for a material in a slab is changed because of the change in density and the addition of a slab escape probability; and second, the flux will be slightly different in each slab due to the sources and absorptions. The first effect is called heterogeneous self-shielding, and the second leads to the advantage and disadvantage factors for cell homogenization.

For the cell of Fig. 4, the self-shielding of ²³⁹Pu in E should be fairly well represented by the mixture effect in region E plus the escape cross section for a 0.5588 cm slab with a Dancoff correction corresponding to outside regions 4.965cm thick ($\bar{\ell}_m = 9.930$ cm) with the homogenized composition of A+B+C+D+clad on each side of the fuel slab. For the ²³⁶U in E, the outside regions extend only to B and would, therefore, be 1.626 cm thick with the C+D+clad composition on each side. The ²³⁸U in E sees different outside regions to the left (A+B+C+D+clad) and to the right (B+C+D+clad). The SS304 structures should be well represented by a 0.444 cm slab with outside regions with the A+B+C+D+E composition. Note that the mean chord would be only one-half of the infinite slab value, as $\bar{\ell}_m =$ 5.08 cm. These kinds of arguments can be completed for the other materials and regions in the cell.

This approach leads to separate two-region cell problems for each class of material (*i.e.*, all the plutonium isotopes in E from one two-region calculation, all the U isotopes in B from another), and TRANSX is capable of setting up the problem



Figure 4: Simplified core unit cell for ZPR-6 Assembly 7. Region A is the SS304 structure (0.222 cm), B is a U_3O_8 slab (0.635 cm), C is Na in an SS304 can (1.270 cm), D is an Fe_2O_3 slab (0.3175 cm), and E is a Pu/U/Mo fuel slab in a can (0.635 cm). All can thicknesses are 0.0381 cm).

in this way. However, the result is very complex, and many dummy compositions are required.

A simpler representation is the multiregion slab cell known in Fig. 5. This cell neglects the egg-crate SS304 structure and will overestimate the σ_0 values for region 1. It also neglects the effect of the structure on any paths to regions 2 through 3; thus, there will be too much sodium and not enough iron. The resulting model is given in the following sample problem.

Sample Problem 8. ZPR-6/7 Sph. Heterogeneous Model

```
TEST 8 -- ZPR-6/7 SPH. XETERO NODEL
٥
             1
                 1
                    Δ
                        3
                           0
                              0
       Δ
          1
                   2
                               1
                                  ۵
80
       50
            0
                0
                       7
                          50
COLL
       R1
     300
                   6 2R1.09/
EID
           . 2223
U203
       300
             .6350
                   -6/
    300
          1.270
TA
                 -6/
       300
              .3175
FE203
                    -6/
CLAD
      300
            . 0381
                    -6/
FUEL
      300
            . 2794
                   -6/
BLAYKT/
              5.0E-5/
٥
   1
      NOTAT
0
      7256
               4.4742-2/
   1
0
      CL52
               1.257E-2/
   1
               5.479E-3/
0
   1
      FI58
```



Figure 5: Approximate multiregion slab cell for ZPR-6 Assembly 7. The dashed lines represent reflection planes. Region 1 is the SS304 end slab (0.2223 cm), 2 is the U₃O₈ slab (0.6350 cm), 3 is the sodium (1.270 cm), 4 is the Fe₂O₃ slab (0.3175), 5 is the cladding (9.0381 cm), and 6 is the fuel (0.2794 cm).

0	1	XII 55	1.01 E-3/	
0	2	U235	3.36 E- 5/	
0	2	U238	1.57 25-2/	
0	2	016	4.7248-2/	
0	3	NA23	2.2358-2/	
0	4	016	4.724E-2/	
0	4	FES6	3.215 E-2/	
0	4	NOFAT	6.3 E- 5/	
0	5	FE56	5.706 E -2/	
0	5	C R 52	1.6396-2/	
0	5	NI58	8.118E-3/	
0	5	NH 55	1.41 E-4/	
0	6	PU239	9.8506E-3/	,
0	6	PU240	1.308E-3/	
0	6	PU241	1.473 E-4/	
0	6	U235	6.178-5/	
0	6	U238	2.7584E-2/	,
0	6	NOFAT	2.498E-3/	
1	6	PU239	8.8672E-4	CC/
1	6	PU240	1.1 944E-4	CC/
1	6	PU241	1.33E-5	CC/
:	6	U235	7.1 4E-6	CC/
1	2	U235	5.468-6	CC/
1	6	U238	2.520 3E-3	CC/
1	2	U238	3.2601E-3	CC/
1	8	NONAT	2. 3575-4	CC/
1	3	NA23	9.290E-3	CC/

1	4	016	8.947E-3	CC/
1	2	016	5.0338-3	CC/
1	5	FE56	1.26E-3	C C/
1	4	FE56	5.929E-3	cc/
1	1	FE56	5.7/8E-3	CC/
1	5	CR52	4.96E-4	CC/
1	1	CR52	2.213 E-3	CC/
1	5	NI58	2.27E-4	cc/
1	1	WI58	1.013E-3	cc/
1	5	NN55	3.9 2-5	cc/
1	1	NN 55	1.73E-4	cc/
2	7	U235	8.56E-5/	
2	7	U238	3.96179E-2	2/
2	7	NONAT	3.8E-6/	
2	7	016	2.4E-5/	
2	7	FE56	4.637E-3/	
2	7	CR52	1.295E-3/	
2	7	N I58	5.635E-4/	
2	7	NN 55	9.98E-5/	
CH	I			
ST	OP			

This run is similar to problem 7 except that additional regions are defined for the infinite slab model of the core cell. These regions have IMIX=0 so that their cross-sections are not written onto the output file. However, the macroscopic crosssection for each region is calculated and used during the self-shielding iteration. To compute the Dancoff correction, the code sums up the optical path to the left from one region containing a particular material to the next region containing the same material, and then it does the same thing to the right. In either case, if the sum gets to the cell edge before the "black" region is found, the optical path is doubled to account for the reflective boundary condition. If the left optical path is zero, it is set equal to the right path; the complementary procedure is used if the right optical path vanishes. Therefore, both symmetric and asymmetric two-region equivalent cells are formed.

It is clear from the way the multiregion cell was constructed that the homogenized core cross sections cannot be computed by homogenizing regions 1 through 5 directly. In addition, the material loadings varied from cell to cell in the actual assembly; and other structure, instrumentation, and control regions also contributed to the net densities for the assembly. TRANSX handles this problem by defining the core mix using the constituent cross sections from regions 1-5 but using the homogenized densities from the homogeneous model. When a particular isotope occurs in several different regions, the homogenized density is apportioned between them (e.g., 81.7% of the 304 stainless steel is in region 1, and 18.3% is in region 5 with different background cross sections). The CC command is used to request constituent cross sections because these densities are cell-averaged densities.

The model used in this example accounts for the heterogeneous self-shielding effect, but not for the advantage and disadvantage effects. They can be obtained by using a flux solution by regions to do a flux and volume weighting of each cross section in the assembly. This can be done using two TRANSX runs and an S_N code as shown below. These examples also illustrate how to use thermal upscatter cross sections.

F. Self-Shielded Thermal Reactor Problems

Thermal neutrons are usually important when light materials (for example, water or graphite) are present. Such problems are characterized by the presence of upscatter. In addition, thermal assemblies are often strongly heterogeneous because of high absorption cross sections. Thermal upscatter cross sections are available on some of the MATXS librariec (see Section V), and the heterogeneity is usually treated by fine-group structures and/or heterogeneous self-shielding methods.

Sample Problem 9. Homogenized Thermal Critical Assembly

```
TEST 9 -- ORBL-1 BENCHMARK
0 1 0 1 1 1 0 3 0
                         0
69 2 93 20
                          1 0
             40
                 1
                    1
                       7
GLOP
GLOP
     300. 1.
                 34.595/
               1
         .066228 120/
     E1
  1
          .033736 FLEE/
     016
1
  1
     N14 1.869E-4 FREE/
1
     U234 5.28E-7 PREE/
1
  1
     U235 4.8066E-5 FREE/
1
  1
     U236 1.382-7 FREE/
1
  1
  1 U238 2.807E-6 FREE/
1
CHI
STOP
```

This benchmark is modeled as an homogeneous sphere of uranyl nitrate in water without a container.¹⁷ It is important to use a MATXS library like one of the 69-group libraries that has both upscatter cross sections and self-shielding factors. Note that 20 upscatter groups are allowed, for a total table length of 4+69+20 =93 groups. Using 40 thermal groups gives a thermal cutoff of 2.60 eV; that is, there is no upscatter to groups above this energy. The self-shielding treatment accounts for the mixture effects and the possibility of escape from the sphere (the mean Table 3: Moderator materials on the ENDF/B-VI thermal data files. The names in parentheses give the binding state for the principal scatterer. For example, C in graphite, or H in water. The MATXS names are based on this binding condition. Elastic scattering can be either coherent (coh) or incoherent (iel). In either case, the MATXS name is constructed by appending a dollar sign (\$). The last column gives the appropriate treatment to use for the secondary atom, if any (see text).

Material	MATXS	Elastic	Secondary
Name	Name	Ireatment	Ireatment
Be	BE	coh	
BeO	BEO	coh	none
C(graphite)	GRAPH	coh	
C(polyethylene)	POLY	iel	free C
C ₆ H ₆	C6H6		none
$D(D_2O)$	D2O		free O
H(H ₂ O)	H2O		free O
Zr(ZrH _n)	ZRH	icl	H(ZrH _n)
H(ZrH _n)	ZRH	iel	$Zr(ZrH_n)$

chord length is 34.595 cm). Free-gas thermal scattering is used for all materials except hydrogen, which is bound in water.

Most ENDF/B-VI thermal evaluations give the scattering from an atom as bound in a molecule or crystal lattice, and the effects of the different atoms in a molecule must be combined in TRANSX to obtain the net thermal scattering cross sections. As an example, water in the sample problem above is a combination of H bound in H₂O and free-gas oxygen. However, some ENDF/B evaluations give the scattering for the entire molecule in the tabulation for the principal scatterer. In these cases, the thermal cross sections are renormalized to be used with the principal scatter. The TRANSX user must he careful not to include free-gas scattering for the secondary atoms. Table 3 lists the thermal evaluations available in ENDF/B-VI and indicates which treatment of secondary atoms was used. This table also indicates which evaluations require both inelastic and elastic parts. The elastic names (whether coherent or incoherent) are obtained by appending a dollar sign (\$) to the names given.

As examples, the TRANSX material specifications for graphite, beryllium oxide, methane, and ZrH might take the forms

1 1 CUAT 1. GRAPH GRAPH\$/ 2 2 BE9 1. BE0 BE0\$/

2	2	016	1.	TOR	S/
3	3	H1	6.	COH	/
3	3	CHAT	6.	NONE	S/
4	4	E1	1.	ZRH	ZRE\$/
4	4	ZR	1.	ZRE	ZK H\$ /

For self-shielding problems like this one, it is necessary to define a mixture in order to calculate the σ_0 values. However, the user often needs the "constituent" selfshielded microscopic cross sections for such applications as concentration searches or depletion calculations. These constituents can be requested by defining additional output mixes and by using mix specifications containing the key words "CC" (for cell constituent) or "RC" (for region consituent). Sample problem 9 would become

69 2 93 20 40 8 1 :4 1 0 GLOP 81 114 U234 016 U235 U236 U238 U239 2.807E-6 FREE/ 1 1 2 1 81 1. CC/ 016 1. CC/ 3 1 U238 1. CC/ 8 1 STOP

The entry "CC" tells the code to use the microscopic cross section that was added into the mix by the normal mix specification for this material and region. (The "CC" and "RC" commands are equivalent for a homogeneous case like this one.) The transport code now has the choice of using the macroscopic cross section or remixing using the micros. If the densities in a concentration search were to vary enough to change the self-shielding, it might be necessary to repeat the TRANSX calculation for the new mix.

Sample Problem 10. TRX-1 by Regions

TEST 10 -- TRI-1 REGIONS 0 1 2 1 0 3 0 0 20 40 4 3 69 96 3 3 6 2 FUEL CLAD NOD 300 .75892 3 1.15 1.15/ FUEL.

```
CLAD 300 .280851 -3/

MOD 300 1.78489 -3/

1 1 U235 6.253E-4 FREE/

1 1 U238 4.7205E-2 FREE/

2 2 AL27 5.1728E-2 FREE/

3 3 H1 6.676E-2 H20/

3 3 016 3.338E-2 FREE/

CHI F238 F235 C238

2 NFTOT 1 U238

3 NFTOT 1 U238

3 NFTOT 1 U238

5 TOP
```

The TRX-1 thermal benchmark²³ is a hexagonal lattice of aluminium clad pins of enriched uranium in water. The heterogeneity option IHET=3 is used to compute the Dancoff corrections for an infinite hexagonal lattice. A 69-group library is used because it contains thermal scattering matrices. Note that free gas scattering is called up for most of the materials, but the cross sections for hydrogen in water are used for H1. Using 40 thermal groups gives a cutoff at 2.60 eV. Upscatter is limited to 20 groups; any larger energy gains would be truncated so as to preserve the scattering cross-section. The table length is 20+69+3+4, so all down=catter groups will be kept.

The output P_1 cross sections are written in group ordering on a file in GOXS format, which can be read directly by the ONEDANT S_N code if its name is changed to MACRXS. Another copy of this file must also be made with the name SNXEDT for use by the ONEDANT edit module. The ONEDANT input for this case follows:

```
1
TKL-1 INFINITE LATTICE CELL - 69 GROUPS - ENDF/8-VI
DINENS= 2 69 8 3 3 3 4 40 T
INESH= 0. .4915 .5042 .5753 .9482 XINTS= 17 3 3 17
ZOWES= 1 0 2 3 T
LIB= MACRIS T
ASSIGN= MATLS T
SOLIN= 1 1 0 0 3 SOLOUT= 1 0 0 0 0 0 0 1 NORM= 1. T
RZMFLX= 1 T
```

Note that no mixing is done in ONEDANT because the MACRXS file already contains the appropriate macroscopic cross sections for the two regions. The two key statements are LIB=MACRXS and ASSIGN=MATLS. When the flux calculation is complete, ONEDANT writes a flux file that can be used as input to a second TRANSX run. The RZMPLX file is a nonstandard CCCC file for regular zone moments flux (see Section IV). Sample Problem 11. TRX-1 Infinite Lattice Cell

......

```
TEST 11 -- TRI-1 INF. LATTICE CELL
0 5 0 1 2 1 0 3 69
                         3
2 2 10 40 1 1 3 5 4 3
CELL
          .75892 3 1.15 1.15/
FUEL
     300
         . 28085
                -3/
    300
CLAD
MOD 300 1.78489 -3/
1 1 U235 6.253E-4 FREE/
     U238 4.7205E-2 FREE/
  1
Ń
     AL27 5.1728E-2 FREE/
  2
1
     H1 6.676E-2 H20/
  3
1
    016 3.338E-2 FREE/
  3
1
        F235 C238
CHI F238
2 IFTOT
         1 U238
  JFTOT
        1 U235
3
  IG
     1 U238
45 24/
STOP .
```

The entire self-shielding calculation is repeated, but this time everything is homogenized into IMIX=1 using the given volumes and the computed region flux moments from the RZMFLX file. Note that the results are collapsed into two groups. The resulting breakpoint will be at 0.625 eV so that the response edits can be used to calculate the epithermal-to-thermal ratios usually quoted for thermal critical assemblies. These two-group cross sections could also be used to calculate the net leakage from the system using a two-dimensional RZ transport code, thereby determining whether the basic cross-section set gives a multiplication close to critical.

IV. INTERFACE FILES

TRANSX uses interface files for cross-section input, flux input, and crosssection output. Some of the output files are simple card-image files, but the interface files²⁵ follow the pattern of the CCCC interface files.

A. The MATXS Generalized Cross-Section File

The standard CCCC files, ISOTXS, BRKOXS, GRUPXS, DLAYXS, and ISOGXS, have a number of shortcomings that have made them difficult to use outside of the fast-reactor community where they were developed. ISOTXS is limited to a particular set of reactions that is inadequate at the higher energy found in fusion problems; it does not allow for such important supplementary information as heating (KERMA) and damage energy cross sections, and it does not allow for the partial cross sections needed for sensitivity analysis. BRKOXS doesn't provide for self-shielded group-to-group scattering cross sections. ISOGXS has the same problems as ISOTXS. There is no format for delayed photons or decay heat. These files use different formats to represent data that are basically similar, so the overhead of coding required to use them is high.

The MATXS format was designed to generalize and simplify the existing files. The first design principle was that all information would be identified using lists of Hollerith names. As an example, if the list of reactions included in the file contains entries such as NF, NG, and N2N, it is trivial to add additional reactions such as KERMA or DPA. The second design principle was that the file would be designed to hold sets of vectors and rectangular matrices and that the same format would be used regardless of the contents of the vectors and matrices. As a consequence, once a code is able to handle $n \leftarrow n$ matrices, it can also handle $\gamma \leftarrow \gamma$ data; once a code can handle $\gamma \leftarrow n$, it can also handle $n \leftarrow \gamma, \beta \leftarrow n$, or even $p \leftarrow d$. It was originally decided to divide the file into "data types" identified by its input and output particles. As an example, $\gamma \leftarrow n$ is a data type characterized by input particle equals neutron and output particle equals photon. The matrices contain cross sections for producing photons in photon group n_g due to reactions of neutrons in group n_n . The vectors, if any, would contain photon production cross sections versus neutron group. Then, each data type was divided into materials and submaterials. A material might be a particular isotope or mixture (e.g., U238 or SS304). Each submaterial could then represent a particular temperature and background cross section in a basic library, or perhaps a particular region for a reactor. Starting with NJOY 91.0, this organization was changed to have the material as the outermost loop. Each material was then divided into submaterials, where each submaterial could belong

Name	Definition
N	Neutron
G	Photon (gamma or x-ray)
Ρ	Proton
D	Deuteron
Т	Triton
Н	³ lle
A	Alpha (⁴ He)

Table 4: Particle types defined for current MATXS libraries.

to a given data type and have a given temperature and background cross section. This organization makes it much easier to add, delete, or extract materials from MATXS files. The MATXS library format that resulted from following these general principles is described in Appendix B. The standard CCCC form is used.

As in all CCCC-type files, a MATXS library starts with a file identification record that gives the name of the file, a user identification string, and a version number. This is followed by the file control record; here NPART refers to the list of particles found in the file (e.g., N, G, B); each particle has its own distinctive group structure. NTYPE is the count of data type names, and NHOLL is the count of words in a general description of the contents of the file, processing methods, accuracy criteria, and so on. NMAT is the number of materials given in the library. MAXW is the maximum size of a record to be found in the file. It helps in establishing storage requirements. Finally, LENGTH gives the total number of records in the library.

The file data record gives the particle and data type names (see Tables 4 and 5), the names of the materials, and the number of groups for each particle. It also characterizes each data type by identifying its input and output particles. Finally, it gives the number of submaterials for each material and the LOCM vector. The LOCM vector makes it easy to skip directly to a desired material without reading all the intervening data.

Each material starts with a material control record, which gives the name of the material and information on the submaterials. Once again, it is possible to jump directly to a desired submaterial by using the LOCS vector.

The first record for each submaterial is the vector control record. The vector data are organized by reactions with Hollerith names (HVPS). To keep the

Name	Definition	
NSCAT	Neutron scatting	
GSCAT	Photon scattering	
NG	Photon production	
NTHERM	Thermal neutron scattering	
PSCAT	Proton scattering	
PN	Proton-to-neutron matrix	
NP	Neutron-to-proton matrix	
•••	clc.	

Table 5: Data types defined for current MATXS libraries.

maximum record size reasonable, the vectors can be given in several blocks; each vector is taken to be nonzero only between the given limits (NFG, NLG). Thus, excess zeroes are omitted for threshold reactions or for reactions with no high-energy groups such as thermal scattering. Any reaction that can be named can be stored in MATXS.

The convention used for naming most reactions is clear from the example in Appendix H (*c.g.*, NG, NA, or N2N). The symbols NWTO, NWT1, NTOTO, and NTOT1 refer to the flux- and current-weighted components of the library weight function and the neutron total cross section. Fission may be split into the partial reactions NP, NNF, and N2NF; in any case, NFTOT is the total fission reaction. Discrete-inelastic scattering in ENDF is denoted by the level plus 50; thus, MT51 is (n, n'_1) . On the MATXS library, this is given as NO1. If the level decays by particle emission, the particle ide tifiers are tacked on after the number (*e.g.*, N62DAA). Emission of ³He is denoted by H.

The neutron files also contain some special quantities. HEAT and DAME are the local heat production and damage energy production cross sections. NUBAR, XI, GAMMA are the continuous slowing-down parameters. INVEL is the group-averaged reciprocal velocity. Finally, NUD and CHID are for delayed fission as described in Sec. 11-B.

In the thermal data type, reaction names identify the binding condition; FRZE for free gas, H2O for light water, D2O for beavy water, POLY for polyethylene, BENZ for benzene, GRAPH for graphite, ZRHYD for zirconium, and BEO for beryllium oxide. The elastic part is distinguished from the inelastic part by a \$ sign on the end of the name. In the photon interaction data type, the reactions are GWTO, GTOTO, GHEAT, and GDAME analogous to the neutron names, GCOH for coherent scattering, GINCH for incoherent scattering, GPAIR for pair production, and GABS for photoelectric absorption.

After the vector blocks comes the matrix control record. The representation used for cross-section data is similar to ISOTXS in that it packs the data as bands of source groups that scatter into each given sink (final) group.

The matrix names used in the current libraries are the same as for the corresponding vectors. All the partial scattering matrices are kept separate for maximum flexibility. If groups of these partial matrices were to be combined by an auxiliary program, new names would have to be defined such as TOTAL or INELAS.

Summary descriptions of several existing MATXS libraries will be found in Section V. These libraries were derived from ENDF/B·V and ENDF/B·V1; they use neutron group structures with 30 to 187 groups and photon structures with 12 or 24 groups.

B. RTFLUX, RZFLUX, RMFLUX, and RZMFLX Files

Four different flux input files are allowed. RTFLUX is a CCCC standard interface file²⁵ for regular total flux, which is normally used to give the \mathbb{P}_2 flux by fine mesh interval. TRANSX uses it as if it contained region or coarse mesh fluxes; if only one region is defined, RTFLUX is a handy way to pass central flux into TRANSX from a CCCC standard code. RZFLUX is a similar CCCC file for gegular gone averaged flux.

What TRANSX really needs are the P_0 , P_1 ,... moments of the flux given by coarse mesh regions. However, as all S_N codes have $\phi_0, \phi_1, ...$ available from the calculation of the source term, the moments can be made available for output in a moments flux file. The ONEDANT and TWODANT codes now have the capability to write the RMFLUX (regular moment flux) file by fine mesh or RZMFLX (regular gone averaged moment flux) by coarse mesh. The format of these non-standard CCCC files is the same and is given in Appendix E.

It should be noted that ordering of the flux moments becomes more complex for cylindrical or two-angle plane geometry when the associated Legendre functions appear in the formulas. A good discussion of the ordering used in the Los Alamos transport codes appears in the ONETRAN report.²⁶ TRANSX should be limited to P_1 sets for these cases.

C. CARD 6E12 Format

This is a simple material-ordered card format that can be read by almost all transport and diffusion codes. The γ put file starts with a block for the first Legendre order of the first material, then gives the second order, third order, and so on. The file then continues with the first order of the second material, elc., until all the data blocks have been written.

Each block starts with a simple header card that gives the material name, Legendre order, table length, and number of groups. The data are then given using a 6E12.5 format starting with position 1 of final group 1, position 2, position 3, ..., up to position NTABL of the first group. The data then continue with all the positions for the second group, *etc.*, until the block is complete. Each block contains NTABLxNGRGUP words, six per line, with a sequence number on each line. An example of this format is given with the Sample Problem 1 output in Appendix A.

D. CLAW Format

This is also a material-ordered format very similar to ICJT=1. The NED+3 edit positions are written out in a separate response edit block with its own header card containing the name of the material and the word "EDITS." All the groups for position 1 are written first in GE12.5 format. A reaction identifier and sequence number are added to each card in columns 73-80. The same process is followed for the rest of the edits. The first NED positions are then removed for each group of the scattering blocks, and the blocks are written out using 6E12.5 cards. Each card has its Legendre order and a sequence number in co¹umns 73-80.

The edit tables are very easy to access with a text editor in this form, and the positions reserved for edits in the high-order tables have been removed for economy in transport code storage.

E. FIDO Output

The transport tables for IOUT=1 and 2 are usually more than half zeroes. This waste of words on the input file can be reduced by using FIDO format in which a group of N zeroes is replaced by one entry that says "repeat zero N times". TRANSX uses fixed-field FIDO, which divides a 12-column field into a 3-column part and a 9-column field without a decimal or an E (e.g., 1234567.6 is 1.234567). A negative number or a number with a two-digit exponent field ends up with less precision. To obtain the repeat option, "nnR" is put into the 3-column field where nn is the number of times that the number in the field following R is to be repeated.

A table is terminated by an entry with T in column 3 of the 12-column field and every other column is blank.

The materials and scattering blocks are written in the same order as for IOUT=1, and a header card is provided for each block. The tables are much more compact but harder to read. Some FIDO processors are much more elaborate than this one, but TRANSX uses only the R and T options.

F. ANISN Output

There has been a long tradition in which ORNL codes wrote the source term of Eq. (1) as shown, but Los Alamos codes kept a factor of $(2\ell+1)$ explicit in the sum. The only difference between the FIDO and ANISN options is that ANISN tables include the $(2\ell+1)$ appropriate for ORNL codes and FIDO tables leave it out, as required by the Los Alamos codes. Modern codes like ONEDANT often have options to read either type of table. An easy way to distinguish between them is to look at the scattering matrix for a high-energy group with almost deltafunction forward scattering. Los Alamos tables will have $P_0 \ge P_1 \ge P_2 \ge P_3$ The higher orders will tend to grow for ORNL tables.

G. ISOTXS Output

As pointed out above, the ISOTXS format has some shortcomings as a general library format; however, it is well suited as an output format from TRANSX because many codes can use it as input. For the convenience of the TRANSX user, the ISOTXS format specification²⁰ is reproduced in Appendix F.

When selecting ISOTXS output, the edit names can be STRPL, SNGAN, SFIS, CHISO, SNALF, SNP, SN2N, SND, or SNT. The first and second must be present (TRD and NG, respectively), and SFIS must be present (if not zero) to allow $\bar{\nu}$ to be calculated from $\nu\sigma_J$. The others are optional. As an example,

```
$ IOUT=8, NED=9, NEDS=9
...
CEISO STRPL SNGAN SFIS SN2N
SNALF SNP SND SNT/
1 CHI/
2 TRD/
3 NG/
4 NPTOT/
5 N2N/
6 NA/
7 NP/
```

The ordering of the names is not important. Note that the edit specifications for SND could load HEAT or DAME into the corresponding position rather than the (n, d) cross section. This kind of trick must be used with care because some S_N codes may calculate their particle-balance-absorption edit from SNGAM +SFIS+SNALF+SNF+SND+SNT-SN2N. A separate vector chi is computed for each isotope with fission, but no set chi vector or matrix is provided.

The output coding produces a total scattering matrix only. It uses the option in which a different "block" is used for each Legendre order. Subblocking is not used, so the scattering blocks each consist of a single record containing all source and sink groups for that order. No attempt is made to take full advantage of JBAND and IJJ to remore zeroes, but the matrices are triangularized and truncated using MAXUP and MAXDN for some savings in output file size.

The parameters AMASS through KBR in the isotope control record are set to zero.

H. GOXS Output

As pointed out in Section 11, group ordering is the natural organization of the cross sections in an S_N code. Using a group-ordered library file saves the time required to flip the data from material to group ordering, and it allows large group structures to be used.

The GOXS (pronounced go-ex-ess) file is described in Appendix G. It was designed as a CCCC-compatible binary file to provide an efficient internal cross-section representation for any S_N code. It is compatible with the MACRXS file used in ONEDANT and TWODANT.

Like MATXS, the GOXS format uses Hollerith names for materials and vector cross sections. Thus, the number and types of response-function edits that can be included are essentially unlimited. Also, the availability of names for materials and reactions enables the code designer to produce a very readable printed output.

As noted previously, most of the vector cross sections are not needed during the flux calculation but only at the end of the job when responses are computed. If the transport code being used has such a post-edit capability, the TRANSX user can run a separate edit job as in Sample Problem 4, and the GOXS output coding will prepare a short file with no scattering tables. This file is then compatible with the SNXEDT file used by the EDIT module of ONEDANT and TWODANT. Scattering matrices on GOXS are packed using "bands" as in ISOTXS to minimize the number of zeroes on the file. It is also advantageous to minimize the number of zeroes in the memory region where cross sections are stored. The same JBAND and IJJ arrays can be used to pack the scattering data into memory, if desired. Thus, GOXS is just an image of the internal data representation of the transport code giving maxiumum efficiency in data transmission. ONEDANT and TWODANT use this feature of GOXS.

No card-oriented group-ordered format is provided because (1) no de facto standard such as FIDO exists, and (2) it makes more sense to convert the existing S_N codes to read GOXS.

V. SOME MATXS LIBRARIES

MATXS libraries are produced from evaluated nuclear data in the ENDF/B format using the NJOY nuclear data processing system.^{12, 13, 14, 15} As of versiou 2.0, the preferred libraries for TRANSX are based on ENDF/B-VI. However, several older libraries based on ENDF/B-V have been converted to the current version of the MATXS format also.

A. MATXS10

This is a library with 30 neutron groups and 12 photon groups (30 x 12) that includes 117 materials from ENDF/B-VI. Photon production and photon interaction are included, but self-shielding isn't. The Legendre order for the scattering matrices is P₄. The materials are given at a temperature of 300 K. A complete list of materials and reactions included can be obtained from an INDEX of the library (a partial listing of this index will be found in Appendix H).

MATXS10 was prepared with a fusion + fission + 1/E + thermal Maxwellian weight function. It is useful for many high-energy calculations, including criticals such as GODIVA. It works surprisingly well for many shielding problems for which resonance self-shielding is not too important. The library is compact, and TRANSX runs usually require only a few seconds on a fast computer.

B. MATXS5

This library is similar to MATXS10, except that it was generated from ENDF/B-V. It includes neutron data for 100 materials (including 13 Dosimetry isotopes), photon production for 65 materials, and photon interaction cross sections for 87 elements. Several special Los Alamos evaluations and a lumped fission product from the 'Nestinghouse Hanford Engineering Development Laboratory (HEDL) are also included. See the INDEX for more details.

C. MATXS11

This is an 80 x 24 library from ENDF/B-VI intended for fast-reactor analysis. However, the group structure and weight function used also make MATXS11 a reasonable choice for fusion studies and shielding calculations.

This library features extensive self-shielded cross sections for the temperatures 300, 400, 600, 800, 1200, 1600, 2000, 3000, and 4000 K. Background cross sections vary from infinity down to a lower limit that depends on the normal applications

of the materials. The Legendre order is P_5 for both neutrons and photons. See the INDEX for a complete list.

The 24-group photon structure is much better than the 12-group structure for low photon energies and for energies around the annihilation peak at 0.5 MeV. It also does a better job of describing the high-energy "shoulder" of capture gammas in some systems. This can be important for heating calculations.

MATXS is more expensive to use than the 30×12 libraries, but it is capable of giving good results for a wider range of problems.

D. MATXSe

This 80 \times 24 library is similar to MATXS11, except that it was prepared from ENDF/B-V. There are 148 materials, including 72 with photon production.

E. MATXS12

This is a 69 x 24 library from ENDF/B-VI for light-water-reactor problems. It includes self-shielded cross sections for the important actinides. Thermal scattering data are given for all the materials, with bound scattering for the important moderators. The group structure has 42 thermal groups extending up to 4 00 eV.

F. MATXS7

This is also a 69 x 24 library, but it is based on ENDF/B-V. There are 118 materials, including the most important fission products. Photon production is included for 42 materials.

G. MATXS13

The MATXS13 library has 187 neutron groups and 24 photon groups for a limited number of materials useful for shielding problems. It includes neutron data, photon data, thermal data, and self-shielding data in one unified package.

The photon data are the same as in MATXS11, but the increased neutron accuracy leads to better calculations of capture, hence better photon spectra.

The thermai structure includes 55 groups below 4.45 eV, thereby providing better resolution around the $1 \cdot eV^{240}Pu$ resonance and more accuracy at energies below 0.0253 eV. The fine epithermal group structure tracks the slowing down of neutrons better than the coarse groups of MATXS12. As more materials are

added to this library, it will become an accurate general purpose tool for thermal problems.

Finally, for high-energy problems, the thermal groups can be collapsed out. The result will still have a finer group structure than MATXS10 or MATXS11, and it can be used to attack difficult problems that do not seem to converge with the simpler libraries.

H. MATXS8

This library is similar to MATXS13, except that it was processed from ENDF/B-V. It is an 80-group library for fast-reactor and fusion-reactor problems.

VI. PROGRAMMING DETAILS

A. Version Control

TRANSX is maintained using the UPD code,²⁷ which is similar in concept to the UPDATE code used on Cray and CDC systems. It keeps a history of changes to a computer code that is very reful as part of a Quality Assurance (QA) program. However, UPD is much simpler than UPDATE, thereby facilitating transport from system to system. This transportability is very important, because it ensures that one set of "updates" will produce the same code for compilation on all the target systems. With other systems used in the past, modifications sometimes behaved differently on different systems.

UPD reads two files. The file named SRC contains the base version of the source code for the program. This file can be broken up into separately reamed sections with "*DECK name" cards. However, TRANSX uses only the single deck named "TRANSX." For simplicity, there is no counterpart to the binary "program library" used in the UPDATE system. The second file is called UPW, and it contains the commands to modify the base source file into a compiler input file for the current version.

The following example illustrates the features of the UPN file needed for maintaining the TRANSX code:

```
+CPL
•LST
+IDENT UP1
*/ 6 JAN 92 -- INCREASE SIZE OF MAIN CONTAINER ARRAY
•D TRANSI.182
     DINZUSION A(600000)
+D TRAISI.222
     NANAX=5.00000
•IDERT VERS
VERSION UPDATE FOR TRANSX
+D TRANSI.12
     VER5. 2.1 (6 JAN 92)
С
+D TRANSI.216
     DATA VERS/' VERS. 2.1 (6 JAN 92)'/
IDENT UNICOS
•/ _____
•/ MACHINE-DEPENDENT CHANGES TO TRANSI
*/ FOR CRAY-IMP MACHINES USING UNICOS AT LOS ALANOS
•/ ELAPSED TIME
O TRAISI. 3800
     CALL SECOND(TIME)
+/ DATE
+D TRANSX.3816
```

CALL DATE(HDATE) •/ WALL CLOCK TIME •D TRANSX.3826 CALL CLOCK(HTIME)

The first line of this example gives the command CPL, which causes a compiler input file named CPL to be written containing the updated program. The command LST causes a listing file named LST to be written. These listing files are a little easier to read than normal CPL files.

The first set of changes begins with the card •IDENT UP1. The modified lines on the CPL or LST files will contain labels like UP1.2, UP1.3, etc. This particular "ident" changes two lines in order to increase the size of the main container array. The symbol "•/" introduces a comment line. It is very important for QA purposes to include comments in each ident that describe the purpose of the change. It is also very important that each change be dated as shown.

The next ident updates the version label on the code listing and the version string that is written to every TRANSX listing file. The numbering convention used for TRANSX is suggested by this example. Including an ident called UP1 increments the TRANSX version to 2.1; UP2 would give 2.2; and so on. Note that the date from the last ident is transferred to ident VERS. This scheme will be used for all "official" changes to TRANSX. Users are free to make private changes using their own special ident names.

The third ident implements some machine-dependent changes. These changes will be discussed below.

The general format for the delete command is

D liae1, line2

"", line2" is omitted, line2 is taken to be the same as line1. The actual values for the line references can refer to lines in the base code (e.g., TRANSX.234) or to changed lines (e.g., UP1.2). In general, any line label that appears on the current CPL or LST file can be used. The delete command causes the range of lines named to be removed, and any text following the command is inserted at that point in the file. The insert command has the form

•I line1

Any text following the command is inserted after the named line. There is also a "before" command:

•B line1

that causes the following lines to be inserted before the named line. Other features of the UPD code and more details on its use will be found in the UPD report.

B. Code Conversion

TRANSX is designed to be converted easily between long-word machines (CDC, CRAY) and short-word machines (IBM, VAX, Sun). All Hollerith variables begin with the letter "II" and can be defined to be double-precision with a statement like IMPLICIT REAL+8 (H). Expressions for the lengths of arrays and pointers for the starting locations of arrays all use the parameter MULT, which is set equal to 1 if the machine word can contait, 8 or more characters and 2 if not (an IBM word contains 4 characters). The input routine FREE contains a machine-dependent variable MACHWD for the number of characters per machine word (MACHWD=10 'or CDC, 8 for CRAY, and 4 for IBM, VAX, or Sun).

Most of the changes required to go back and forth between short-word and longword machines are carried out automatically using the conditional capabilities of the UPD code. As an example,

```
•IF SW
INPLICIT REAL+8 (H)
•ELSE
INPLICIT INTEGER (H)
•ENDIF
```

The user only has to be sure to include the line "•SET SW" in the UPN input file for the UPD run (put it before the first "ident").

After these changes have been made, there may still be some incompatible system calls connected with input/output, CP time. date, clock time, and fatal error handling. Fatal errors use subroutine ERROR, which can often be adapted to provide trace back information or to leave a "drop file" for later analysis by an on-line debugging code. The subroutine TIMER is used to obtain elapsed CP time in seconds. DATER is used to obtain a date string (which can be formatted in different ways as iong as it contains only 8 characters; examples are "mm/dd/yy," "dd/mm/yy," or "ddmmmyy"). WCLOCK is used to obtain the wall clock time; that is, the time of day for the TRANSX run. UPD idents are provided for several common machines and operating systems. These decks can also be used as starting points in constructing the updates required for other systems.

The problems of input/output have been alleviated somewhat by FORTRAN. 77, but some installers may still find incompatibilities. TRANSX uses the CCCC standard²⁵ subroutines SEEK, REED, and RITE. Local optimized versions may be available at installations that use existing reactor-physics codes. RITE is used to write a binary record and is very simple. REED is used to read binary records, but the reads may be nonconsecutive. At Los Alamos, nonconsecutive reads are supported using forward and backward record skipping. On some systems (especially VAX), it may be more efficient to rewind and skip forward rather than to use the backspace command. The SEEK routine is used to connect external files by name to logical unit numbers. The FORTRAN-77 version used in the code should work for most systems.

The Los Alamos CTSS version of TRANSX uses the program card to assign the input, output, and tty units, and the FILEREP call to enable reassignment of units or the execute line (e.g., transx input=test1,output=out1). Some systems may require explicit OPEN statements for these units.

C. Storage Allocation

TRANSX uses variable dimensioning and dynamic storage allocation for efficient use of available memory. If insufficient memory is available, paging and multiple passes through the library file are used. This allows very large transport sets to be produced, if necessary, at the expense of running time.

For material ordering, the code determines how many materials and Legendre tables can fit into the available storage. It then makes one pass through the library file, produces the cross sections for the first set of materials and orders, and writes them out on the output file. It then rewinds the data file and repeats the entire process for the second set of materials and orders. This process continues until all the desired materials have been produced. The minimum set is one Legendre order and one material; if this does not fit, group ordering can be used.

For group ordering, the code determines how many groups (NGMAX) can fit in memory at one time. It then makes one pass through the data file for each set of NGMAX groups, writing out the group cross sections after each pass.

This paging strategy allows TRANSX to be run on small machines if necessary, but if a large memory is available, the code will run much more efficiently. Storage is assigned to a single container array A with NAMAX words (NAMAX is 300000 in the Los Alamos CRAY version). The allocation can be changed easily by changing the two statements DIMENSION A(300000) and NAMAX=300000.

A similar scheme is used during the calculation of self-shielding σ_0 values. Cross section data are read in for groups JGLO to JGNI, the σ_0 values are computed and

printed, and the code repeats the process for the next group range, JGLO to JGHI.

Part of the NAMAX words of storage is reserved for reading the records of the MATXS cross section file by the statement INMAX=6000). This should be sufficient for MATXS files prepared with the default value of MANW (5000).

D. Description of TRANSX

TRANSX begins by initializing MULT, the units for system input and output (5 and 6), the zeratch unit used for self-shielding data (20), the storage sizes (see Section V-C), and the parameter N12, which determines the maximum line length for the printed output file.

1. User Input After printing \triangle heading, TRANSX begins to read the user's input. Most values are stored in variables with the name given in the input instructions (e.g., IPRINT, NTABL) or in the container array A using a pointer named by prefixing L to the name from the input instructions [e.g., IMIX values are stored at $\triangle(LINIX+1)$, $\triangle(LINIX+2)$, etc.]. The exception is the heterogeneity data INET, CHORD, etc., which are packed in the container array as they are read; the array with pointer LIMET locates the parameters for each region in the big array.

Several additional arrays are defined using the input data. LNCHX points to an array used to check whether any materials requested in the mix specifications were not found on the library (sometimes this means that the name was misspelled). LGTAB points to an array containing the coarse-group index for each fine group that is constructed from the slightly different representation used for input. INIXR is the "real mix" index; it runs over the NNIXR mix specifications that are not flagged as constituent cross sections with the labels "CC" or "RC". A related array is ICONS, which points to the "real" mix command for each constituent mix command. The number and names of the unique materials found in the mix specifications are also determined (see NUNIQ and the array HUNIQ starting at LHUNIQ+1). The array JNIX contains pointers to the unique material for each mix specification (or zero if the material is a constituent material), and the array KNIX contains a counter that distinguishes between the subsequent occurrences of a unique material in the mix specifications. Finally, MAXU is the largest number of times any unique material occurs. It is used later for dimensionally the FGAM and FFIS arrays.

If a spatial collapse is indicated by the input M_{i} , commands, the flag ISPC is set. In addition, TRANSX comparises the cell volume from the region volumes RVOL given by the use: the prove of the program in the cell

61

The IRF array is used later to flag when the weighting flux for each particular particle has been read in.

At this point, memory is allocated for the flux arrays. The Legendre components of the fine flux are stored as if they were in an array FFLUX(fine group, region, order). For no collapse, this array starts at LFLUX+1. If collapse is requested, a coarse-flux array of the form CFLUX (coarse group, mix, order) starts at LFLUX+1, and the fine-flux array starts at the location

LFLUX+HGROUP+HMIX+(HLMAX+1).

The use of NLNAX+1 orders provides for the flux moments needed for inflow transport corrections.

If an input flux file has been defined, it is read following the comment "READ FLUX FROM INTERFACE FILE." If there are fewer than NREG regions on the input file, the last region is used to fill out the table. If there are not NL+1 orders on the file, the last order is used to fill out the table.

If self-shielding has been requested, the appropriate background cross sections and self-shielding factors are computed by calling subroutine SIGZER. This routine uses temporary storage defined with respect to the pointer LC. It writes its results on the scratch tape MSS=MSCR.

Next, storage is allocated for the scratch arrays. LCNORM points to the total fission neutron production used to normalize chi; that is, the denominator of Eq. (19). LPNU points to the array of prompt fission $\dot{\nu}$ values. LDNU and LDCHI point to the delayed fission data. LSFIS is a pointer for the microscopic infinitely dilute fission cross section of the current material. LSGAM is used for the microscopic infinitely dilute capture cross section. LFFIS and LFGAM are used to store the self-shielded cross sections for fission and capture. LABSC is a pointer to the array used to accumulate the particle-balance absorption by the direct method. And finally, LTERPT locates the array of interpolation factors used for temperature interpolation.

Next, the storage above LC is allocated for the cross-section table using different strategies for material and group ordering, as described in the preceding section. The maximum numbers of materials, orders, and groups that can fit into memory at one time are NHMAX, NLMAX, and NGMAX. The parameters IMLO, INHI, ILLO, ILHI, JGLO, and JGHI are used throughout the code to decide which materials, orders, and groups get loaded into the accumulating cross-section array. 2. Main Processing Loop The code is now ready to begin a pass through the MATXS library accumulating the appropriate cross sections into the transport table. This is done by calling subroatine RMATXS. When the subroutine returns control to the TRANSX level, the accumulated cross sections for the current range of materials or groups are ready to be printed out or written to the desired output file.

All the material-ordered output options are given first inside a loop over output mix and order (IMLO to IMHI. ILLO to ILHI). The transport table in memory is stored starting at LC as if it were an array C(position, order, material, group). The "order" and "material" indices are often collapsed into the single variable JM, which takes on NJM=NLMAX=NMMAX values. The logic required to extract data from the table is clear from the coding for printed output. The CARD, CLAW. FIDO, and ANISN output options use simple subroutines to generate the appropriate output. The design of the WISOT routine is described in Section IV-G. Note that TRANSX reacts to the specific edit names defined for ISOTXS (e.g., SNGAM, SFIS, SW2M, etc.) only. The pointers LLSFIS, LLSW2M, etc., tell where the corresponding cross sections are located in TRANSX memory. The scattering blocks are written using the same fixed size for all materials; no attempt is made to squeeze out additional zeroes. On the printed listing, the numbers are arranged as N12 columns. A programmer can easily change N12 to make the output look good on either a terminal or a line printer.

The group-ordered output options are produced next inside a loop over the range of groups in memory (JGLO to JGHI). Only printed and GOXS outputs are provided for group ordering, but other output options could be added at this point. The tables in memory have the form C(position, order, material, group) starting at LC. The logic needed to write the tables is fairly clear from the coding for printed output. The GOXS output routine WGOXS writes out all the response edits specified in the TRANSX input, but they are reordered so that the first four vectors are always σ_{tot} , σ_e , $\bar{\nu}\sigma_f$, and χ . The transport cross section for diffusion TRD is written next if available. The scattering matrix is packed to minimize the number of zeroes written to the output file.

After completing the output for this range of materials, orders, and groups, TRANSX loops back to process the next set until all the requested tables have been produced. Final messages are printed (including the list of materials not found from LMCHK), files are closed, and the job terminates. 3. Reading the MATXS file When the RMATIS routine is entered, the accumulation arrays are zeroed, and the material loop is begun. The code checks whether each material on the MATXS library is needed; if not, it skips it entirely. The LOCM array and the random access capabilities of REED make this fairly efficient.

The MATXS material control record is read in to memory at pointer L2, and a new pointer L3 is defined. Note that the material control data stays in memory while this material is being processed. A loop over all the submaterials for this material is executed; that is, a loop over data type, temperature, and background σ_0 . The first submaterial in each data type is called the "base material." By scanning up from each base material, the list of **STERP** temperatures **TTENP** for this data type is obtained. On subsequent steps of the loop, the current type, temperature, and background cross section are extracted (ITYPE, TMAT, SMAT). All background cross sections except infinity (1.E10) are skipped in **RMATXS** because the self-shielding calculation has already been performed in SIGZER.

With the data type identified, the values that depend on data type are assigned. These values include the identity of the input particle and the number of input groups (JINP, NING), the identity of the the output particle and the number of output groups (JOUTP, NOUTG), and the offsets for shifting the data block in a coupled set (JC1, JC2).

With the temperature identified, the appropriate interpolation weight for this submaterial is computed. This version of TRANSX limits itself to linear interpolation. Because each material may occur in several different mixtures or regions, the interpolation weights are stored in an array TERPT versus mix specification index INX. If this particular submaterial does not contribute to any mix or region, the parameter ITERPT has the value zero, and the entire submaterial can be skipped. Otherwise, the submaterial is processed by starting the code segment with the comment "PROCESS THIS SUBMATERIAL."

The vector reactions are processed first. The MATXS vector control record is read into memory at pointer L3, and a new pointer L4 is defined for the blocks of vector cross-section data. The g heral procedure is to loop over the complete set of reactions needed for the transport tables and edits. For each group, the selected cross sections are multiplied by TERPT and/or density and added into the appropriate storage location. If necessary, self-shielded cross sections from the MSS scratch file are substituted for the cross sections from the MATXS vector block. The additional factor RVF is used to correct constituent cross sections for the definition of the density used ca input; they can be either cell densities or region densities. See the block of coding after "GET GROUP CROSS SECTION." The model weight functions in the cross-section block are treated specially. The data for each particle are read into the WIN array using the offset JC1, and the WIREAD flag is set.

If an elastic scattering correction has been requested (INITF<0), the expansion coefficients representing the smoothed fine-group flux are computed (X1, X2, X3) and used to compute the value and slope of the the smoothed flux at the bottom of each group. The coefficients used later for the removal corrections are stored in the array ESCC. If self-shielded cross sections were prepared in SIGZER, an appropriate self-shielded model flux has already been prepared. If there is no flux input file, the FFLUX and CFLUX values are computed using the model weight functions.

Note that only the base submaterial is processed for self-shielded reactions in the non-thermal range because the self-shielded cross section already includes the effects of temperature. All submaterials for these reactions are processed in the thermal range. All submaterials for non-shielded materials are also processed. This allows the temperature effects to be incorporated by using the TERPT factors.

The block of coding that loads the σ_{e} , σ_{1ot} , and $\sigma_{g \leftarrow g}$ positions is especially complex because of the transport correction options defined by Eqs. (7) through (13). The P₀ total is put into the σ_{e} position, and the correction defined by Eq. (21) is subtracted later while the scattering matrices are being processed. The contributions of the P₀ and P₁ total crows sections to the total and in-group positions are made here also, and the matrix contributions needed for the transport correction, if any, are added later.

The block of coding after the comment "FISSION CROSS SECTION" only comes into play for data that have not been self-shielded (normally the thermal data type). The variable FFIS thus includes the effect of temperature on $\bar{\nu}\sigma_f$.

The transport cross section for diffusion (TRD) is calculated next. The P_1 total is added in here (or P_0 if P_1 is not available), and the total P_1 scattering cross section is computed and subtracted later during matrix processing.

The thermal correction to the total cross section was described in Section II-J. Note that both the absorption and total positions are involved; if the reaction is elastic, the cross section is subtracted, and if the reaction is the coherent or incoherent one specified in the mix specifications, the cross section is added.

The final block of coding in the material loop adds in any contributions to the edit positions using the multiply-and-accumulate algorithm, which allows any linear combination to be calculated. After the material loop is complete, the scratch cross sections are saved. These are quantities characteristic of this particular submaterial that will be needed later in the vector loop for processing the scattering matrices, or for calculating steadystate fission. SFIS and SGAM contain the infinitely dilute fission and capture cross sections used in computing $\dot{\nu}$ and the shielding factors for photon production. The scratch data DNU and DCHI are the delayed neutron yield and spectrum, respectively. Finally, GANNA is γ_g used in the elastic scattering correction defined by Eq. (38).

The reaction loop is continued until all the required vector reactions for this submaterial have been processed. Then the loop over the matrix reactions is started. Each matrix block (reaction) can be divided up into a number of subblocks. A subblock may be part of the main matrix, in which case it contains data for a range of final groups JL to JU that contains fewer than MAXW words. Or it may be a constant spectrum or production cross-section subblock. The coding after the comment "SCATTERING SUB BLOCKS" handles this loop over subblocks.

If the current submaterial is needed for one of the accumulating mixes, the code sets up a loop over final energy group (sink group), initial energy group (source group), and Legendre order. The group-to-group cross sections XSNOW for each order may contain a matrix part and/or a part computed from the constant spectrum and production vectors. If the reaction is elastic scattering, it may be replaced by a self-shielded cross section. Next, the thermal correction is made, if desired (the elastic scattering at low energies is replaced by the appropriate incoherent and coherent thermal scattering). The resulting group-to-group cross section is multiplied by the appropriate factor from TERPT to affect temperature interpolation, and, optionally, an elastic removal correction factor is applied.

Following the comment "GET TRANSFER CROSS SECTION," this final inicroscopic cross section XSNOW is converted to a macroscopic cross section (note the correction for cell or region densities implicit in the factor RVF) and multiplied by the flux ratios needed for collapse and homogenization. In addition, the neutronphoton matrices for capture and fission are self-shielded using the data in the FGAM and FFIS arrays.

Separate blocks of coding are provided to accumulate the transport matrix with truncation of the table, if necessary, to compute a photon source in the edit positions (GAN option), to finish the calculation of the particle-balance absorption σ_a , and to complete the transport cross section for diffusion. In the block labeled "FISSION VECTORS," the fission matrix is multiplied by the Nux and summed over source group: to get the prompt part of the numerator in Eq. (19). The same term is added to the array in CNORM to be used to normalize χ later.

The last step in the reaction loop adds the prompt component of fission into the microscopic $\bar{\nu}\sigma_f$ in PNU. When the reaction loop finishes, the data in PNU are into mptly converted to $\bar{\nu}$ by dividing by σ_f , which was saved at LSFIS during the $f = \frac{1}{2} \log \sigma_f$ in the vector cross sections.

After the end of the submaterial loop, all of the effects of self-shielding and the approxime have been included in the cross sections. The contributions from the approximation to the prompt or steady-state values of $D\sigma_j$ are then added into the transport table. Any delayed contributions to CNORM, the denominator of the expression for χ , are also added in here.

Unis process is continued until all materials have been processed. The fission γ is it in normalized using the data in CNORM. In addition, the particle-balance absorytion in the transport table is compared with the directly-computed value in ABSC. If the value in ABSC is small with respect to the transport cross section but still will have a large effect on σ_{\bullet} , it is used to replace the previous value of σ_{\bullet} . The tables are now complete for the ranges of mixes, orders, and groups currently allowed in memory. RMATXS returns to TRANSX, so they can be printed and/or written onto the requested output interface file.

4. Self-Shielded Cross Sections The effects of self-shielding are computed in subroutine SIGZER, and the resulting self-shi Jed cross sections are written out onto the scratch file NSS. In order to allow for possible storage limitations, the calculation is done by "pages" of MAIGRP groups (see JGLO and JGHI). The input MATXS file is read once for each page of groups.

The file data is already in place at pointer L1, and each pass starts by looping over all the "real" materials needed from the MATXS library. The procedure is similar to the one used in subroutine AMATXS, except that only the total, elastic, capture, fission, and elastic downscatter cross sections are extracted. They are all stored in the array SIGS(ISIGZ,IX,IMX,IG), where the indices refer to the background σ_0 , the reaction type, the mix specification, and the group index, respectively. Each cross section is interpolated to the temperature appropriate to IMX before being stored in SIGS. The code allows for up to 20 elastic scattering elements. For a P₄ set, this corresponds to the in-group element and 4 downscatter terms.

68

Once all the cross sections for this range of groups have been read in, the σ_0 values can be computed by iteration. The code begins a loop over all the "real" materials in the mix specifications. The current material IMX lies in region IRG, which determines the heterogeneity parameters to be used. The σ_0 values for each group is computed in two parts. First, the inixture part is computed using Eq. (25), where the sum runs over all the other materials in IRG. The code then begins to calculate the escape part of σ_0 .

For IHET=1, the simple result of Eq. (26) is used with an effective mean chord specified by the user.

For lattices of cylinders (IHET=2 through 4), a macroscopic cross section homogenized over the two regions outside the fuel is required. Because the fuel is in the first region, the locp sums the macroscopic cross section times volume for the next two regions, obtaining the desired average in SHOD. The rest of the calculation of the escape cross section is straightforward.

For the slab options (IHET=5 through 8), the sweep through the cell starts from IRG and moves to the right, summing the product of the macroscopic total cross section for each region times the thickness of the region. The sweep continues until a region containing the current material or the edge of the cell is reached. if the edge of the cell is reached, the sweep either jumps to the other end of the cell and continues to the right (periodic cells) or reverses direction and sweeps back to the left (reflective cells). When a region containing the current materials is reached, the code repeats the entire process sweeping to the left. The final results are XXR and XXL, which can be used for λ_R and λ_L in Eqs. (33) or (34). Note that regions at periodic boundaries have their full thickness, but only half the thickness is given for regions lying on reflection planes. The calculation of the escape cross section can now be completed using Eq. (35).

Finally, the escape part is added into σ_0 , the new c_0 is checked against the previous value, and the loop over materials and groups is continued. If convergence to within EPSZ has been achieved for all groups and materials, ISOK will be one, and the loops will terminate.

The final steps after convergence are to print out the table of computed σ_0 values by material and group for this range of groups, to write out the results to the scratch file, and to continue the loop over pages of groups.

E. Error Messages

The error messages written by TRANSX are listed below along with an analysis of the problem and recommended corrections.

FULL TABLE REQUIRED FOR COUPLED SETS

It doesn't make sense to truncate a coupled set by giving a table length less than 3+NED+NGROUP because this would remove parts of the n-gamina block. Increase the table length requested.

REGION INPUT ERROR

The region temperature has an unreasonable value. Check your input. This kind of thing can result from a missing "/" on a previous card.

MIX INPUT ERROR

Either INIX or IREG has an unreasonable value. Check your input. Once again, there may be a missing terminator on a previous card.

NO CONSTITUENT CROSS SECTION MATCH

The code is unable to find a constituent cross section with the given material name in the given region. Check your input.

INCONSISTENT COLLAPSE

The number of fine groups found by adding up the NGROUP values given in the collapse specification (input card 10) must be equal to the number of fine groups entered in ICOLL on input card 2.

INCONSISTENT FLUE FILE

This message usually means that the group structure on the flux fi.e does not agree with the structure implied by the TRANSX input file. It may also mean that this is a three-dimensional flux file.

INCONSISTENT GROUP STRUCTURE

The number of groups found on the input library for the various particles are not consistent with the input quantities NGROUP and NFINE. Sometimes this means that the wrong MATXS library was mounted. Otherwise, check the values of the parameters, especially for complex coupled sets.

STORAGE EXCEEDED FOR LSS

There is not enough storage available for the self-shielding calculation. Either reduce the number of mix specifications and/or the number of groups, or increase the size of the main container array. The instructions on how to do this are given at the beginning of this section.

INSUFFICIENT STORAGE FOR MATERIAL ORDERED FORM

This message means that not even one Legendre order of one material can fit in the available storage. Either change to group ordering (which is more appropriate for very large group structures), or increase the size of the main container array. See above.
INSUFFICIENT STORAGE FOR GROUP ORDERED FORM

This message is unlikely considering the small amount of memory required to process a single group. It means that too much storage was consumed in flux arrays and scratch arrays. The only recourse is to increase the size of the main container array. See above.

ILLEGAL FORM

The IFORM variable can only take on the values 1 and 2. Any other value is illegal.

TEMPERATURE OR SIGNA ZERO MISMATCHED

This message comes from SIGZER. It results from trying to match the T and σ_0 values for one submaterial with the list of possible T and σ_0 values for all the submaterials in the material. It should not occur.

SEEK--BAD OPTION hname nop

This version of SEEK does not handle "nop" values of 4 and 5, nop values outside the normal range 0-7, or the "change" option. This error should not occur for TRANSX.

SEEK -- FILE DOES NOT EXIST hname

For the "read seek" option, the file "hname" must already exist in your local file space.

SEEK--HO UNITS AVAILABLE hname

Seek assigns unit numbers between 10 and 19. This error means that more than 10 units are open. It should not occur for TRANSX.

• Ϊ. :

72

VII. Using BBC

BBC is a utility code for maintaining MATXS libraries. It can be used to convert them back and forth between binary and BCD (i.e., formatted) modes, to selectively list the contents of a library, or to construct an index to the materials and reactions on a library. It can also be used to mod fy a MATXS library in several ways. It can select a subset of the materials from a library in order to make a new smaller library (smaller libraries run faster with TRANSX). The select option could also be used to list the cross sections for a particular material, if desired. BBC can insert new materials into a library or delete old materials. The select and insert capabilities can be used together to combine the information from two MATXS libraries. Finally, BBC can edit the MATXS version number, user identification, and file identification information.

A. BBX Input Instructions

The input instructions follow. They were copied from the comment cards near the start of the BBC source file.

T	IPIT	
•		
CARD	1	
	I I NPF	INPUT FILE TYPE
		1 BIBARY (BARE = MATES)
		-1 CODED (BARE = TELT)
	10077	OUTPUT FILE TYPE
		o iche
		1 BIEARY (HANE = MATES OR MODOUT)
		-1 CODED (NAME = TEXT OR NODOUT)
	INCOF	NODIFY FLAG
		O NO NODIFICATIONS (DEFAULT)
		1 NODIFICATIONS GIVEN
	LIST	LIST OPTION
		0 NOT LISTED
		1 LISTED (DEFAULT)
	INDI	INDEXING OPTION
		9 NOT INDEIRD
		1 INDEXED (DEFAULT)
CARD	2 (LIST	.NE.O CHLY)
	IRD	RECORD TYPE PRIST FLAGS.
		1 FILE IDEBTIFICATION
		2 FILE CONTROL
		3 SET BOLLERITH IDENTIFICATION
		4 FILE DATA
		6 GROUP STRUCTURE
		6 NATERIAL CONTROL
		7 VECTOR CONTROL

8 VECTOR BLOCK 9 SCATTERING MATRIX CONTROL 10 SCATTERING SUB-BLOCK ONE VALUE FOR EACH RECORD TYPE O NOT PRINTED, 1 PRINTED. CARD 3 (INODF. NE.O ONLY) HOPER MODIFY OPTION 5 SELECT NATERIALS FROM NATIS INPUT FILE I INSERT NATERIALS FROM MODISP FILE ENAT1 FIRST NATERIAL IN RANGE LAST MATERIAL (DEFAULT=HMAT1) EHAT2 REPEAT CARD 3 UNTIL & BLANK LINE IS BRAD CARD 4 (INODF. NE.O ONLY) NEW FILE VERSION NUMBER TEVVER BUSER NEW USER ID STRING (UP TO 16 CHARS) CARD 5 (INODF.NE.O ONLY) LINE OF NEW HOLLERITE ID DATA (72 CHAR MAX) HLINK REPEAT LINE 5 FOR RACE LINE OF THE NEW BOLLERITE DATA BLOCK. TERMINATE WITH AN EMPTY LINE. IF NO LINES ARE READ, THE ORIGINAL BLOCK IS UNCHANGED.

Because binary libraries cannot normally be moved between different computer systems, the MATXS file specification includes formatted version of each different record type (See Appendix B. Therefore, BBC includes the capability to convert files back and forth between BCD and binary modes. The simplest input to go from binary to BCD would be

1 -1 0 0 0

and to go from BCD to binary,

-1 1 0 0 0

The binary unit (input or output) is always called "MATXS", and the BCD unit (input or output) is always called "TEXT." It is often useful to make a partial listing and an index at the same time as a binary to BCD conversion. This can be done using

1 -1 0 1 1 1 1 1 1 1 1/

The partial listing on the file named OUTPUT will contain only the first five records of the MATXS library, namely, file identification through group structures. The index listing will appear on the file named INDEX. The cost of running TRANSX can be reduced substantially by preparing a condensed version of a MATXS library that contrins only the materials of interest. For example, a special library to run the TRX benchmark calculations could be prepared using the following input:

```
1 1 1 0 0

S B1/

S 016/

S AL27

S U235/

S U238/

/

112 *T2LANL BBC*/

*SHORTENED VZRSION OF NATIS12*/

*FOR TRI BENCHNARK CALCULATIONS*/

/
```

The original library should be MATXS12 with its name switched to MATXS. The version number is changed to 112, so it would be convenient to change the name of the MODOUT file to MATXS112.

When it becomes necessary to update an isotope on an existing MATXS library or to insert a new isotope into the library, it is inconvenient and expensive to repeat the entire process of preparing the library using the MATXSR module of NJOY. An alternative approach is to prepare a small MATXS library containing only the new material, and then to add it to the library. Assuming that the MATXS10 library hzs been copied to the file MATXS, and that a new version of the ¹¹B data is on unit MODINP, you run BBC using the following input:

```
1 I O O 1
S W1 B10
I B11/
S CWAT ES253 /
10 •T2LAWL BBC•/
```

Note that ¹¹B is replaced and that the original version number and Hollerith descriptive text are used for the new library. The output file MODOUT should then be switched to MATXS10 and used to replace the previous version.

B. BBC Error Messages

The error messages than can result when running BBC are listed below along with recommended actions.

INPUT MUST BE BINARY FOR MODIFICATIONS

Because modifications can be specified in any order. It is problem problem in nary files for both MATXS and MODINP. The randour give contraining the of REED can then be used to jump to the desired materials. Existing for D files can be converted to binary using scharate BBC runs before the modification run is made.

TOO NANY MATERIALS, CHANGE NUCS

Several arrays that depend on the number of mathematicity dependence by NUCS. If more materials are needed, this pumber can be accounted by NUCS=200 in the parameter statement at the start of BBE'.

TOO MANY VECTORS, CHANGE NVEC

Several arrays that depend on the number of reaction and the significant of the second statement of the second statement of the statement of t

SZEK--BAD OPTION hname nop

This version of SEEK does not handle "nope" values of 4 and 5, nop values outside the normal range 0-7, or the "change" contains. This error should not occur for BBC.

SEEK--FILE DOES NOT EXIST hname

For the "read seek" option, the file "hname" must already exist in your local file space.

SEEK -- NO UNITS AVAILABLE haame

Seek assigns unit numbers between 10 and 19. Thus error means that more than 10 units are open. It should not occur: for HEC.

VIII. ACKNOWLEDGMENTS

The author would like to thank the TRANSX users who may helped to find and fix problems in the code over the years since its introduction. More excently Denise George has made significant contributions to TEANSX and did much of the rewrite of BBC.

IX. REFERENCES

- 1. K. D. Lathrop, "DTF IV, A FORTRAN-IV Program for Solving the Multigroup Transport Equation with Anisotropic Scattering," Los Alamos Scientific Laboratory report LA-3373 (November 1965).
- 2. W. W. Engle, Jr., "A User's Manual for ANISN: A One-Dimensional Discrete Ordinates Transport Code with Anisotropic Scattering," Oak Ridge Gaseous Diffusion Plant Computing Technology Center report K-1693 (1967).
- 3. R. E. Alcouffe, "Diffusion Acceleration Methods for the Diamond Difference Discrete Ordinates Equations," Nucl. Sci. Eng. <u>64</u>, 344 (1977).
- 5. R. E. Alcouffe, *L. W. Brinkle, Jr.*, D. R. Marr and R. D. O'Dell, "User's Guide for TWODANT: A Code *Lackage* for <u>Two</u>-Dimensional <u>D</u>iffusion-<u>Accelerated, Noutral-Particle Transport," Los Alamos National Laboratory</u> report LA-10049-M, Revision 1 (1984).
- 6. T. J. Seed, "TRIDENT-CTR User's Manual," Los Alamos Scientific Laboratory report LA-7835-M (May 1979).
- J. W. Davidson, B. A. Clark, D. A. Marr, T. J. Seed, J. Stepnek, C. E. Higgs and D. M. Davierwalla, "TRISM: A Two-Dimensional Finite-Element Discrete Ordinates Transport Code," Los Alamos National Laboratory draft report (1986).
- 8. K. L. Derstine, "DIF3D: A Code to Solve One-, Two-, and Three-Dimensional Finite-Difference Diffusion Theory Problems," Argonne National Laboratory report ANL-82-64 (1984).
- J. F. Briesmeister, Ed., "MCNP-A General Monte Carlo Code for Neutron and Photon Transport, Version 3A," Los Alamos National Laboratory report LA-7396-M, Rev. 2 (1986).
- R. E. MacFarlane, "TRANSX-CTR: A Code for Interfacing MATXS Cross-Section Libraries to Nuclear Transport Codes for Fusion Systems Analysis," Los Alamos National Laboratory report I.A-9863-MS (1984).
- These equations are derived in many standard references on reactor physics. This discussion follows G. I. Bell, G. E. Hansen and H. A. Sandmeier, "Multitable Treatments of Anisotropic Scattering in S_N Multigroup Transport Calculations," Nucl. Sci. E: p. 28, 376 (1967).
- R. E. MacFarlane and R. M. Boicourt, "NJOY: A Neutron and Photon Processing System," Trans. Am. Nucl. Soc. 22, 720 (1975).
- R. E. MacFarlane, R. J. Barrett, D. W. Muir, and R. M. Boicourt, "'IJOY: A Comprehensive ENDF/B Processing System," in A Review of Multigroup Nuclear Cross-Section Processing, Proceedings of a Seminar-Workshop, Oak Ridge, Tennessee, March 14-16, 1978, Oak Ridge National Laboratory report ORNL/RSIC-41 (October 1978).

- 14. R. E. MacFarlane, D. W. Muir, and R. M. Boicourt, "I NJOY Nuclear Data Processing System, Volume 1: User's Manual," Los Jamos National Laboratory report LA-9303-M, Vol. 1 (ENDF-324), (May 1982). R. F. Mac-Farlane, D. W. Muir, and K. M. Boicourt, "The NJOY Nuclear Data Processing System, Volume II: The NJOY, RECONR, BROADR, HEATR, and THERMR Modules," Los Alamos National Laboratory report LA-9303-M, Vol. 11 (ENDF-324) (May 1982). R. E. MacFarlane and D. W. Muir, "The NJOY Nuclear Data Processing System, Volume III: The GROUPR, GAM-INR, and MODER Modules," Los Alamos National Laboratory report LA-9303-M. Vol. III (ENDF-324) (October 1987). D. W. Muir and R. E. Mac-Farlane, "The NJOY Nuclear Data Processing System, Volume IV: The FR-RORR and COVR Modules," Los Alamos National Laboratory report LA-9303-M, Vol. IV (ENDF-324) (December 1985).
- R. E. MacFarlane, "Introducing NJOY 89," in Proceedings of the Seminar on NJOY and THEMIS," 20-21 June 1989, OECD/NEA Data Bank, Saclay, France (1989). See also Nuclear Theory and Application Progress Report, January 1, 1989-April 1, 1990, pp. 21-38, Los Alamos National Laboratory report LA-11972-PR (December 1990).
- 16. I. I. Bondarenko, Ed., Group Constants for Nuclear Reactor Calculations. (Consultants Bureau, New York, 1964).
- R. E. MacFarlane, R. B. Kidman and R. J. LaBauve, "The Background Cross Section Method as A General Tool for Reactor Analysis," in CONF-78040, Proc. ANS Topical Meeting, Gallinberg, Tennessee, April 1978. E. G. Silver, Ed. (1978).
- 18. R. B. Kidman, "Cross Section Structure Factor Interpolation Schemes," Hanford Engineering Development Laboratory report IEDL-TME-71 (1) (1971).
- 19. R. W. Hardie and W. W. Little, Jr., "IDX, A One-Dimensional Diffusion Code for Generating Effective Nuclear Cross Sections," Battelle Northwest Laboratory report BNWL-954 (1969).
- R. Protsik, E. Kujawski, and C. L. Cowan, "TDOWN-IV: A Computer Code to Generate Composition and Spatially Dependent Neutron Cross Sections for Multigroup Neutronics Analysis," General Electric Company report GEFR-00485 (1979).
- 21. R. B. Kidman, R. E. MacFarlane, and M. Becker, "Flux Interpolation for Elastic Downscatter," Trans. Am. Nucl. Soc. 26, 580 (1977).
- 22. J. U. Koppel and D. H. Houston, "Reference Manual for ENDF Thermal Neutron Scattering Data," General Atomic report GA-8774 revised and reissued as ENDF-269 by the National Nuclear Data Center at Brookbaven National Laboratory (1978).
- 23. "ENDF-202, Cross Section Evaluation Working Group Benchmark Specifications," Brookhaven National Laboratory report BNL 19302 (ENDF-202) with various revisions (1974).
- 24. L. R. Greenwood and R. K. Smither, "Displacement Damage Calculations with ENDF/B-V," Proceedings of the Advisory Group Meeting on Nuclear Data for Radiation Damage Assessment and Reactor Safety Aspects, 12-16 October 1981, IAEA, Vienna, Austria (1981).

- 25. R. D. O'Dell, "Standard Interface Files and Procedures for Reactor Physics Codes, Version IV," Los Alamos Scientific Laboratory report LA-6941-MS (1977).
- T. R. Hill, "ONETRAN: A Discrete Ordinates Finite Element Code for the Solution of the One-Dimensional Multigroup Transport Equation," Los Alamos Scientific Laboratory report LA-5990-MS (1975).
- 27. R. E. MacFarlane and D. C. George, "UPD: A Portable Version-Control Program," Los Alamos National Laboratory report LA-12057-MS (1991).

· •

·

·

×0

Appendix A. SAMPLE OUTPUT LISTINGS

This appendix shows portions of the output listing from Sample Problem 1, the "CARDS" output from Sample Problem 1, the output listing from Sample Problem 2, and the "FIDO" output from Sample Problem 3. The parts that were removed are indicated with ellipses.

Sample Problem 1

TRANSI TRANSPOJ PROM NAT	NT CRO	VERS. 2.0 (2 JAN 62) RS SECTIOES RAN ON 01/22/92 BRARIES AT 14:52:05
TITLE		
TEST 1	NA:	FREIAL ORDERFD TABLES
OPTIONS		
IPRINT	0	(O=LONG/1=SNORT)
IOUT	1	(0=NONE/1=CARD/2=CLAW/3=FIDO/4=AHISH/5=GOIS/6=ISOTIS)
IPROS	0	(O=DIRECT/1=ADJOINT)
ISET	1	(1=NN/2=GG/3=COUPLED)
IFORM	1	(1=NATVISE/2=GROUPVISE)
ITINE	1	(1=STEADY-STATE/2=PROMPT)
IDECAY	0	(RESERVED)
ITRC	3	(U=NO TRANSPORT CORR/1=CONS.P/2=DIAG/3=B-N-S/4=INPLOW)
ICOLL	0	(0=¥0 CUL.1P3Z/W)INE=CULLAPSE)
INITP	0	(0=LIBRARY TLUX/1=PO CARDS/2=RTFLUI/3=NZNFLX/NEG FOR ESC)
PARANETE	RS	
IGBOUP	30	GROUPS
NL.	3	TABLES
BTABL	34	POSITIONS IN TABLE
TUP	0	UP SCATTER GROUPS
STEG	0	TUNBER OF THERMAL GROUPS
ENIX	3	NIXES OR NATERIALS
INEG	1	regions
BAIRS	6	NIXTURE 1 VECTFICATIONS
TED	1	LITRA EDIT POSITIONS
NEDS	1	EDIT SPECIFICATIONS
NIX NAME	5	
1 CAR		

2 TUNGST

2 U-236

REGI	.		TEMP(K)	51 28	NETEROGENIETY	
1		:	3.000E+02	1.000 E+00	0	
SPZC	NIX	REG	NUCLIDE	DE85179	THERMAI.	MATRIX NOD
1	1	1	CHAT	1.000 E+00		
2	2	1	W182	2.630E-01		
3	2	1	¥183	1.430E-01		
4	2	1	¥184	3.070 E-0 1		
5	2	1	¥186	2.8602-01		
6	3	1	U235	1.0002+60		

EDIT NAMES

1 REAT

6 II EAD 8

EDIT SPECIFICATIONS

1 BEAT 1.602E-19

INPUT MATIS FILE

FILE ID NATIS T2LANLJOY VERS 10 30112 LIBRARY FROM ENDF/B-VI, JAN 1992

NATERIAL-ORDERED CROSS SECTION TABLES

WORDS AVAILABLE FOR TABLE = 6120 HUNBER OF OUTPUT NIITURES PER PASS = 3 HUNBER OF LEGENDRE ORDERS PER PASS = 2

0.15

FINE FLUX FOR REGION 1 SUN 3.4360E+07 3.723E+03 7.392E+04 1.905E+04 1.410E+04 2.312E+C4 4.396E+04 ... 5.408E+05 1.176E+05 1.031E+06 7.341E+05 5.809E+05 1.156E+06 ... 1.155E+06 1.154E+06 1.155E+06 1.155E+06 1.155E+06 ...

++ CARBON PO ++

POSITION	GROUP 1	group 2	GROUP 3	GROUP 4	
		<u></u>			• • •
1 REAT	4.840E-13	4.141E-13	3.7208-17	2. 648E-13	• • •
2 ABS	6.273E-02	6.843E-02	8.8932-02	9.7 83E-0 2	
4 TOTAL	9.122E-01	8.808E-01	9.027E-01	8.613E-01	
6 INGRP	5.052 E-02	5. 846E-0 2	1.131 E-0 1	1,2538-01	• • •
6	0.000 E+00	2.610E-01	1.895E-G1	2.3378-01	
7	0.000E+00	0.000 E+00	8.684E-02	1.395E-01	
8	0.000E+00	0.000E+00	0.000 E+ 00	1.201E-01	
9	0.000E+00	0.000E+00	0.000E+00	0.000E+00	• • •
10	0.000E+00	0.000 E+ 00	0.0002+00	0.000 E+0 0	• • •
11	0.000E+00	0.000 E+00	0.000E+00	0.0002+00	

0.0002+00 12 13 0.0002+00 0.0002+00 0.0002+00 0.0002+00 ++ CARBON P1 ++ GROUP 1 GROUP 2 GROUP 3 GROUP 4 ... POSITION ------e--------------... -----1.5092-02 5.0078-02 4.900E-02 ... 5 INGRP -3.695X-03 1.1222-01 3.5928-02 ... 6 0.0002+00 1.6852-01 0.000E+00 0.000E+00 -1.072E-02 -5.070E-02 ... 7 8 0.000E+00 0.000E+00 0.0002+00 3.419E-02 . . . 0.0002+00 0.0002+00 0.0002+00 0.0002+00 ... 9 0.000E+60 0.000E+00 0.000E+00 0.000E+00 ... 10 0.0002+00 ... 0.000E+00 0.000E+00 0.000E+00 11 0.000E+00 0.000E+00 0.000E+00 0.000E+00 ... 12 0.000E+00 0.000E+00 0.000E+00 0.000E+00 ... 13 . . . ** TUNGST PO ** POSITION GROUP 1 GROUP 2 GROUP 3 GROUP 4 ... _____ ----------------_____ . . . 1 BEAT 6.395E-14 5.750E-14 5.2858-14 7.722E-14 . . . -2.338E+00 -2.016E+00 -1.9!8E+00 -1.660E+00 ... 2 ABS 4 TCTAL 2.778E+00 2.885E+00 2.9822+00 3.139E+00 ... 7.029E-01 5 INGRP 4.1627-01 5.8768-01 2.8252-01 0.0005+00 9.8125-02 5.6325-02 6.5235-02 ... 0.000E+00 0.000E+00 2.114E-02 3.397E-02 ... 7 0.0002+00 0.0002+00 0.0002+00 6.0002-02 ... 8 0.0002+00 0.0002+00 0.000E+00 0.000E+00 ... 9 0.0005+00 0.0005+00 0.0005+00 0.0005+00 ... 10 0.0002+00 0.0002+00 0.0002+00 0.0002+00 ... 11 . . . 2.35 TRANSI COMPLETE Sample CARD Output CARBON PO 34X 30 TABLE

0.0002+00

0.000E: 10

0.0002+03 ...

0 4.84046E-13 6.27289E-02 0.00000E+00 9.12203E-01 5.05172E-02 0.00000E+00 1 0.000C0E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 2 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 3 0.00000E+00 0.000002+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 4 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 5 0.00000E+00 0.00000E+0C 0.00000E+00 0.60000E+00 4.14145E-13 6.84320E-02 6 0.00006E+00 8.80773E-01 5.84583E-02 2.61000E-01 0.00000E+00 0.00000E+00 7 0.000002+00 0.000002+00 0.000002+00 0.000002+00 0.000002+00 0.000002+00 8

	0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00	9
	0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00	10
	0 000007+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00	11
	0.00000E+00.0.00000E+00.3.71959E-13.8.89298E-92.0.00000E+00.9.02727E-01	12
	1 13132E-01 1 69513E-01 8 68378E-02 0.00000E+00 0.00000E+00 0.00000E+00	13
		14
		15
		16
		17
		19
		10
		20
		20
		21
		22
	0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 2.12/41E-13 1.40423E-01	23
	0.00000E+00 9.23660E-01 9.45104E-02 2.17309E-01 1.53811E-01 1.26440E-01	24
	5.05288E-02 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00	25
	0.00003E+00 D.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+09	26
	•••	_
	CARBON P1 34X 30 TABLE	0
	0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00-3.59481E-03 0.00000E+00	1
	0.00000E+00 0.00000E+00 0.00000E+00 0.0000E+00 0.00000E+00 0.00000E+00	2
	0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00	3
	0.00000E+00 G.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00	4
	0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00	5
	0,00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00	6
	0.00000E+00 0.00000E+00 1.50855E-02 1.68473E-01 0.00000E+00 0.00000E+00	7
	0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00	8
	0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00	9
	0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00	10
	0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00	11
	0,00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.0000E+00 0.00000E+00	12
	5.00742E-02 1.12206E-01-1.07248E-02 0.00000E+00 0.00000E+00 0.00000E+00	13
	0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00	14
	0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00	15
	TUEGST PO 341 30 TABLE	C
	7.72240E-14-2.33829E+00 0.00000E+00 2.77838E+00 4.15159E-01 0.00090E+00	1
	0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00	2
	0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00020E+00	3
	0 00006E+00 0.00000Z+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00	4
		2
	0 00000F+00 0 00000F+00 0 00000F+00 0 00000E+00 6.39486E-14-2.01584E+00	6
	0 00000F+00 2 88828F+00 5 87643E-01 9 81161E-02 0 00000E+00 0 00000E+00	7
		•
		•
		14
	U. UUUUUETUU U. UUUUUETUU D. / DUEVE-19-1. VIGUGETUU U. UGUUETUU 2. VO22/ETUU 7 000000-01 5 001000-00 0 119030-00 0 000000000 0 000000000 0 00000000	44
		13
		24
	0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00 0.00000E+00	10
•	••••	

• • • •

<u>94</u>

Sample Problem 2

•• GROUP 4 ••

. . .

The following selected parts of the output for Sample Problem 2 show the main differences between group-ordered output and material-ordered output. The ellipses show where portions of the output have been removed to make it fit on the page better.

```
GROUP-ORDERED CROSS SECTION TABLES
------
WIRDS AVAILABLE FOR TABLE =293235
BUNDER OF GROUPS PER PASS = 30
                                                                 0.25
++ JROUP
           1 ...
POSITION
            CARBON PO
                        CARBON P1
                                    TUEGST PO
                                               TUNGST P1
                                                          . . .
_____
            -----
                        _____
                                    --------
                                                -----
                                                          . . .
  1 BEAT
            4.840E-13
                        0.000E+00
                                   7.7228-14
                                               0.000E+00
                                                          . . .
  2 ABS
            6.273E-02
                        0.000E+00 -2.338E+00
                                               0.000E+00
                                                          . . .
  3 HUSIGY
            0.0002+00
                        0.000E+00
                                    0.000E+00
                                               0.000E+00
                                                          . . .
  4 TOTAL
            9.1228-01
                        U.000E+00
                                    2.7782+00
                                               0.000E+00
                                                          . . .
  5 INGRP
            5.052E-02 -3.595E-03
                                    4.1528-01
                                                1.063E-01
                                                          . . .
•• GROUP
          2 **
POSITION
            CALBON PO
                        CARBON P1
                                    TUNGST PO
                                               TUNGST P1
                                                          . . .
------
            -----
                        -----
                                    ------
                                               -----
                                                          . . .
  1 BEAT
            4.1412-13
                        0.000E+00
                                    6.395E-14
                                               0.000Z+00
                                                          . . .
  2 ABS
            6.843E-02
                        0.000E+00 -2.016E+00
                                               0.0005+00
                                                          . . .
  3 JUSIGF
            0.000E+00
                        0.0G0E+00
                                    0.000E+00
                                               0.0002+00
                                                          . . .
  4 TOTAL
            8.808E-01
                        0.000E+CO
                                    2.8882+00
                                               0.000E+C0
                                                          . . .
  5 INGRP
            5.846E-02
                        1.5092-02
                                    5.8762-01
                                               1.9582-01
                                                          . . .
            2.610E-01
                        1.6852-01
                                    9.8125-02
                                               3.0832-02 ...
  6
++ GROUP
          3 **
POSITICE
            CARBON 70
                        CARBON P1
                                    TUECST PO
                                               TUBGST P1
                                                          . . .
_____
            ------
                        --------
                                    -----
                                               -----
                                                          ...
  1 NEAT
            3.720E-13
                        0.000E+00
                                    5.750E-14
                                               0.0002+00
                                                          . . .
 2 ABS
            8.893E-02
                        0.0002+00 -1.9182+00
                                               0.009E+00
                                                          . . .
  3 NUSIGF
            0.000E+00
                        0.000E+00
                                    0.0002+00
                                               0.000E+00
                                                          . . .
  4 TOTAL
            9.027E-01
                                   2.9825+00
                        0.0002+00
                                               2.000E+00
                                                          . . .
  5 INGRP
            1.1312-01
                        5.007E-02
                                   7.0291-01
                                               2.3828-01
                                                          ...
            1.8955-01
                        1.1228-01
                                    5.6328-02
                                               1.0338-02
                                                          . . .
 7
            8.684E-02 -1.072E-02 2.114E-02 3.049E-03 ...
```

85

POSITION	CARBON PO	CARBON P1	TUEGST PO	TUNGST P1	•••
		*		.	
I BEAT	2.648E-13	0.000 E+00	5.285 E-14	0.000E+00	• • •
2 ABS	9.783E-02	0.000E+00	-1.660E+00	0.000E+00	• • •
3 NUSIGF	0.000E+00	9.000 E+0 0	0.000E+00	0.000E+00	• • •
4 TOTAL	8.613E-01	0.000E+00	3.139E+00	0.000E+00	
5 INGRP	1.253E-01	4.900E-02	8.825E-01	2.986E-01	• • •
6	2.337E-01	3.592 2-0 2	6.523E-02	1. 405E-0 2	• • •
7	1. 395E-01	-6.070 E -02	3.3972-02	4.507E-03	
8	1.201 E -01	3.4192-02	5.000E-02	7.066E-03	.

• • • •

Sample FIDO Output

This is a portion of the FIDO file produced by Sample Problem 3. Note the use of the repeat option (R) to reduce the number of zeros in the file. Also note that each section of the table is terminated (T).

AL-27 PO 461	42 TABLE				0
4853214+0	· 125011-5 0+0	1370380-6	1351485-741R	0+0	1
4782239+0	-131070-5 0+0	1376950-6	1345360-7 1	579228-7	2
40R 0+0	4172989+0 -137062-5	0+0	1381834-6 1	905284-7	3
1645197-7	4424464-839R 0+0	3024617+0	-145317-5	0+0	4
1372002-6	2143598-? 1436255-7	6515701-8	4029540-838R	0+0	5
1525512+0	-136368-5 0+0	1347490-6	2446186-7 1	603739-7	6
7063104-8	5850607-8 7195758-837	7R 0+0	5759916-1 -	122417-5	7
0+0	1434578-6 3622383-7	2225031-7	8017632-8 7	699577-8	8
8992142-9	8102238-836R 0+0	4084915-1	-883606-6	0+0	9
1744504-6	8556169-7 4623703-7	1966109-7	1720993-7 1	862897-7	10
1686533-7	1343128-735R 0+0	3420562-1	-638749-6	0+0	11
2028878-6	1046921-6 3246706-7	1474077-7	9130573-8 8	814702-8	12
8281922-8	78937348 8470646-834	R 0+0	2874572-1 -	364922-6	13
0+0	2313129-6 1489384-6	4769129-7	1361422-7 7	847955-8	14
8011077-8	8681245-8 8326775-8	8076120-8	8362688-833R	0+0	16
2669560-1	-307740-6 0+0	2559229-6	1725468-6 4	832581-7	16
1 84930 0-7	8692349-8 7190167-8	7714827-6	8330074-8 7	761373-8	17
7435174-8	9456518-832R 0+0	2172306-1	-250957-6	0+0	18
2720071-6	1886117-6 5327229-7	1461140-7	2712444-8 1	036958-7	19
6761745-8	7046503-8 7275467-8	6538448-8	6289563-8 7	994375-8	20
 1 94298-1 0	773477-10 180186-09	294745-09	341299-09 6	10631-09	148
3984604-9 1					149
AL-27 P1 461	42 TABLE				0
4R 0+0	7017552- 8458 0+0	8126679-8	4661935-844R	0+0	1
9478804-8	2160393-8 7284464-943	R 0+0	8426757-8 1	164833-8	2
1679947-8	1781045-942R 0+0	7731200-8	1291785-8 7	735120-9	3
3577547-9	7805440-941R 0+0	1032406-7	3047906-9 4	103804-9	4
8116636-9	6614968-9 1438678-940	k G+0	3001110-7 5	313086-9	5
1238474-8	1287040-8 9870019-9	5948243-9	1102423-839R	0+0	6
4193276-7	3225548-8 2574680-9	4200667-9	5352384-9 57	787241-9	7
7521997-9	7063515-9388 0+0	6207061-7	4131107-9 21	515195-9	8

-147460-8	3856838-9	6351238-9	6564 295-9	6115635-9	4125447-9	9
1187430-8	-258527-640	R 0+0	T			66

Complete output listings from all the sample problems are available in the TRANSX 2.0 distribution package.

•

.

: . .

Appendix B. MATXS FILE DESCRIPTION

The foliowing description of the MATXS format uses the standard CCCC form. It was copied from the comment cards at the start of the MATXSR module of NJOY 91.0.

С PROPOSED 09/09/77 C (NCDIFIED 09/80) C (NONENCLATURE CHANGED 06/8") C (NODIFIED FOR CONST SUB-BLOCKS 06/90) C (ORDERING CHANGED 10/90) C CF NATIS CZ NATERIAL CROSS SECTION FILE С Ć THIS FILE CONTAINS CROSS SECTION CI VECTORS AND MATRICES FOR ALL CI PARTICLES, NATERIALS, AND REACTIONS; CI DELAYED NEUTRON SPECTRA BY TIME GROUP: CI AND DECAY BEAT AND PROTON SPECTRA. 4 1.1 FORMATS GIVES ARE FOR FILE EICHANGE ONLY C С C C---CS i FILE STRUCTURE CS CS RECORD TYPE PRESENT IF CS ************** CS FILE IDESTIFICATION ALVAYS CS FILE CONTROL ALWATS CS -SET HOLLERITH IDANT CONTINUES ALVATS CS FILE PATA ALVAYS CS CS +++++++++++(REPEA: FOR ALL PARTICLES) CS GROUP STRUCTURES ALVAYS r • • ***** CS. CS ***************** (REPEAT FOR ALL MATERIALS) CS • MATERIAL CONTROL ALVAYS CS CS · ······(AEPEAT FOR ALL SUBNATERIALS) CS • • VECTOR CONTROL 1108.GT.0 CS. • • · · ······(REPEAT FOR ALL VECTOR BLOCKS) CS CS • • • VECTOR BLOCK NIDB.CT.O CS • • ••••• CS | . . CS

84

CS NATRIX CONTROL . . . E20. (T. .!.) CS . . . CS • • • • MATRII SUB-BLOCK CS 821 GT.U CS CS . . . JCOIST.GT.O CS . . . CONSTANT SUB-BLOCK CS . . . ---CS *********** С C С CR FILE IDENTIFICATION С CL ENAME. (RUSE(I), I=1,2), IVERS -С -CV 1+3+#ULT C • CB FORMAT(45 OV , A8, 18+, 2A8, 18+, 16) С HNAME NOLLERITE FILE NAME - NATIS - (AB) CD -CD BUSE HOLLPAITH USER IDENTIFIATION (AS) -IVERS CD FILE VERSION NURBER XULT DOUBLE PRECISION PARAMETER CD CD 1- AS VORD IS SINCLE WORD CD 2- AS WORD IS DOUBLE PRECISION WORD C С C CR FILE CONTROL С CL BPART, STYPE, SHOLL, SHAT, MAXW, LENGTH C CV 6 5 CB FORMAT(4H 1D .4I6) С CD NFART TUNDER OF PARTICLES FOR VEICE GROUP CD STRUCTURES ARE GIVEN BTYPEBUNBER OF DATA TYPES PRESENT IN SETNEOLLBUNBER OF VCRDS IN SIT BOLLERITH CD CD CD IDENTIFICATION RECORD CD BNAT SUMBER OF MATERIALS OF FILE CD HAIV NAXINUN RECORD SIZE FOR SUB-BLOCKING LENGTH CD LENGTE OF FILE С C---С

C

C-----_____ SET HOLLERITE IDENTIFICATION CR C CL (ESETID(I), I=1, NHOLL) С CV C FORMAT(4H 2D .8A8/(9A8)) CB C CD ISETID HOLLERITE IDENTIFICATION OF SET (AS) (TO BE EDITED OUT 72 CHARACTERS PER LIFE) CD C C С Ck FILE DATA С CL (EPRT(J), J=1, HPART), (HTYPE(K), K=1, HTYPE), (HMATH(1), I=1, HWAT), CL 1(HGRP(J), J=1, HPART), (JIBP(K), K=1, HTYPE, (JOUTP(K), K=1, HTYPE), CL 2(#SUBN(I)I=1, #NAT), (LOCH(I), I=1, #NAT) C CV (HPART+STYPE+SHAT)+NULT+2+STYPE+SPART+2+SHAT С CB FORMAT(4E 3D , SAS/(9AS)) EPRT, ETYPE, EMATE CB FORMAT(1216) HGRP, JINP, JOUTP, ESURN, LOCH C CD EPRT(J) BOLLEAITE IDENTIFICATION FOR PARTICLE J CD 1 TEUTROE G CD GANDIA CD PROTOS 7 œ D DEUTEROS CD T TAITOR CD 1 HE-3 FUCLEUS CD ALPEA (EE-4 SUCLEUS) CD BETA CD 8 RESIDUAL OR RECOIL CD (HEAVIER TEAH ALPBA) ETYPE(E) BULLERITE IDENTIFICATION FOR DATA TYPE K CD CD **YSCAT** SEUTROS SCATTERING TEUTROS ISDUCED GANNA PRODUCT:58 CD IG GSCAT CD GAMES SCATTERING PROTOS ISDUCED SEVTROS PRODUCTION đ 21 ຕາ . . CD . CD CD DXI DELAYED HEUTROE DATA CD DKIG DELAY BEAT AND GANNA DATA CD DEB DECAY BETA DATA CD ENATE(I) BOLLERITH IDENTIFICATION FOR MATERIAL I œ IGRP(J) SURSER OF ESERGY GROUPS FOR PARTICLE J œ JINP(K) TYPE OF INCIDENT PARTICLE ASSOCIATED WITH CD DATA TYPE K. FOR DE DATA TYPES, JINP IS C.

91

```
JOUTP(K) TYPE OF OUTGOING PARTICLE ASSOCIATED WITH
CD
CD
             DATA TYPE K
   HSUBH(I) NUMBER OF SUBMATERIALS FOR MATERIAL I
CD
   LOCH(I) LOCATION OF MATERIAL I
CD
C
C
C
GROUP STRUCTURE
CR
С
   (GPB(I),I=1,NGR),ENIN
CL
C
CC
  NGR=NGRP(J)
С
CV
  HGRP(J)+1
С
CB
   FORMAT(48 4D ,1P5E12.5/(6E12.5))
С
CD GPB(I) MAXIMUM ENERGY BOUND FOR GROUP I FOR PARTICLE J
   ENIN NININUN ENERGY ROUND FOR PARTICLE J
CD
С
С
С
CR
       NATERIAL CONTROL
С
CL
   HMAT, AMASS, (TEMP(1), SIGZ(1), ITYPE(1), H:D(1), H2D(1),
CL
  1LOCS(I), I=1, WSVBW)
С
CV
  NULT+1+0+NSUBM
С
CB
  FORMAT(4H 6D ,A8,1H+,1P2E12.5/(2E12.5,5I6))
С
CD
   IMAT
          BOLLERITE MATERIAL IDENTIFIER
          ATONIC VEIGHT RATIO
CD
   AMASS
   TENP
CD
          AXBIEST TEXPERATURE OR OTHER PARAMETERS FOR
CD
              SUBNATERIAL I
   SIGZ DILUTION FACTOR OR OTHER PARAMETER'. FOR
CD
              SUBNATERIAL I
                                               -
CD
   ITYPE
CD
          DETA TYPE FOR SUBMATERIAL :
          SUMBER OF VECTORS FOR SUBMATERIAL I
CD
   W1D
          SUMBER OF MATRIX BLOCKS FOR SUBMATERIAL I
CD
   12D
   LOCS LOCATION OF SUBNATERIAL Z
CD
C
C
С
VECTOR CONTROL
CL
C
CL (EVPS(I), I=1, E1D), (EPG(I), I=1, E1D), (ELG(I), I=1, E1D)
```

C CV (WULT+2)+#1D С CB FORMAT(4E 7D ,8A8/(9A6)) IVPS CB FORMAT(1218) IBLK, BYG, BLG С CD EVPS(I) NOLLERITE IDESTIFIER OF VECTOR CD TELIS NEUTRON ELASTIC SCATTERING CD 125 $(\mathbf{I}, 2\mathbf{I})$ Ð IIF SECOND CHANCE FISSION CD GABS GANNA ABSORPTION CD P21 PROTONS IN. 2 NEUTRONS OUT CD C0 CD CD #FG(I) FUNDER OF FIRST GROUP IN BAND FOR VECTOR I CD HLG(1) NUMBER OF LAST GROUP IN BAND FOR VECTOR I C C-----C C **...** VECTOR BLOCK C CL (VPS(I),I=1,KMAX) С KNAX-SUN OVER GROUP BAND FOR EACH VECTOR IN BLOCK J CC С CV XNAX C CI FORMAT(48 8D ,1P5E12.5/(6E12.5)) C CD VPS(I) DATA FOR GROUP BARDS FOR VECTORS IN BLOCK J. ¢ BLOCK SIZE IS DETERMINED BY TAKING ALL THE GROUP BANDS THAT HAVE A TOTAL LENGTH LESS THAN OR EQUAL CD -CD TO HALV. C C--C C Cì SCATTERING MATRIX CONTROL C CL ENTX, LORD, JCOUST. CL 1(JBAND(L),L=1,NOUTG(X)),(IJ3(L),L=1,NOUTG(E)) C CY NULT+2+2+NOUTG(X) C PORMAT(48 90 .48/(1216)) CB ENTI, LORD, JCONST, CL JBAED.IJJ C œ XXXX BOLLERITE IDENTIFICATION OF BLOCK Ð LORD NUMBER OF ORDERS PRESENT

CD JCOIST FUNBER OF GROUPS WITH CONSTANT SPECTRUN JBAWD(L) BANDWIDTH FOR GROUP L CD IJJ(L) LOVEST GROUP IN BAND FOR GROUP L CD С C-----______ C C CL SCATTERING SUB-BLICK C CL. (SCAT(K), K=1, KMAX)C XMAX-LORD TIMES THE SUM OVER ALL JBAND IN THE GROUP RANGE OF CC CC TEIS SUB-BLOCK С CB FORM/T(5E 10D ,1P5E12.5/(6E12.5)) С C¥ XMAX C CD SCAT(X) MATRIX DATA GIVEN AS BANDS OF ELEMENTS FOR INITIAL GROUPS THAT LEAD TO EACH FINAL GROUP. THE ORDER CD CD OF THE ELEMENTS IS AS FOLLOWS: BAND FOR 70 OF CROUP I, BAND FOR PI OF GROUP I, ..., BAND FOR PO CD OF GROUP I+1, BAND FOR P1 OF SROUP I+1, ETC. THE CD GROUPS IN EACH BAND ARE GIVEN IN DESCENDING UNDER. CD CD THE SIZE OF EACH SUB-BLOCK IS DETERMINED BY THE TOTAL LENGTH OF A GROUP OF BANDS THAT IS LESS THAN -CD CD OR EQUAL TO MAXY. CD IF JCONST.GT.O, THE CONTRIBUTIONS FROM THE JCOSST CD -LOW-ENERGY GROUPS ARE GIVEN SEPARATELY. CD С C----С С CONSTANT SUB-BLOCK CR С (SPEC(L), [.=1, HOUTG(K)), (PROD(L), L=L1, HIHG(K)) CL С L1=BING(X)-JCOBST+1 CC C CV HOUTG(K)+JCONST С CB FORMAT(48110 ,1P5E12.5/(6E12.5)) C SYEC HORNALIZED SPECTRUM OF FINAL FARTICLES FOR INITIAL CD PARTICLES IN GROUPS LI TO NING(N) CD PRODUCTION CROSS SECTION (E.G., NU-SIGY) FOA CD P100 -INITIAL GROUPS LI TEROUGE SING(R) CD CD -THIS OPTION IS NORMALLY USED FOR THE ENERGY-INDEPENDENT CD NEUTRON AND PROTON SPECTRA PROM PISSION AND RADIATIVE C

CD	CAPTURE USUALLY SEEN AT LOW ENERGIES.	-
С		-
C		

.

Appendix C. RTFLUX FILE DESCRIPTION

The following description of the RTFLUX format uses the standard CCCC form. These lines were copied from the CCCC-IV reference.

REVISED 11/30/76 С С CF RTFLUX-IV REGULAR TOTAL FLUXES CE 7 C C C------CR FILE IDENTIFICATION C CL ENAME, (HUSE(1), I=1,2), IVERS С CV 1+3+NULT С CD HNAME HOLLERITH FILE NAME - RTFLUX - (A6) HOLLERITH USER IDENTIFICATION (A6) CD HUSE IVERS FILE VERSION NUMBER CD NULT CD DOUBLE PRECISION PARAMETER CD 1- A6 WORD IS SINGLE WORD 2- A6 WORD IS DOUBLE PRECISION WORD CD С С С SPECIFICATIONS (1D RECORD) CR С CL NDIN, NGROUP, NINTI, NINTJ, NINTK, ITER, EFFK, POWER, WBLOK С C₩ 9=NUMBER OF WORDS С CD NDIN NUMBER OF DIMENSIONS CD IGROUP NUMBER OF ENERGY GROUPS ۹. NINTI CD NUMBER OF FIRST DIMENSION FINE MESH INTERVALS CD NUMBER OF SECOND DIMENSION FINE MESH INTERVALS -NINTJ CD WINTJ.EQ.1 IF WDIN.EQ.1 CD NUMBER OF THIRD DIMENSION INTERVALS NINTK -CD NINTK.EQ.1 IF NDIN.LE.2 _ CD ITER OUTER ITERATION NUMBER AT WHICE FLUX WAS WRITTEN -CD EFFK EFFECTIVE MULTIPLICATION FACTOR POVER POWER IN WATTS TO WHICH FLUX IS NORMALIZED CD -DATA BLOCKING FACTOR CD NBLOK CD IF NDIN.EQ.1. THE GROUP VARIABLE IS BLOCKED -CD INTO NBLOK BLOCKS (SEE 2D RECORD BELOW) CD IF NDIN.GE.2, THE 2ND DIMENSION VARIABLE IS -

15

CD BLOCKED INTO NBLOK BLOCKS (SEE 3D RECORD) -С С С CR ONE DIMENSIONAL REGULAR TOTAL FLUX (2D RECORD) С _ CC PRESENT IF NDIM.EQ.1 С CL ((FREG(I, J), I=1, WIWTI), J=JL, JU)----SEE STRUCTURE BELOW----С C¥ **WINTI+(JU-JL+1)=WUNBER OF WORDS** -С -С DO 1 M=1, WBLOK -С 1 READ(N) +LIST AS ABOVE+ _ С WITH M AS THE BLOCK INDEX. JL=(N-1)+((NGROUP-1)/WBLOK+1)+1 -CC CC AND JU=MINO(NGROUP, JUP) WHERE JUP=N+((NGROUP-1)/NBLOK+1) -С CD FREG(I, J) ONE DIMENSIONAL REGULAR TOTAL FLUX -CD BY IJTERVAL AND GROUP _ С С С HULI-DIMENSIONAL REGULAR TOTAL FLUX (3D RECORD) CR С _ CC PRESENT IF NDIN.GE.2 С CL ((FREG(I,J),I=1,WIWTI),J=JL,JU)----SEE STRUCTURE BELOW----_ С -WINTI+(JU-JL+1)=NUMBER OF WORDS C¥ -С -С DO 1 L=1, MGROUP -С DO 1 M=1.WINTK -С DO 1 M-1, WBLOK _ С 1 READ(H) +LIST AS ABOVE+ C -CC WITH M AS THE BLOCK INDEX, JL=(M-1)+((WINTJ-1)/WBLOK+1)+1 -CC AND JU=NINO(NINTJ, JUP) WHERE JUP=N+((NINTJ-1)/NBLOK+1) -C _ CD FREG(I, J) MULTI-DIMENSIONAL REGULAR TOTAL FLUX _ CD BY INTERVAL AND GROUP С _ -----

Appendix D. RZFLUX FILE DESCRIPTION

The following description of the RZFLUX format uses the standard CCCC form. These lines were copied from the CCCC-IV reference.

С REVISED 11/30/76 С CF R2FLUX-IV CE REGULAR ZONE FLUX BY GROUP, AVERAGED OVER EACH ZONE С С С CR FILE IDENTIFICATION С CL HWANE.(HUSE(I), I=1,2), IVERS С CV 1+3•NULT С CD BUAME HOLLERITH FILE NAME - RZFLUX - (A6) EUSE CD HOLLERITH USER IDENTIFICATION (86) CD IVERS FILE VERSION NUMBER CD NULT DOUBLE PRECISION PARAMETER CD 1- A6 WORD IS SINGLE WORD CD 2- A6 WORD IS DOUBLE PRECISION WORD С C---С С CR SPECIFICATIONS (1D RECORD) С TIME, POWER, VOL, EFFK, EIVS, DKDS, TNL, TNA, THSL, THBL, THBAL, THCRA, CL 1(X(I), I=1,3), WBLOK, ITPS, WZONE, WGROUP, WCY CL С CV 20=NUMBER OF WORDS С CD TIME REFERENCE REAL TIME, DAYS CD POVER POWER LEVEL FOR ACTUAL NEUTRONICS PROBLEM. CD WATTS TERMAL CD VOL VOLUME OVER WHICH POWER WAS DETERMINED, CC CD EFFX EFFECTIVE MULTIPLICATION FACTOR CD EIVS EIGENVALUE OF SEARCE PROBLEM CD DKDS DERIVATIVE OF SEARCH PROBLEM CD THL TOTAL NEUTRON LOSSES CD TNA TOTAL NEUTRON ABSORPTIONS CD TISL TOTAL NEUTRON SURFACE LEAKAGE CD TRBL TOTAL NEUTRON BUCKLING LOSS CD TEBAL TOTAL NEUTRON BLACK ABSORBER LOSS CD THCRA TOTAL NEUTRON CONTROL ROD ABSORPTIONS

CD	X(I),I=1,3	RESERVED	-
CD	BBLOK	DATA BLOCKING FACTOR. THE GEOMETRIC ZONE	-
CD		VARIABLE IS BLOCKED INTO BBLOK BLOCKS.	-
CD	ITPS	ITERATIVE PROCESS STATE	-
CD		=0, NO ITERATIONS DONE	-
CD		=1, CONVERGENCE SATISFIED	-
CD		=2. NOT CONVERGED, BUT CONVERGING	-
CD		=3, NOT CONVERGED, NOT CONVERGING	-
CD	NZONE	NUNBER OF GEUNETRIC ZONES	-
CD	NGROUP	NUMBER OF ENERGY CROUPS	-
CD	XCY	REFERENCE COUNT (CYCLE NUMBER)	-
С			-
C			
C			
С			
C			
CR	FLUX	VALUES (2D RECORD)	-
С			-
CL	((2GF(K,J),	K=1,NGROUP), J=JL, JU)SEE STRUCTURE BELOW	-
С			-
C¥	NGROUP+(JU-	JL+1)=HUNBER OF WORDS	-
С			-
С	DO 1 M=1.WB	LCK	-
С	1 READ(N) +LI	ST AS ABOVE•	••
С			-
CC	PTIV	M AS THE BLOCK INDEX, JL=(N-1)+((NZONE-1)/NBLOK+1)+1	-
CC	AND J	U=MINO(NZONE,JUP) WHERE JUP=N+((NZONE-1)/NBLOK+1)	-
C		· · · · · · · · · · · · · · · · · · ·	-
CD	ZGF(K,J)	REGULAR ZONE FLUX BY GROUP, AVERAGED OVER ZONE.	-
CD		NEUTRONS/SEC-CN++2.	-
C			-
C			

• • •

Appendix E. RZMFLX FILE DESCRIPTION

The following description of the RZMFLX format uses the standard CCCC form.

С PROPOSED С CF RZMFLI-IV 3D REGULAR ZONE-AVERAGED NOMENTS FLUX С С С CR FILE IDENTIFICATION С CL HNAME, (HUSE(I), I=1,2), IVERS С CV 1+3+NULT С CD BNANE HOLLERITH FILE NAME - RZMFLI -(46) CD HUSE HOLLERITH USER IDENTIFICATION (16) CD IVERS FILE VERSION NUMBER CĎ MULT DOUBLE PRECISION PARAMETER CD 1- A8 WORD IS SINGLE WORD CD 2- A8 WORD IS DOUBLE PRECISION WORD С C-С С CR SPECIFICATIONS С CL NDIN, NGROUP, NINTI, NINTJ, NINTK, NORD, EPFK, POWER С CV 8=NUMBER OF WORDS С CD NDIN NUMBER OF DIMENSIONS NGROUP CD NUMBER OF GROUPS CD NINTI NUMBER OF FIRST DIMENSION INTERVALS CD BINTJ NUMBER OF SECOND DIMENSION INTERVALS CD NINTJ.EQ.1 IF NDIM.EQ.1 CD BISTK NUMBER OF THIRD DIMENSION INTERVALS CD WINTK.EQ.1 IF NDIM.LE.2 NORD CD NUMBER OF LEGENDRE MOMENTS CD EFFK EFFECTIVE MULTIPLICATION FACTOR CD POVER POWER IN WATTS TO WHICH PLUX IS NORMALIZED С C--С С

C		
CR	REGULAR NONENTS FLUXES ON MULTIDINENSIONAL INTERVALS	-
C		-
C	((PLUX(W,I),W=1,WUND,I=1,WIWII)WUTE STRUCTURE BELCH	-
CW	NORD+NINTI=NUMBER OF WORDS	_
C		-
С		-
C		

•

Appendix F. ISOTXS FILE DESCRIPTION

The following description of the ISOTXS format uses the standard CCCC form. It was copied from the CCCC·IV report.²⁵

REVISED 11/30/76 C C ISOTIS-IV CF MICROSCOPIC GROUP NEUTRON CROSS SECTIONS CE С THIS FILE PROVIDES & BASIC BROAD GROUP C¥ LIBRARY, ORDERED BY ISOTOPE. CI С FORMATS GIVEN ARE FOR FILE EICHANGE ONLY CI С _____ C С C----CS FILE STRUCTURE CS PRESENT IF CS RECORD TYPE CS 3325532325323332 FILE IDENTIFICATION CS ALVAYS CS FILE CONTROL ALVAYS CS FILE DATA ALVAYS CS FILE-WIDE CHI DATA ICHIST.GT.1 CS *****************(REPEAT FOR ALL ISOTOPES) CS ISOTOPE CONTROL AND GROUP CS INDEPENDENT DATA CS ALVAYS PRINCIPAL CROSS SECTIONS CS ALVAYS • CS ISOTOPE CHI DATA ICHI.GT.1 • CS • •••••••••(REPEAT TO ISCHAY SCATTERING BLOCKS) CS • • •••••••(REPEAT FROM 1 TO USELOK) CS • • • SCATTERING SUB-BLOCK LORDH(N).GT.O CS *********** CS С **C**-С С CR FILE IDENTIFICATION C CL HWAME, (HUSE(I), I=1,2), IVERS С CH 1+3+NULT С CB FORMAT(4H OV , A8, 1H+, 2A6, 1H+, 16) С

CD	BNAME	HOLLERITH FILE	NAME - MATIS - (A8)	-
CD	HUSE	HOLLERITH USER	IDENTIFIATION (A8)	-
CD	IVERS	FILE VERSION N	JMBER	-
CD	NULT	DOUBLE PRECISI	DU PARAMETER	-
CD		1- A6 VOR	D IS SINGLE WORD	-
CD		2- A6 VOR	D IS DOUBLE PRECISION WORD	-
С				-
C				
С				
С				
C			\	
CR	FIL	E CONTROL (1D R	ecord)	-
С				-
CL	NGROUP, NIS	O, NAXUP, NAXDU, HAXO	RD, ICHIST, USCHAI, USBLOK	-
С				~
CV	8=NUMBER O	F WORDS		-
C				-
CB	FORMAT(4H	1D ,4I6)		-
C				-
CD	IGROUP	JUNBER OF ENERGY	GROUPS IN FILE	-
CD	IISU MARUD	NURBER OF ISOTOP	23 IN FILE	-
CD	MAIUP	MAXIMUM NUMBER U	P UPSCAITER GROUPS	-
	MAXORD	MAXINUM SCATTERY	P DUBBCAILER GRUUPS	-
	REAURD	ISCREDER FYDA	ICTOR THREE HERD TH FILE)	_
	108167	LEVEBURE EATA	ISIUN INDER USED IN FILE). I Concernin Stac	_
	108121	FILL-WIDE FISSIO	NO PTID_UTOP COPCTBIN	-
		ICHIST EQ.U	TIT-TIDE CHI VECTOR	-
		ICHIST CT 1	FILE-VIDE CHI VECTOR	_
	NSCHAT	MATNIN WINEPP OF	BLOCKS OF SCATTERING DATA	-
	NCRI OK	SUBBLACKING CONT	DESCRIPTION SCATTER WATRICES THE	-
CD CD	# JULUR	SCATTERING DA	TA ARE SUBBLOCKED INTO USBLOK	-
CD		RECORDS (SUBBL	CKS) PER SCATTERING BLOCK.	-
c				-
C				
c				
c				
C				
CR	FILE	DATA		-
С				-
CL	(HSETID(I)	,I=1,12),(HISONM(I), I=1,HISO),	-
CL	1(CHI(J), J=	1, MGROUP), (VEL(J),	J=1, NGROUP,	-
CL	2(EMAX(J),J	=1, NGROUP), EMIN, (L	DCA(I),I=1,WISO)	-
С	-			-
C₩	(#IS0+12)+	NULT+1+NISO		-
C¥	+NGROUP+(2	+ICHIST+(2/(ICHIST	+1)))=JUNBER OF VORDS	-
С				-
СВ	FORMAT(4H	2D ,1H+,11A6,1H+	esetid, eisonn	-
CB	118+, 46, 18+	,9(1X,A6)/(10(1X,A	5)))	-
CB	FORMAT(SE1	2.5)	CEI (PRESENT IF ICHIST.EQ.1)	-
СВ	FORMAT(6E1	2.5)	vel , enax , enix	-
CB	FORMAT(121	6)	LOCA	-
C				-

. .

-•

CD HSETID(I) HOLLERITH IDENTIFICATION OF FILE (A6) CD HISONM(I) HOLLERITH ISOTOPE LABEL FOR ISOTOPE I (A6) CD CHI(J) FILE-WIDE FISSION SPECTRUM (PRESENT IF ICHIST.EQ.1) CD VEL(J) MEAN NEUTRON VELOCITY IN GROUP J (CM/SEC) CD ENAX(J) MAXIMUM ENERGY BOUND OF GROUP J (EV) CD ENIN MININUM ENERGY BOUND OF SET (EV) LOCA(I) CD NUMBER OF RECORDS TO BE SKIPPED TO READ DATA FOR CD ISOTOPE I. С C--С С CR FILE-WIDE CHI DATA (3D RECORD) С CC PRESENT IF ICHIST.GT.1 С CL ((CHI(K, J), K=1, ICHIST), J=1, NGROUP), (ISSPEC(1), I=1, NGROUP) С CV MGROUP+(ICEIST+1)=#UNBER OF WORDS С CB FORMAT(4H 3D ,1P5E12.5/(6E12.5)) CHI CB FORMAT(1216) ISSPEC С CD CHI(K,J) FRACTION OF NEUTRONS ENITTED INTO GROUP J AS A CD RESULT OF FISSION IN ANY GROUP, USING SPECTRUM K CD ISSPEC(I) ISSPEC(I)=K IMPLIES THAT SPECTRUM K IS USED CD TO CALCULATE EMISSION SPECTRUM FROM FISSION -CD IN GROUP I С C-----С С C-----CR ISOTOPE CONTROL AND GROUP INDEPENDENT DATA (4D RECORD) С HABSID, HIDENT, HNAT, ANSS, EFISS, ECAPT, TEMP, SIGPOT, ADENS, KBR, ICHI, -CL CL 1IFIS, IALF, INP, IN2N, IND, INT, LTOT, LTRN, ISTRPD, CL 2(IDSCT(W), W=1, WSCMAX), (LORD(W), W=1, WSCMAX),CL 3((JBAND(J,W), J=1, WGROUP), W=1, WSCMAX),CL 4((IJJ(J, W), J=1, WGROUP), W=1, WSCMAX)С CV 3+NULT+17+NSCNAI+(2+NGROUP+2)=NUMBER OF GROUPS С CB FORMAT(4H 4D ,3(1X,A6)/6E15.5/(1216)) С HOLLERITH ABSOLUTE ISOTOPE LABEL - SAME FOR ALL CD HABSID CD VERSIONS OF THE SAME ISOTOPE IN FILE (AG) CD HIDEET IDENTIFIER OF LIBRARY FROM WHICH BASIC DATA CD CAME (E.G. ENDF/B) (A6) CD AMASS GRAM ATOMIC WEIGHT CD EFISS TOTAL THERMAL ENERGY YIELD/FISSION (V.SEC/FISS) CD ECAPT TOTAL THERMAL ENERGY YIELD/CAPTURE (W.SEC/FISS)

CD	TENP	ISOTOPE TEMPERATURE (DEGREES KELVIN)	-
CD	SIGPOT	AVERAGE EFFETIVE POTENTIAL SCATTERING IN	-
CD		RESONANCE RANGE (BARNS/ATON)	-
CD	ADENS	DENSITY OF ISOTOPE IN MIXTURE IN WRICH ISOTOPE	-
CD		CROSS SECTIONS WERE GENERATED (A/BARN-CN) -	,
CD	KBR	ISOTOPE CLASSIFICATION	-
CD		O=UNDEFINED	-
CD		1=FISSILE	-
CD		2=FERTILE	-
CD		3=OTHER ACTIVIDE	-
CD		4=FISSION PRODUCT	-
CD		5=STRUCTURE	-
CD		6=COOLANT	-
CD		7=CONTROL	-
CD	ICHI	ISOTOPE FISSION SPECTRUM FLAG	-
CD		ICHI.EQ.O USE FILE-WIDE CHI	-
CD		ICHI.EQ.1 ISOTOPE CHI VECTOR	-
CD		ICHI.GT.1 ISOTOPE CEI MATRIX	-
CD	IFIS	(W,F) CROSS SECTION FLAG	-
CD		IFIS.EQ.O NO FISSION DATA IN PRINCIPAL	-
CD		CROSS SECTION RECORD	-
CD		IFIS.EQ.1 FISSION DATA PRESENT IN PRINCIPAL	-
CD		CROSS SECTION RECORD	-
CD	IALF	(N, ALPHA) CROSS SECTION	-
CD		SAME OPTIONS AS IFIS	-
CD	INP	(N,P) CRO5S SECTION	-
CD		SAME OPTIONS AS IFIS	-
CD	IN2N	(N,2N) CROSS SECTION	-
CD		SAME OPTIONS AS IFIS	-
CD	IND	(N,D) CROSS SECTION	-
CD		SAME OPTIONS AS IFIS	-
CD	INT	(H,T) CROSS SECTION	-
CD		SAME OPTIONS AS IFIS	-
CD	LTOT	NUMBER OF MOMENTS OF TOTAL CROSS SECTION PROVIDED	-
CD		IN PRINCIPAL CROSS SECTIONS RECORD	-
CD	ITRE	NUMBER OF NOMENTS OF TRANSPORT CRUSS SECTION	-
CD		PROVIDED IN PRINCIPAL CROSS SECTIONS RECORD	-
CD	ISTRPD	NUMBER OF COORDINATE DIRECTIONS FOR WHICH	-
CD		COORDINATE DEPENDENT TRANSPORT CROSS SECTIONS	•
CD		ARE GIVEN. IF ISTRPD=0, NO COORDINATE DEPENDENT	-
CD		TRANSPORT CROSS SECTIONS ARE GIVEN.	-
CD	IDSCT(1)	SCATTERING WATRIX TYPE IDENTIFICATION FOR	-
CD		SCATTERING BLOCK N. SIGNIFICANT ONLY IF	**
CD		LORD(U).GT.O	-
CD		IDSCT(N)=000 + NE, TOTAL SCATTERING, (SUN OF	-
CD		ELASTIC, INELASTIC, AND NON SCATTERING	-
CD		MATRIX TERNS).	-
CD		100 + NN, ELASTIC SCATTERING	-
CD		200 + NN, INELASTIC SCATTERING	-
CD		300 + NN, (N,2F) SCATTERING,SEE	-
CD		NOTE BELOW	-
CD		WHERE NN IS THE LEGENDRE EXPANSION INDEX OF THE	-
CD		FIRST MATRIX IN BLOCK N	-

.

•
CD LORD(I) NUMBER OF SCATTERING ORDERS IN BLOCK N. IF CD LORD(W)=0. THIS BLOCK IS NOT PRESENT FOR THIS CD ISOTOPE. IF WW IS THE VALUE TAKEN FROM CD IDSCT(N), THEN THE MATICES IN THIS BLOCK CD HAVE LEGENDRE EXPANSION INDICES OF NN. NN+1. CD NH+2,...,NH+LORD(J)-1 CD JBAND(J, N) NUMBER OF GROUPS THAT SCATTER INTO GROUP J. CD INCLUDING SELF-SCATTER, IN SCATTERING BLOCK N. CD IF JBAND(J,N)=0, NO SCATTERING DATA IS PRESENT CD IN BLOCK N. CD IJJ(J,M) POSITION OF IN-GROUP SCATTERING CROSS SECTION IN CD SCATTERING DATA FOR GROUP J, SCATTERING BLOCK CD N, COUNTED FROM THE FRIST WORD OF GROUP J DATA. CD IF JBAND(J, M). NE.O THEM IJJ(J, M) MUST SATISFY CD THE RELATION 1.LE, 1JJ(J.W).LE.JAND(J.W) С CD NOTE- FOR N.2N SCATTER, THE MATRIX CONTAINS TERMS. CD SCAT(J TO G), WHICH ARE EMISSION (PRODUCTION)-BASED, I.E., ARE DEFINED SUCH THAT MACROSCOPIC CD CD SCAT(J TO G) TIMES THE FLUX IN GROUP J GIVES CD THE RATE OF EMISSION (PRODUCTION) OF MEUTRONS CD INTO GROUP G. С C· С С CR PRINCIPAL CROSS SECTIONS (5D RECORD) С CL ((STRPL(J,L), J=1, MGROUP), L=1, LTRM), Cf. 1((STOPL(J,L), J=1, WGROUP)L=1, LTOT), (SHGAW(J), J=1, HGROUP), CL 2(SFIS(J), J=1, MGROUP), (SMUTOT(.'), J=1, MGROUP), CL 3(CHISO(J), J=1, MGROUP), (SWALF(J), J=1, MGROUP). CL 4(SHP(J), J=1, HGROUP), (SH2H(J), J=1, HGROUP),CL 5(SWD(J), J=1, WGROUP), (SWT(J), J=1, WGROUP),CL 6(STRPD(J,I), J=1, MGROUP), I=1, ISTRPD) С C₩ (1+LTRN+LTOT+IALF+INP+IN2H+IND+INT+ISTRPD+2+IFIS+ CV ICHI+(2/(ICHI+1)))+HGROUP=HUMBER OF WORDS С CB FORMAT(4H 5D ,5E12.5/(6E12.5)) С CD STRPL(J,L) PL WEIGHTED TRANSPORT CROSS SECTION CD THE FIRST ELEMENT OF ARRAY STRPL IS THE CD CURRENT (P1) WEIGHTED TRANSPORT CROSS SECTION CD THE LEGENDRE EXPANSION COEFFICIENT FACTOR (2L+1) CD IS NOT INCLUDED IN STRPL(J,L). CD STOTPL(J,L) PL WEIGHTED TOTAL CROSS SECTION CD THE FIRST ELEMENT OF ARRAY STOTPL 13 THE CD FLUX (PO) WEIGHTED TOTAL CROSS SECTION CD THE LEGENDRE EXPANSION COEFFICIENT FACTOR (2L+1) -CD IS NOT INCLUDED IN STRPL(J,L). CD SHGAM(J) (N.GANMA)

CD SFIS(J) (**I**.F) (PRRSENT IF IFIS.GT.O) CD SHUTOT(J) TOTAL NEUTRON YIELD/FISSION (PRESENT IF IFIS.GT.O) CD SCHISO(J) ISOTOPE CHI (PRESENT IF ICHI.EQ.1) CD SWALF(J) (N,ALPHA) (PRESENT IF IALF.GT.O) (N,P) (N,2N) CD SNP(J) (PRESENT IF INP.GT.O) CD SN2N(J) (PRESENT IF IN2N.GT.0) ----SEE CD NOTE BELOW----CD SWD(J) (W.D) (PRESENT IF IND.GT.O) CD SHT(J) (**I**,T) (PRESENT IF INT.GT.O) STRPD(J.I) COORDINATE DIRECTION I TRANSPORT CROSS SECTION CD CD (PRESENT IF ISTRPD.GT.O) С CI NOTE - THE PRINCIPAL N, 2N CROSS SECTION SN2H(J) CI IS DEFINED AS THE N, 2N REACTION CROSS SECTION. CI I.E., SUCH THAT MACROSCOPIC 2828(J) TIMES THE CI FLUX IN GROUP J GIVES THE MATE AT WHICH N, 2N CI REACTIONS OCCUR IN GROUPJ. THUS, FOR M.2M CI SCATTERING, SH2H(J) = 0.5 + (SUM OF SCAT(J TO G))CI SUMMED OVER ALL G). С C---С С CR ISOTOPE CHI DATA (6D RECORD) С CC PRESENT IF ICHI.GT.1 С CL ((CHIISO(K, J), K=1, ICHI), J=1, JGROUP), (ISOPEC(I), I=1, HGROUP) С CV NGROUP+(ICHI+1)=NUMBER OF WORDS С CB FORMAT(4H 6D ,5E12.5/(6E12.5)) CHIISO CB FORMAT(1215) ISOPEC С CHIISO(K.J) FRACTION OF NEUTRONS EMITTED INTO GROUP J AS A CD CD RESULT OF FISSION IN ANY GROUP. USING SPECTRUM K CD ISOPEC(I) ISOPEC(I)=K IMPLIES THAT SPCTRUM K IS USED CD TO CALCULATE ENISSION SPECTRUM FROM FISSION CD IN GROUP I С C---С С CR SCATTERING SUB-BLOCK (7D RECORD) С CL ((SCAT(K,L),K=1,KMAX),L+1,LORDW С CC KMAX=SUM OVER J OF JBAND(J,N) WITHIN THE J-GROUP RANGE OF THIS CC SUB-BLOCK. IF N IS THE INDEX OF THE SUB-BLOCK. THE J-GROUP CC RANGE CONTAINED VITHIN THIS SUB-BLOCK IS CC JL=(N-1)+((HGROUP-1)/HSBLOK+1)+1 TO JU=WINO(HGROUP, JUP),

```
CC
       VHERE JUP=N+((NGROUP-1)/NSBLUK( ')
C
CC
     LORDE=LORD(E)
CC
     С
C¥
     KNAX+LORDH=FUNBER OF GROUPS
Ċ
CB
     FORMAT(4H 7D ,5E12.5/(6E12.5))
С
CD
     SCAT(K,L) SCATTERING NATRIX OF SCATTERING ORDER L, FOR
CD
                  REACTION TYPE IDENTIFIED BY IDSCT(N) FOR THIS
CD
                  BLOCK. JBAND(J, N) VALUES FOR SCATTERING INTO
CD
                  GROUP & ARE STORED AT ; PCATOPHS L=SUN FROM 1
                                                              -
CD
                  TO (J-1)( OF JBAND(J,N) PLUS 1 TO K-1+JBAND(J,N).
                                                              -
CD
                  THE SUN IS ZERO WHEN J=1. J-TO-J SCATTER IS
CD
                  THE IJJ(J, N)-TH ENTRY IN THE RANGE JBAND(J, N).
                                                              -
CD
                  VALUES ARE STORED IN THE ORDER (J+JUP).
CD
                  (J+JUP-1), \ldots, (J+1), J, (J-1), \ldots, (J-JDW),
CD
                  WHERE JUP=IJJ(J,N)-1 AND JDN=JBAND(J,N)-IJJ(J,N). -
C
C---
```

Appendix G. GOXS FILE DESCRIPTION

The following description of the GOXS format uses the standard CCCC form. The GOXS format is the same as the MACRXS and SNXEDT formats used in the ONEDANT and TWODANT codes.

С DATE 05/12/83 С CF GOIS CE GROUP-ORDERED CROSS SECTIONS FOR SM С THIS FILE CONTAINS CROSS SECTION CI CI TRANSPORT TABLES AND RESPONSE EDITS IN A GROUP-ORDERED FORM CONVENIENT CI FOR USE IN SN TRANSPORT CODES. CI С C С C---______ CS FILE STRUCTURE CS RECORD TYPE PRESENT IF CS CS 392283222222222223233874336892 282223387243239 CS FILE IDENTIFICATION ALWAYS CS FILE CONTROL ALWAYS CS FILE DATA ALVAYS CS EDIT CROSS SECTIONS CS ALVAYS • CS • SCATTERING CONTROL HORD.GT.O CS SCATTERING DATA NORD.GT.O CS ************* С C---С С C----_____ CR FILE IDENTIFICATION С CL HWAME, (BUSE(I), I=1,2), IVERS C CV 1+3+NULT С FORMAT(48 OV ,A8,18+,248,18+,16) CB С CD ETANE HOLLERITH FILE JAME - GOIS -(88) CD AUSE HOLLERITH USER IDENTIFIATION (A6) CD IVERS FILE VERSION NUMBER CD NULT DOUBLE PRECISION PARAMETER 1- AS WIRD IS SINGLE WORD CD 2- AS WORD IS DOUBLE PRECISION WORD CD

С C-----С С CR FILE CONTROL С CL NGROUP, WMAT, NORD, NED, IDPF, LNG, MAXUP, MAXDN, NPRIN, I2LP1 С C₩ 10 С CD IGROUP NUMBER OF GROUPS CD NUMBER OF MATERIALS TAME CD NORD NUMBER OF LEGENDRE ORDERS CD TED NUMBER OF EXTRA EDIT CROSS SECTIONS CD (IN ADDITION TO THE BASIC PRINCIPAL CROSS SECTIONS) CĽ IDPF DOUBLE PRECISION FLAG FOR EDIT AND SCATTERING DATA CD (1/2=SINGLE/DOUBLE) CD LBG LAST NEUTRON GROUP FOR COUPLED SETS CD NAXUP MAXIMUM BUMBER UF UPSCATTER GROUPS CD MAXDN MAXIMUN NUMBER OF DOWNSCATTER GROUPS CD NUMBER OF PRINCIPAL CROSS SECTIONS (0, 4, OR 5) NPRIN 2L+1 TERN INCLUDED IN LIBRARY (0/1=NO/YES) CD I2LP1 С C--С С CR FILE DATA С CL (HNAT(I), I=1, WNAT), (HED(J), J=1, WEDT), (VEL(K), K=1, MGROUP), 1(EG(K),K=1,WGROUP),EMIN CL С CV (HMAT+HED) + MULT+2+HGROUP+1 С HMAT(I) CD BOLLERITH MATERIAL NAMES HED(J) CD SOLLERITH EDIT CROSS SECTION NAMES CD VEL(K) AVERAGE VELOCITY FOR GROUP K (CM/SEC) EG(K) CD MAX ENERGY BOUND FOR GROUP K (EV) CD ENIN NIN ENERGY BOUND FOR GROUP STRUCTURE (EV) NEDT CD NED+NPAIN С C---С С C----CR EDIT CROSS SECTIONS С CL ((EDXS(I,J),I=1,WMAT),J=1,WEDT) С CV NMAT+NEDT С CD EDIS(I,J) EDIT CROSS SECTIONS. THE FOLLOWING 4 REACTIONS

ARE ALVAYS PRESENT IN THIS ORDER WHEN MPRIN=4: CD CD TOTAL TOTAL TRANSPORT ISEC _ CD ABS BALANCE ABSORPTION ISEC CD BUSIGF FISSION NEUTRON PRODUCTION ISEC -CD CHI FISSION SPECTRUM CD ALSO PRESENT WHEN JPRIN=5: TRD TRANSPORT FOR DIFFUSION CD CD ABY EITRA EDITS FOLLOW THESE STANDARD EDITS. CD WEEN NPRINEO, ANY NAMES ARE ALLOWED. С C С SCATTERING CONTROL CR С ((HJJ(L,I),L=i,NORD),I=1,NMAT), CL 1((IJJ(L,I),L=1,WORD),I=1,WMAT) CL С C¥ 2+EORD+EMAT С NJJ(L,I) NUMBER OF GROUPS IN BAND CD CD IJJ(L,I) FIRST GROUP IN BAND С С С C------CR SCATTERING DATA С SCAT(I), I=1, WTAB) CL С CV TAB С CD SCAT(K) SCATTERING DATA. CD SUN OF HJJ(L,I) OVER L FROM 1 TO HORD AND BTAB CD I FRON 1 TO HMAT C

.

Appendix H. EXAMPLE OF A MATXS INDEX FILE

The following text is part of the index for MATXS10, the 30 x 12 library based on ENDF/B-VI. Sections that were removed are marked with ellipses.

FILE DESCRIPTION

30X12 LIBRARY FROM ENDF/B-VI, JAN 1992 P4, 300X AT INFINITE DILUTION STANDARD CLAW WEIGET FUNCTION

FILE DATA

PARTICLE	NANE	NGRP	
1	T	30	
2	G	12	
DATA TYPE	TAKE	JINP	JOUTP
1	NSCAT	1	1
2	NG	1	2
3	GSCAT	2	2

NATERIAL	TAN	NSUB	LOCA
1	H1	3	0
2	8 2	3	16
3	H3	2	34
4	HE3	2	49
5	HE4	2	62
6	LI6	3	75
7	LI7	3	163
8	BES	3	23.
9	B 10	3	250
10	B 11	3	330
11	CHAT	3	388
2	X14	3	426
د:	3 15	3	511
14	015	3	562
15	017	2	644
16	F19	3	689
17	¥423	3	780
18	NGNAT	3	820

_

10	MODA	2	626
	NY87 A107		697
20		3	¥0/
21	SIBAT	. 3	1033
22	3128	J	1110
23	P31	3	1267
24	SWAT	3	1290
25	532	3	1395
26	CLEAT	3	1418
27	AR40	2	1494
28	KHAT	3	1505
29	X41	2	1603
30	CANAT	3	1616
31	SC45	2	1686
32	TINAT	3	1697
22	TTAR	2	1744
24	**** 7147	•	1767
31	1141	4	1740
30	1140	4	1770
30	VIAL	3	1///
37	C R5 0	3	1893
38	C R 52	3	1912
39	C R53	3	1987
40	C154	3	2075
41	N265	3	2119
42	7 2 64	3	2271
43	PE58	3	2334
44	C059	3	2377
45	NI58	3	2449
46	N159	2	2507
47	BIGO	3	2518
48	NI6 1	3	2598
49	N162	3	2661
50	¥164	3	2712
51	CU63	3	2755
52	CU65	3	2878
63	GANAT	3	2965
54	780	3	2986
55	TRUS	1	3035
56	NGYAT		3112
67	CDEAT	i	8186
50	THEAT		1179
50	10041	<i></i>	9050
DV	DA130		3697
00	EU101		3282
61	W153	3	3393
62	EU165	2	3390
63	10165	3	3431
64	EFN).T	2	3484
65	TA181	3	3525
66	V182	3	3563
67	V183	3	3636
68	V184	3	3694
69	V186	3	3760
70	RE186	2	3826
71	32167	2	3843

72	AU197	3	3914
73	PB206	3	3967
74	PB207	3	4150
75	PB208	3	4333
76	BI209	3	4404
77	TE23 0	2	4473
78	TH 232	3	4529
79	PA231	2	4588
80	PA233	2	4634
81	U232	2	4682
82	U233	3	4730
83	U234	2	4770
84	U235	3	4806
85	U236	2	4908
86	U238	3	4988
87	HP2 37	3	5075
88	N P238	2	5225
89	NP239	2	52 38
90	PU236	2	5270
91	PU237	2	5302
92	PU238	2	5346
93	PU239	3	5400
94	PU240	3	5494
95	PU241	3	55 64
96	PU242	3	5623
97	PU243	3	5690
98	PU244	2	5718
99	AN241	3	57 5 2
100	AN242	2	5816
101	AN242N	3	5831
102	AM243	3	5888
103	CH241	2	5953
104	CN242	3	5983
105	CN243	3	6020
106	CH244	3	6081
107	CN245	3	6118
108	CN246	3	6144
109	CN247	3	6192
110	CN248	3	6218
:11	BK249	2	6253
112	CP249	2	6315
113	CF250	3	6367
116	CF251	3	6393
115	CF252	3	6419
116	CP253	2	6445
117	ES253	2	6458

II SUB 1 VECTORS: EVIO 3.000E+02 1.000E+10 **ESCAT** BTOTO NELAS IG REAT KERKA DANE NUBAR II GAXIMA INVEL

MATRICES: WELAS

- E1 SUB 2 3.000E+02 1.000E+10 NG MATRICES: NG
- H1 SUB 3 0.000E+00 1.000E+10 G*CAT VECTORS: GWTO GTOTO GCOH GINCH GPAIR GABS GHEAT NATRICES: GCOH GINCH GPAIR

d_d__dd_____

- H2 SUB 1 3.000E+02 1.000E+10 BSCAT VECTORS: EVTO BTOTO BELAS N2N NG HEAT KERNA DANE NUBAR II GAMMA INVEL MATRICES: BELAS N2N
- H2 SUB 2 3.000E+02 1.000E+10 HG NATRICES: NG
- H2 SUB 3 0.000E+00 1.000E+10 GSCAT VECTORS: GVTO GTOTO GCOH GINCN GPAIR GABS GHEAT NATRICES: GCOH GINCH GPAIR

- H3 SUB 1 3.000E+02 1.000E+10 NSCAT VECTORS: NYTO NTOTO NELAS N2N HEAT KERMA DAME MUBAR XI GANNA INVEL NATRICES: NELAS N2N
- E3 SUB 2 0.000E+00 1.000E+10 GSCAT Vectors: GVTO GTOTO GCOE GINCE GPAIR GABS GHEAT MATRICES: GCON GINCE GPAIR

- HE3 SUB 1 3.000E+02 1.000E+10 NSCAT VECTORS: EVTO NTOTO EELAS NG NP ND BEAT KERMA DANE NUBAR XI GAMMA 1NVEL NATRICES: HELAS
- NES SUB 2 0.000E+00 1.000E+10 GSCAT VECTORS: GVTO GTOTO GCOE GINCE GPAIR GABS GNEAT MATRICES: GCOE GINCE GPAIR

HEA SUB 1 3.000E+02 1.000E+10 HSCAT VECTORS: EVTO BTOTO BELAS HEAT KERNA DANE MUBAR XI

GANNA INVEL MATRICES: NELAS

HE4 SUB 2 0.000E+00 1.000E+10 GSCAT VECTORS: GWTO GTOTO GCOH GINCH GPAIR GABS GHEAT NATRICES: GCOH GINCH GPAIR

LI6 SUB 1 3.000E+02 1.000E+10 NSCAT VECTORS: NVTO NTOTO NELAS NINEL N2NA NO1D NO2D NO3D NO4D NO5D NO6D NO7 NO8D NO9D N10D N11D N12D N13D N14D N16D N17D N18D N19D N20D N21D N22D N23D N24D N25D N26D N27D N28D N29D NG NP NT HEAT KERNA DAME NUBAR XI GAMMA INVEL NATRICES: NELAS N2NA NO1D NO2D NO3D NO4D NO5D NO6D NO7 NO8D NO9D N10D N11D N12D N13D N14D N16D N16D N17D N18D N19D N20D N21D N22D N28D N24D N26D N27D N28D N24D N22D

```
LIG SUB 2 3.000E+02 1.000E+10 NG
MATRICES: NO7 NG
```

LIG SUB 3 0.000E+00 1.000E+10 GSCAT VECTORS: GWTO GTOTO GCOE GINCE GPAIR GABS GHEAT MATRICES: GCOE GINCE GPAIR

- LI7 SUB 2 3.000E+02 1.000E+10 NG NATRICES: NO1 NG
- LI" SUB 3 0.000E+00 1.000E+10 GSCAT VECTORS: GVTO GTOTO GCOE GINCE GPAIR GABS GHEAT NATRICES: GCOE GINCE GPAIR

- SUB 1 3.000E+02 1.000E+10 HSCAT 8E9 VECTORS: NUTO NTOTO NELAS N2N NG NP ID **N**T KERNA DANE NT600 NT650 P00 HA . BEAT P01 NT800 MUBAR XI GANNA INVEL MATRICES: NELAS 121 BE9 SUB 2 3.000E+02 1.0C0E+10 TG MATRICES: NG P01 BE9 SUB 3 0.000E+00 1.000E+10 GSCAT VECTORS: GWTO GTOTO GCOB GINCE GPAIR GABS GHEAT MATRICES: GCOH GINCH GPAIR B10 SUB 1 3.000E+02 1.000E+10 MSCAT VECTORS: NYTO NTOTO NELAS NINEL NO1 102 103 104 NOBA NOSA NOSA NOSA NO7 N10A **X11A** ¥12DAA N14DAA N15P N16A N17A N18DAA N19A **I**13A **H20DAA** H21DAA H22A H23DAA H24DAA H25A H26DAA H27DAA H28P H29DAA HG **XP** ۳D XA NT2A HEAT KERMA DANE NT600 NT601 NT603 NTC04 NT605 NT800 CABS NT801 NUBAR XI GANNA INVEL 103 MATRICES: HELAS HO1 102 104 NOSA NOSA NOT NOBA NOPA NIOA **W11A** N12DAA N13A **H14DAA H15P** N16A N17A N18DAA N19A N20DAA N21DAA N22A N23DAA N24DAA N25A N26DAA N27DAA N28P N29DAA
- B10 SUB 2 3.000E+02 1.000E+10 NG NATRICES: NG MT301 NINEL NP
- B10 SUB 3 0.000E+00 1.000E+10 GSCAT VECTURS: GVTO GTOTO GCOH GINCH GPAIR GABS GHEAT MATRICES: GCOH GINCH GPAIR

U235	SUB	1	3.000E+02	1	.000E+10	ISCA:	Γ		
	VECTORS :	EVTO	NTOTO	NELAS	NIVEL	N2N	N3N	NFTOT	NF
		NNF	N2NF	N3NF	H 01	102	N 03	N 04	NO5
		106	X 07	N 08	109	W10	X11	#12	W13
		X14	¥15	N 16	X17	W18	X19	3 20	W2 1
		X 22	N23	X24	X 25	N26	X 27	X 29	329
		N 30	N 31	3 32	X 33	134	ICI	NG	HEAT
		KERNA	NUBAR	XI	GAMMA	INVEL	TUDEL.	CHID	
	MATRICES:	WELAS	NPT OT	N 01	N 02	NO 3	X04	¥05	N06
		1 07	BOB	109	N 10	W11	W12	X13	X 14

. . .

		W1 5	N16	#17	N18	W19	N 20	W21	X 22
		#2 3	3 24	N 25	N 26	N 27	¥28	¥29	N 30
		N 31	¥32	1 33	N34	N2N	X 3N	NCN.	
· ·	SUB	2	3.000E+02		1.000E+10	NG			
	MATRICES:	NINEL	NFTOT	NG	NNONEL				
U235	SUB	3	0.000E+00		1.0002+10	GSCAT			
	VECTORS:	GWTO	GTOTO	GCOH	GINCH	GPAIR	GABS	GHEAT	
	MATRICES:	GCOH	GINCH	GPAIR	l				
•••									
INDE	CONPLETE								2555. 6 5

A complete listing of this file is included in the TRANSX 2.0 distribution package.